Method for Data-Driven Automated Parameterization of Energy Flexibility Models of Production Systems

Vom Fachbereich Maschinenbau an der Technischen Universität Darmstadt zur

Erlangung des Grades eines Doktor-Ingenieurs (Dr.-Ing.) genehmigte

Dissertation

vorgelegt von Martin Lindner, M. Sc.

Berichterstatter:	Prof. DrIng. Matthias Weigold
Mitberichterstatter:	Prof. DrIng. Reiner Anderl
Tag der Einreichung:	01.07.2024
Tag der mündlichen Prüfung:	22.10.2024

Darmstadt 2024 D17 Lindner, Martin: Method for Data-Driven Automated Parameterization of Energy Flexibility Models of Production Systems Darmstadt, Technical University of Darmstadt Year of publication of the dissertation on TUprints: 2024 URN: urn:nbn:de:tuda-tuprints-288256 Day of the oral examination: 22.10.2024

Published under CC BY 4.0 International

Erklärung

Hiermit erkläre ich, dass ich die vorliegende Arbeit, abgesehen von den in ihr ausdrücklich genannten Hilfen, selbständig verfasst habe. Die "Grundsätze zur Sicherung guter wissenschaftlicher Praxis an der Technischen Universität Darmstadt" und die "Leitlinien zum Umgang mit digitalen Forschungsdaten an der TU Darmstadt" wurden in den jeweils aktuellen Versionen bei der Verfassung der Dissertation beachtet.

Frankfurt am Main, 01.07.2024

Martin Lindner

Forword of the Editor

The German Federal Government has set the goal of achieving climate neutrality by 2045. The expansion of renewable energy sources is being accelerated to meet this target. With the 2022 "Easter Package" from the Federal Ministry for Economic Affairs and Climate Action (BMWK), the aim is to triple the speed of this expansion. Integrating such a high share of renewable energy successfully requires an energy system capable of managing an increasingly fluctuating electricity supply.

The Federal Ministry of Education and Research (BMBF) launched the Kopernikus Projects in 2015 to facilitate the energy transition. One of these projects, "SynErgie," focuses on aligning industrial processes with a fluctuating energy supply. To use the existing energy flexibility of industrial processes, they must be technically enabled, and an appropriate market and electricity system, along with automation through information and communication technology (ICT), is also essential. ICT is a key element in connecting production and production infrastructure with the market and the electricity system. To achieve this successfully, a standardized description of energy flexibility is essential. The application of a standardized data model to describe energy flexibility enables automated communication between various parties within the energy system.

The Energy Flexibility Data Model (EFDM) represents energy flexibility in a flexibility space on the one hand and specific energy flexibility measures on the other. The EFDM thus enables the exchange of information on energy flexibility in a generic format, regardless of the actual underlying physical infrastructure and measures. This is where this dissertation by Mr. Martin Lindner comes in. The overarching aim of his dissertation is to develop a method for the automated parameterization of an energy flexibility data model using machine learning approaches and tools. Mr. Lindner refers to this developed approach as "Data-Driven Energy Flexibility Modeling" (DD-EFMod).

Darmstadt, November 2024

Prof. Dr.-Ing. Matthias Weigold

Acknowledgements

This thesis was written during my time as a research associate at the research group of Energy Technologies and Applications in Production (ETA) at the Institute of Production Management, Technology and Machine Tools (PTW) at the Technical University of Darmstadt. Working at the institute and writing this thesis has been a time full of opportunities, challenges, and growth. I am deeply indebted to the many individuals and organizations that have played a crucial role in completing this thesis.

The academic community provided by TU Darmstadt, PTW, and the ETA research group has been a key element in shaping my research by creating an atmosphere of intellectual exchange. My special thanks go to Prof. Dr.-Ing. Matthias Weigold, the institute's director, who gave me the necessary guidance, support, freedom, and trust throughout my time there. I would also like to thank Prof. Dr.-Ing. Reiner Anderl, who took on the role of the co-lecturer.

The research in this thesis was conducted as part of the German research project SynErgie, funded by the Federal Ministry of Education and Research (BMBF). I want to thank all the research and industry partners I worked with in these project for their discussions and contributions to this work.

I want to thank my colleagues at the PTW, especially Tuğrul Öztürk, for all the discussions, and a lot of thanks goes to the ETA research group members Fabian Borst, Heiko Ranzau, Lukas Theisinger, Benjamin Arzt, Tobias Koch, Dr.-Ing. Benedikt Grosch, for contributing to this work's success through technical or practical support, discussions, or other assistance. I especially thank Michael Frank, Tim Frieß, and Dr.-Ing. Daniel Fuhrländer-Völker for their cooperation and intensive research exchange in the last years, their valuable feedback, and proofreading, which significantly improved the quality of this thesis.

My heartfelt thanks go to my family and friends, who have supported me to the best of their ability in all situations and have always placed their deepest trust in me. Finally, I would like to thank you most of all, Alexandra, for your unlimited support. Your encouragement has been a constant source of strength, even during the most challenging times.

Frankfurt am Main, November 2024

Martin Lindner

Abstract

Implementing industrial energy flexibility is a complex challenge within complex production systems. To successfully utilize energy flexibility, it is crucial to ensure product quality, manage production schedules, and understand systemic dependencies. By adapting production processes to volatile energy prices, industrial energy flexibility makes it possible to reduce costs without compromising productivity and minimize the carbon footprint by using renewable energy efficiently. In addition, energy flexibility opens up potential revenue opportunities by trading flexibilities in future dynamic energy systems and markets.

One of the most important aspects of this adaptability is the use of a standardized data model to identify and model flexibilities. However, the complexity of industrial processes and the need for extensive domain knowledge make it difficult to model all relevant production assets. This thesis presents a methodology that simplifies the modeling process for describing energy flexibility. Therefore, the aim of the thesis is to develop an automated parameterization methodology for an energy flexibility model, hypothesizing that data-driven, automatically parameterized models and machine learning techniques can be used.

Using the Design Research Methodology, this thesis provides a comprehensive understanding of the current state of science and technology related to industrial energy systems, digital production, and energy flexibility modeling. The research identifies a research need in this area, formulates research questions and hypotheses, and develops the Data-Driven Energy Flexibility Modeling (DD-EFMod) method. This method is validated using a use case that confirms the feasibility of using data analytics and machine learning algorithms to parameterize energy flexibility, with batch clustering methods showing promising results.

In addition, the work shows the energy and cost savings through energy-flexibility measures based on the detailed modeling of energy flexibility. The prototypical application use case at the ETA Research Factory shows that the energy-flexibility measure *change processing sequence* enables cost savings of 9.2%. Additional cost savings of up to 69.4% was achieved through a combination of the energy-flexibility measures *change processing sequence* and *shift start of job* within the validation of the use case.

Key Words List:

Demand Response, Manufacturing, Energy Flexibility, Energy Flexibility Data Model, Energy Flexibility Modeling, Data-Driven Modeling

Zusammenfassung

Die Umsetzung von industrieller Energieflexibilität stellt eine komplexe Herausforderung innerhalb komplexer Produktionssysteme dar. Um die Energieflexibilität erfolgreich zu nutzen, ist es entscheidend, die Produktqualität zu gewährleisten, Produktionszeitpläne zu verwalten und systemische Abhängigkeiten zu verstehen. Durch Anpassung von Produktionsprozessen an volatile Energiepreise ermöglicht industrielle Energieflexibilität, ohne die Produktivität zu beeinträchtigen, Kosten zu senken und den CO_2 -Fußabdruck zu minimieren, indem erneuerbare Energien effizient genutzt werden. Darüber hinaus eröffnet Energieflexibilität potenzielle Einnahmequellen durch Handel mit Flexibilitäten in zukünftigen dynamischen Energiesystemen und -märkten.

Einer der wichtigsten Aspekte dieser Anpassungsfähigkeit ist die Verwendung eines standardisierten Datenmodells zur Ermittlung und Modellierung von Flexibilitäten. Die Komplexität industrieller Prozesse und der Bedarf an umfangreichem Fachwissen machen es jedoch schwierig, alle relevanten Produktionsanlagen zu modellieren. In dieser Arbeit wird eine Methodik vorgestellt, die den Modellierungsprozess zur Beschreibung der Energieflexibilität vereinfacht. Ziel der Arbeit ist es daher, eine automatisierte Parametrisierungsmethode für ein Energieflexibilitätsmodell zu entwickeln, wobei die Hypothese aufgestellt wird, dass datengesteuerte, automatisch parametrisierte Modelle und Techniken des maschinellen Lernens verwendet werden können.

Unter Verwendung der Design Research Methodology wird in dieser Arbeit ein umfassendes Verständnis des aktuellen Stands von Wissenschaft und Technologie in Bezug auf industrielle Energiesysteme, digitale Produktion und Energieflexibilitätsmodellierung vermittelt. Die Forschung identifiziert einen Forschungsbedarf in diesem Bereich, formuliert Forschungsfragen und Hypothesen und entwickelt die Methode der datengetriebenen Energieflexibilitätsmodellierung (DD-EFMod). Diese Methode wird anhand eines Anwendungsfalls validiert, der die Machbarkeit des Einsatzes von Datenanalyse und Algorithmen des maschinellen Lernens für die Parametrierung der Energieflexibilität bestätigt, wobei Batch-Clustering-Methoden vielversprechende Ergebnisse zeigen.

Darüber hinaus zeigt die Arbeit die Energie- und Kosteneinsparungen durch Energieflexibilitätsmaßnahmen auf der Grundlage der detaillierten Modellierung der Energieflexibilität. Der prototypische Anwendungsfall in der ETA-Forschungsfabrik zeigt, dass die Energieflexibilitätsmaßnahme Änderung der Bearbeitungsreihenfolge Kosteneinsparungen von 9.2% ermöglicht. Durch eine Kombination der Energieflexibilitätsmaßnahmen Änderung der Bearbeitungsreihenfolge und Verschiebung des Auftragsbeginns wurden im Rahmen der Validierung des Anwendungsfalls zusätzliche Kosteneinsparungen von bis zu 69.4% erzielt.

Stichwörter:

Demand Response, Fertigung, Energie-Flexibilität, Energie-Flexibilitäts-Datenmodell, Energie-Flexibilitäts-Modellierung, Datengesteuerte Modellierung

Table of Contents

Li	st of	Figures	5	xvi
Li	st of	Tables		xix
Li	st of	Abbrev	viations	xxii
Li	st of	Symbo	ls	xxiv
1	Intr	oductio	on	1
	1.1	Motiv	ation	2
	1.2	Gener	al Objective and Contribution	4
	1.3	Resea	rch Methodology and Structure of Work	6
2	Fun	damen	tals	8
	2.1	Indus	trial Energy Systems	9
		2.1.1	Demand-Side Management and Industrial Energy	
			Flexibility	10
		2.1.2	Modeling of Energy Flexibility	14
	2.2	Digita	al Manufacturing	20
	2.3	Data	Analytics and Machine Learning	24
		2.3.1	Data Analytics	24
		2.3.2	Machine Learning and Data Stream Mining	28
		2.3.3	Validation of Clustering Algorithms	32
		2.3.4	Implementation of Machine Learning Algorithms	
			with CRISP-ML(Q)	34
	2.4	Interi	m Summary	36

3	Stat	te of the Art and Research	38
	3.1	Procedure of Systematic Literature Review	39
	3.2	Results of the Literature Review	43
	3.3	Conclusion on the State of the Art and Research Deficit	50
4	Res	earch Gap and Concept	52
	4.1	Research Objective	53
	4.2	Research Questions and Hypotheses	54
	4.3	Requirements and Assumptions	55
	4.4	Interim Summary	59
5	Met	thod Development	60
	5.1	Approach and Structure of the DD-EFMod Method $~$	61
		5.1.1 Layer One - Energy-Flexible System	63
		5.1.2 $$ Layer Two - Application and Implementation $$.	65
		5.1.3 Layer Three - Energy Flexibility Modeling	72
	5.2	Software Concept	76
	5.3	Interim Summary	78
6	Use	Case and Application at the ETA Research Factory	81
	6.1	Process and Machines of the Use Case	82
	6.2	Hardware and Software Enabling	85
	6.3	Scenario Definition	86
	6.4	Interim Summary	87
7	Imp	lementation and Validation	88
	7.1	Scenario 1 - EMAG GT	89
		7.1.1 Key Figure Definition	90
		7.1.2 Implementation \ldots \ldots \ldots \ldots \ldots \ldots \ldots	92
		7.1.3 Evaluation and Results	99
	7.2	Scenario 2 - EMAG GT and Product Storage	109
		7.2.1 Key Figure Definition	110
		7.2.2 Implementation	111
		7.2.3 Evaluation and Results	113
	7.3	Scenario 3 - Cooling Machine and Cold Storage	116

		7.3.1 Key Figure Definition	117
		7.3.2 Implementation	119
		7.3.3 Evaluation and Results	121
	7.4	Scenario 4 - Identification of Dependencies	126
		7.4.1 Key Figure Definition	127
		7.4.2 Implementation	128
		7.4.3 Evaluation and Results	129
	7.5	Evaluation of Flexible Load Measures	134
	7.6	Interim conclusion	140
8	Con	clusion and Outlook	146
	8.1	Summary and Conclusion	147
	8.2	Outlook and Further Research	150
Α	Арр	endix	153
	A.1	Further Related Work	154
		A.1.1 The EFDM GUI Service	154
		A.1.2 The Energy Synchronization Platform (ESP)	155
	A.2	Additional Results	157
	A.3	Detailed Description of the EFDM	160
Re	feren	ices	167
0	vn Pi	ublications	188
Сι	ırricu	lum Vitae	192

List of Figures

1.1	Overview of the energy market in Germany	3
1.2	Methodology for energy flexibilization of factories	5
2.1	Spaces of the energy-flexibility potentials	11
2.2	Classes of the Energy Flexibility Data Model	14
2.3	Schematic explanation of a flexible load measure and all	
	relevant describing periods and times	19
2.4	Reference Architecture Model Industry 4.0	21
2.5	Subdivision of machine learning methods into subgroups	28
2.6	Phases of the CRISP-ML(Q) Process	34
3.1	Overall steps of the systematic literature review follow-	
	ing the PRISMA statement	40
3.2	Results of the metadata analysis	43
5.1	Flow chart of the DD-EFMod method for data-driven	
	automated parameterization of energy flexibility models	62
5.2	Data preparation step	69
5.3	Energy flexibility model setup	71
5.4	Energy flexibility key figure determination sub-process .	73
5.5	Energy flexibility model synthesis sub-process	75
5.6	Software concept for prototype implementation $\ldots \ldots$	77
6.1	Production process chain of the ETA Research Factory .	82
6.2	Flexible loads of the use case	83
6.3	Use case process and systems	84

7.1	System of scenario 1	90
7.2	Electrical power consumption of EMAG GT	91
7.3	Preprocessed electrical power consumption of EMAG GT	99
7.4	K-means clustering results EMAG GT	101
7.5	Boxplot of the clusters formed for the EMAG GT	102
7.6	Resulting flexibility space of EMAG GT	106
7.7	System of scenario 2	109
7.8	Resulting flexibility space of EMAG GT and product	
	storage	114
7.9	System of scenario 3	116
7.10	Measured data of power consumption from cooling machine	e120
7.11	Measured data of temperature from cold storage	122
7.12	Resulting flexibility space of eChiller45 and the cold stor-	
	age	124
7.13	System of scenario 4	126
7.14	Power consumption of the flexible loads EMAG GT and	
	eChiller45 and the cooling temperature for EMAG GT .	127
7.15	Lagged correlation of eChiller45 and EMAG GT $~$	131
7.16	Combined flexibility space of eChiller45 and EMAG GT	133
7.17	Energy prices on 31.08.2023	135
7.18	Electrical energy consumption and costs for reference	
	case operation	135
7.19	Electrical energy consumption and costs for energy-flexibili	ty
	measure change processing sequence	136
7.20	eq:electrical energy consumption and costs for energy-flexibility of the energy of the ener	ty
	measure shift start of job on Day-Ahead market	138
7.21	Electrical energy consumption and costs on Intraday mar-	
	ket for energy-flexibility measure <i>shift start of job</i>	139
A.1	Mini-batch K-means clustering results EMAG GT $~$	157
A.2	CluStream clustering results EMAG GT \hdots	158
A.3	Batch K-means clustering Results EMAG GT simulated	
	data set \ldots	158

A.4	Mini-batch K-means clustering results EMAG GT sim-	
	ulated data set \ldots	159
A.5	CluStream clustering results EMAG GT simulated data	
	set	159

List of Tables

2.1	Classification of energy-flexibility measures	12
2.2	Description of the key figures in the EFDM of class flex-	
	ible loads	16
2.3	Description of the key figures in the EFDM of class storages	16
2.4	Description of the key figures in the EFDM of class de- pendencies	17
25	Description of the key figures in the EFDM of a flexible	11
2.0	load measure	18
3.1	Search strings for each dimension of the systematic lit-	
	erature review	42
3.2	Results of the metadata search of the systematic litera-	
	ture review \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots	44
3.3	Relevant reports as results of the systematic literature	
	review	51
4.1	Summary of requirements and prerequisites	58
6.1	Software and hardware used for implementation of the	
	software concept	85
7.1	Technical key figures in the EFDM for describing the	
	flexibility space of scenario 1	92
7.2	Priority of measured values of the EMAG GT for sce-	
	nario 1	93
7.3	Results of data verification and preparation of scenario 1	99

7.4	EMAG GT K-means clustering statistical results $\ . \ . \ .$	101
7.5	Quality measures of batch K-means clustering	102
7.6	Results of batch K-means clustering $\ldots \ldots \ldots \ldots$	103
7.7	Identified holding durations of EMAG GT	104
7.8	Identified power gradients of EMAG GT	104
7.9	Automated determined key figures of the EFDM param- eterized via the DD-EFMod method for the EMAG GT	105
7.10	Comparison of clustering algorithms performance relat- ing to measured data set	107
7.11	Comparison of clustering algorithms performance relat-	
	ing to simulated data set	108
7.12	Technical key figures in the EFDM for describing the flexibility space of scenario 2	110
7.13	Priority of measured values of the EMAG GT for sce-	
	nario 2	111
7.14	General conditions for the product storage	111
7.15	Automated determined key figures of the EFDM pa- rameterized via the DD-EFMod method for the product	
	storage	113
7.16	Technical key figures in the EFDM for describing the	
	flexibility space of scenario 3	117
7.17	Priority of measured values of the eChiller45	118
7.18	General boundary conditions for the cold storage	118
7.19	Automated determined key figures of the EFDM param- eterized via the DD-EFMod method for the cooling ma-	
	chine eChiller45	123
7.20	Automated determined key figures of the EFDM param-	
	eterized via the DD-EFMod method for the cold storage	123
7.21	Technical key figures in the EFDM for describing the flexibility space of scenario 4	128
7.22	Correlation matrix of flexible loads of scenario 4	130

7.23	Automated determined key figures of the EFDM param-	
	eterized via the DD-EFMod method for the dependency	
	between two flexible loads $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	132
7.24	Flexible load measure for <i>change processing sequence</i>	136
7.25	Flexible load measure for <i>shift start of job</i>	138
7.26	Fulfillment of the requirements for the method and im-	
	plementation as part of the validation $\ . \ . \ . \ .$.	142
A.1	Detailed description of the class flexible load of the EFDM	
	(mandatory)	160
A.2	Detailed description of the class storage of the EFDM	
	(optional)	163
A.3	Detailed description of the class dependency of the EFDM	
	(optional)	164
A.4	Detailed description of the class flexible load measure	
	(FLM) of the EFDM (mandatory)	166

List of Abbreviations

Notation	Description
CIRP	The International Academy for Production Engi-
	neering
CRISP-DM	Cross-Industry Standard Process for Data Mining
CRISP-ML(Q)	Cross-Industry Standard Process model for the de-
	velopment of Machine Learning applications with
	Quality assurance methodology
CSV	Comma-separated values
DD-EFMod	Data-Driven Energy Flexibility Modeling
DRM	Design Research Methodology
EER	Energy Efficiency Ratio
\mathbf{EF}	Energy Flexibility
EFDM	Energy Flexibility Data Model
EFKF	Energy Flexibility Key Figure
EFM	Energy-Flexibility Measure
EMS	Energy Management System
EPEX	European Power Exchange
ERP	Enterprise-Resource-Planning
FL	Flexible Load
FLM	Flexible Load Measure
FR-EF	Functional Requirements for Energy Flexibility
	Modeling
FR-M	Functional Requirements on the Method
GUI	Graphical user interface
ICT	Information and Communication Technology
IEEE	Institute of Electrical and Electronics Engineers

Notation	Description
ISO	International Organization for Standardization
JSON	JavaScript Object Notation
ML	Machine Learning
OPC UA	Open Platform Communications Unified Architec-
	ture
PDF	Portable Document Format
PLC	Programmable Logic Controller
PPC	Production Planning and Control
\mathbf{PR}	Prerequisites
PRISMA	Preferred Reporting Items for Systematic Reviews
	and Meta-Analyses
QR	Quality Requirements
RAMI 4.0	Reference Architecture Model Industry 4.0
UML	Unified Modeling Language
VDI	Association of German Engineers e.V.

List of Symbols

\mathbf{Symbol}	Description	Unit		
Dependencies				
$A_{\rm C}$	Applicability conditions	_		
$T_{\rm App}$	Applicability duration	\mathbf{S}		
$ID_{\rm Dep}$	Dependency identification (ID)	_		
Typ_{Log}	Logical type	_		
ID_{Tar}	Target flexible load	_		
Typ_{Temp}	Temporal type	_		
ID_{Tri}	Trigger flexible load	—		
Flexibilit	y Space			
\mathfrak{D}	Degree of dependency of flexible loads	_		
\mathbf{A}	Matrix of dependencies between flexible	_		
	loads			
\mathcal{K}_i	Individual EFKF	_		
Κ	Vector of relevant EFKF	_		
x_n	Raw data input	_		
X	Vector of raw data input	_		
${\mathcal D}$	Flexibility space for dependency	_		
${\cal F}$	Flexibility space	_		
${\cal L}$	Flexibility space for flexible load	_		
S	Flexibility space for storage	_		
Flexible Loads				
$N_{ m Mod}$	Modulation number	_		
$N_{\rm Use}$	Usage number	_		
∇P_{Act}	Activation gradient	$\frac{kW}{s}$		

\mathbf{Symbol}	Description	\mathbf{Unit}
∇P_{Dea}	Deactivation gradient	kW s
∇P_{Mod}	Modulation gradient	$\frac{kW}{s}$
$P_{\rm S}$	Power state sequence	_
Р	Vector of all power states of the ID_{Load}	_
	specific flexible load	
$T_{\rm H}$	Holding duration	s
$T_{\rm V}$	Validity	s
$T_{\rm Rea}$	Reaction duration	s
$T_{\rm Reg}$	Regeneration duration	\mathbf{S}
c_{Load}	The costs associated with the use of flexible	€
	load	
$ID_{\rm Load}$	Flexible load identification (ID)	_
$L_{\rm Load}$	Metering point designation for the geo-	€
	graphical and power grid topological loca-	
	tion of a flexible load	
p	Price of the flexible load to be achieved on	€
	the market as a minimum	
Flexible 1	Load Measure	
$T_{\rm Act}$	Activation duration	s
$T_{\rm D}$	Delivery time	\mathbf{S}
T_{Mod}	Modulation duration	s
$\mathbf{P}_{\mathrm{FLM}}$	Load change profile	(kW,s)
$ID_{\rm FLM}$	Flexible load measure ID	_
r	Reward received by a company for the im-	€
	plementation of a flexible load measure	
t_0	Start time of a flexible load measure	s
$t_{\rm end}$	End time of a flexible load measure	\mathbf{S}
$t_{\rm OCD}$	Order confirmation deadline	\mathbf{S}
$t_{\rm V,end}$	End of the validity $T_{\rm V}$	S
$t_{\rm V,start}$	Begin of the validity $T_{\rm V}$	S
Math and	1 Statistics	
\mathcal{A}	Algorithm	_

\mathbf{Symbol}	Description	\mathbf{Unit}
W	Width of bin	_
b	Boundary	_
В	Model output	_
R_{ij}	Cluster similarity	_
\mathbf{c}'	Recalculated cluster center	_
c_i	Cluster center	_
\mathbf{C}	Vector of cluster centers	_
cov	Covarianz	_
р	Vector of data samples	_
p	Single data samples	_
D	Data set	_
DB	Davies-Bouldin index	_
d	Distance function	_
η	Learning rate	_
LT	Sum of the timestamp of the last update	_
LS	Linear sum vector of the data points	—
μ	Mean value	_
Y	Model output	_
\mathcal{M}'	Resulting model	—
\mathcal{M}	Model	_
\mathbb{N}	Set of natural numbers	_
ho	Pearson's correlation coefficient	_
s	Silhouette coefficient	_
ST	Squared sum of the timestamp of the last	_
	update	
SS	Squared sum vector of the data points	_
σ	Standard deviation	_
au	Lag of time	s
k	Discrete time	s
x	Time signal	_
t	Continual time	s
M	Window size	_

\mathbf{Symbol}	Description	\mathbf{Unit}
\mathbf{x}_{MA}	Moving average signal	_
$x_{\operatorname{norm},i}$	Normalization of signal x at timestep i	_
$x_{\mathrm{std},i}$	Standardiszation of signal x at timestep i	_
Measured	l Raw Data	
Ι	Electric current	А
$T_{\rm Intraday}$	Cycle time at intraday market	\mathbf{S}
$t_{\rm cycle}$	Cycle time	\mathbf{S}
P	Electric power	W
f	Frequency	Hz
$T_{\rm pro}$	Frequency	$^{\circ}\mathrm{C}$
$t_{\rm pro}$	Processing time	\mathbf{S}
$t_{\rm shift}$	Shift time	\mathbf{S}
$s_{ m da}$	Machine state - disabled	binary
$s_{ m op}$	Machine state - operational	binary
$s_{ m st}$	Machine state - standby	binary
$s_{ m wk}$	Machine state - working	binary
U	Electric voltage	V
Storages		
$C_{\rm S}$	Usable capacity	$\rm kWh$
c_{Stor}	Cost for operation of flexible storage	€
E_{Drain}	Energy drain from storage	$(s \rightarrow kW)$
E_{to}	Initial energy content at the start time of	(kWh,s)
	the validity of the associated flexible load	
$E_{\rm Loss}$	Energy loss	<u>%</u>
E_{Tar}	Target energy content at the end time of	(kWh,s)
	the validity of the associated flexible load	
ID_{Stor}	Storage identification (ID)	_
$S_{ m S}$	Energy supply to storage	(-,-)

1 Introduction

"Great things are done by a series of small things brought together."

Vincent Van Gogh

This chapter describes the motivation of this thesis and sets it into the current social and economic context in Section 1.1. Based on this, the problem is discussed in Section 1.2, from which the main contribution of this thesis is derived. The chapter concludes in Section 1.3 with an overview of the structure of this thesis based on the used research methodology.

1.1 Motivation

Climate change is one of the greatest challenges of our time and affects all levels of society and economy. This was recognized by the international community, so that in 2015 the Paris Climate Agreement was adopted to counteract this challenge [1]. The European Union has committed itself to become climate neutral by 2050 within the framework of the Green Deal [2]. As a member state of the European Union, the German government has set itself the ambitious target of achieving this goal as early as 2045 [3]. To achieve this goal, the energy transition is one of the biggest levers and goals for the future of the German economy [4, 5, 6]. The measures adopted as part of the national climate protection plan and the energy transition include the strong expansion of renewable energies. As a result, around 55 percent of electricity demand was met from non-fossil energy sources in 2022 (see Figure 1.1b) [7]. This increasing feed-in of energy from renewable energy producers into the energy grids increases the volatility of the energy supply. This, on the one hand, and the additional acceleration of the energy crisis due to the Ukraine conflict since 2022, on the other hand, have caused extremely strong fluctuations in international and national energy prices (see Figure 1.1c) [8]. These effects are particularly relevant for industry, which accounts for the majority of energy demand in Germany at around 43 percent (see Figure 1.1a) [9]. To solve these energy challenges, digitalization of industry plays a central role [10]. On the one hand, production processes can be better monitored, and equally, equally, more targeted control of production is also possible. This also applies to the energy supply and energy management of a company. On the one hand, demand response measures can better ensure the stability of the power grids, and on the other, production can be adapted to the volatile supply of electricity. Such adaptability of an industrial company, also called energy flexibility, is a central aspect of future energy supply [11]. This thesis is framed in this context.



Figure 1.1: Overview of the energy market in Germany. Figure (a) shows net electricity consumption by sector in 2021 [9]. Figure (b) shows the primary energy generated and the percentage share of renewable energy in it [7]. Figure (c) shows the development of the wholesale electricity price and its increasing volatility [8]. (own figures)

1.2 General Objective and Contribution

In order to apply energy flexibility successfully in industry, it has been researched in the SynErgie research project since 2016 with a focus on its applicability in industry [12]. For this purpose, various methods were developed within the project to identify and exemplarily realize industrial energy flexibilities [5]. Due to the high complexity of industrial production, the successful application of industrial energy flexibility is associated with high obstacles. Some of the obstacles identified include the following [12, p. 234]:

- product quality must be ensured
- scheduling of the production in terms of time and quantity
- total costs should be further reduced
- production systems are complex systems with different dependencies

Taking these aspects into account, a company must also operate profitably and efficiently. This increasingly includes taking energy costs into account, especially due to the increasing volatility of energy prices. Industrial energy flexibility offers a solution to this problem. Through the targeted use of energy flexibility, it is possible to adapt a production process to fluctuating electricity prices while maintaining the same level of productivity. Thus, on the one hand energy costs can be saved and on the other hand the CO_2 footprint of a production can be reduced by increased use of renewable energies [5]. In addition, flexibilities can be traded in a future energy system, which can generate additional revenues. To enable a factory to become energy-flexible, the process shown in Figure 1.2 was developed [13]. This consists of the steps potential analysis, conceptualization and planning, application and implementation, operative flexibility marketing, controling and monitoring and optimization. Each step of the process is explained in detail in Section 2.1.



Figure 1.2: Methodology for energy flexibilization of factories (own figure based on [13]).

To implement the entire process, it is necessary to describe the flexibilities identified in the potential analysis by a standardized data model in step three. Such a model for describing energy flexibility was developed in the SynErgy project. It allows the generic description of energy flexibility [14]. This model has been designed to be applied to any technical plant or storage facility in order to realize the highest possible flexibility potential. Since industrial processes are complex and contain many different assets and dependencies, it is also difficult to create such a model for each relevant asset of a production system. In addition, a high degree of expert knowledge and system understanding is required. This is the starting point of this thesis.

The main contribution of this work is the development of the Data-Driven Energy Flexibility Modeling (DD-EFMod) method, which simplifies the parameterization of the Energy Flexibility Data Models. The following steps are applied:

- Development of the DD-EFMod method. To achieve this, the methodology for the energetic flexibilization of factories is extended in the implementation step with a focus on the parameterization of Energy Flexibility Data Models. For this purpose, the Cross-Industry Standard Process model for the development of Machine Learning applications with Quality assurance methodology (CRISP-ML(Q)) approach is adapted and transferred accordingly to this parameterization.
- 2. Development of a software concept. In order to ensure the applicability of the DD-EFMod method, a software concept is devel-

oped, whereby the implementation of the method corresponds to a standardized procedure.

- 3. To demonstrate the practical application of the DD-EFMod method, this is validated using a use case with four different scenarios:
 - a) The first scenario comprises the modeling and parameterization of the Energy Flexibility Data Model for a machine tool.
 - b) In the second scenario, the Energy Flexibility Data Model for the machine tool is extended to include the data model of the associated product storage system.
 - c) In the third scenario, the system boundary is changed so that the cooling water supply of the machine tool is considered and the associated Energy Flexibility Data Model is parameterized.
 - d) The fourth scenario focuses on the dependencies between the systems of the previous scenarios and parameterizes the associated class of the Energy Flexibility Data Model.
- 4. The validation of the method concludes with an examination of the economic potential, showing that the detailed modeling of energy flexibility in the Energy Flexibility Data Model offers advantages. To this end, the energy-flexibility measures *change of processing sequence* and *shift start of job* are compared and the costs are considered with regard to the Day-Ahead and Intraday electricity market.

1.3 Research Methodology and Structure of Work

To achieve the overarching goal, this research work is based on the Design Research Methodology (DRM). [15]. This approach is suitable for achieving the objective, as it involves the following four phases as an iterative process:

According to the first phase (Research Clarification), the research field and the underlying question are identified based on literature and experience, and the overarching research objective is formulated (Chapter 1) [15].

Based on this, a fundamental understanding of the state of the art in science and technology is established in the second phase (Descriptive Study I) (Chapter 2 and 3) [15]. This work presents the fundamentals of industrial energy systems and energy flexibility in interaction with digital production. Furthermore, existing approaches to data analysis and modeling of energy flexibility are presented, and the research deficit is identified through literature research and analysis.

This work focuses on the third phase (Prescriptive Study) [15]. Based on the identified research gap, the research questions, hypotheses, and requirements for the solution method for data-driven automated modeling of energy flexibility are formulated in Chapter 4. This is followed by the development of the DD-EFMod method in Chapter 5, which provides a structured procedure for the automated modeling of energy flexibility.

In the fourth phase of the DRM (Descriptive Study II), the DD-EFMod is initially applied and validated to provide conclusions for further use in this research work [15]. To this end, the use case that forms the framework for the validation is first presented in Chapter 6. The method is then applied prototypically to various scenarios, and the results are discussed (Chapter 7). This thesis concludes with a conclusion, a summary, and an outlook in Chapter 8.

2 Fundamentals

"What I'm trying to do is to maximise the probability of the future being better."

Elon Musk
After defining the overall objective of this thesis in the previous chapter, this section delves into the relevant fundamentals for the development of the methodology for the automated parameterization of energy flexibility metrics. To this Section 2.1 first provides an introduction to the complexity of industrial energy systems. The focus is on the systemtheoretical connection between the production system and the different energy supply systems in the industrial environment. Furthermore, the importance of demand-side management and energy flexibility as well as the basics of modeling energy flexibility are shown. This is followed by relevant principles in the context of digital manufacturing in Section 2.2. Subsequently, Section 2.3 focuses on data analysis and machine learning in the context of the development and implementation of the method for DD-EFMod.

2.1 Industrial Energy Systems

Production systems are at the heart of modern industrial societies and are the driving force behind the production of goods and services that characterize our everyday lives. A production system is defined in the context of this work according to:

```
Definition 2.1: Production System [16, p. 2]
```

A production system is a collection of people, equipment, and procedures organized to perform the manufacturing operations of a company.

A crucial aspect that determines the performance and sustainability of production systems is their close connection with industrial energy systems. Industrial energy systems supply production systems with the required energy and thus play a key role in ensuring operational efficiency and minimizing environmental impacts [17, pp. 7–9]. Optimizing these systems is therefore crucial to overcoming the challenges of energy efficiency and environmental sustainability in production [18, pp. 15–25]. In the face of global challenges such as climate change and resource scarcity, there is an increasing focus on research in the field of industrial energy systems to develop innovative solutions for a reliable, cost-efficient, and environmentally friendly energy supply [19]. One advanced strategy that is playing an increasingly important role in this context is demand-side management and the use of industrial energy flexibility. By specifically controlling energy consumption depending on the availability and cost of energy, companies can significantly reduce their operating costs while contributing to stabilizing the power grid and integrating renewable energy [20]. Therefore, the research and application of demand-side management and energy flexibility concepts in industrial energy systems plays a key role in the design of sustainable and efficient production systems [21].

2.1.1 Demand-Side Management and Industrial Energy Flexibility

To get the possibility of adapt the power consumption of a consumer, the concept of demand-side management was defined by the American Electric Power Research Institute in 1992 [22]. Based on this the term demand response was defined in 2006 and is used in this work as:

```
Definition 2.2: Demand Response [23]
```

Demand response are the changes in electric usage by end-use customers from their normal consumption patterns in response to changes in the price of electricity over time, or to incentive payments designed to induce lower electricity use at times of high wholesale market prices or when system reliability is jeopardized.

For the realization of industrial demand response the Association of German Engineers e.V. (VDI) defines energy flexibility in 2020 as a capability:

Definition 2.3: Energy Flexibility [24]

Energy Flexibility (EF) is the ability of a production system to adapt quickly and in a process-efficient way to changes in the energy market.

When considering energy flexibility, a distinction is made between different energy-flexibility potentials [24] as follows:

- Theoretical potential is the mathematical factor determined by the rated power of all forms of final energy.
- The technical potential is the possibility of varying the power requirement within the scope of technological conditions, particularly system-specific conditions that influence power consumption and single time periods.
- Economic potential is the proportion of the technical potential that can be used commercially.
- Practical potential is the subset of the technical potential which takes into account factors such as regulatory and administrative obstacles.
- Realisable potential is the intersection of economic potential and practical potential.

Figure 2.1 shows the relationships between the different potentials. To leverage energy-flexibility potential, appropriate measures are required.



Figure 2.1: Spaces of the energy-flexibility potentials (own figure based on [12]).

These are referred to as energy-flexibility measures.

Definition 2.4: Energy-Flexibility Measure [24]

An Energy-Flexibility Measure (EFM) is a deliberate action taken to implement a defined change of state in a production system and encompasses the change in state of a production station and the interactions in the production system which this change entails.

Based on the definitions multiple measures can be used to achieve the adjustments. Table 2.1 overviews those various measures. A production system must be enabled for such an energy-flexible operation to implement a measure. A corresponding procedure is shown by the VDI in [13]. The individual steps of the enablement process are shown in Figure 1.2 and discussed below.

Table 2.1: Classification of energy-flexibility measures by duration [24] and by levels of the automation pyramid defined according to [25] (own table).

Automation level	Implementable energy- flexibility measure	Duration
Level 4: Business planning and logistics (Business level)	 Shift production start Shift break times Change production sequence Adjust shift times Adjust capacity planning 	long-term
Level 3: Manufacturing operations and	 Interrupt job Store energy Shift start of job 	week
control (Manufacturing level)	 Change job sequence Adjust energy procurement Adjust resource allocation 	day
Level 2: Monitoring, supervision and	 Interrupt process Store energy (inherently) Change processing sequence 	hour
control (Control level)	 Operate with bivalent energy Adjust process parameters 	minute
Level 1:		

Sensing and actuation (Field level)

- 1. The initial step of the six-stage methodology focuses on a noninvasive and low-effort estimation of technical and economical energy flexibility potentials. First, promising systems are assessed for energy-flexible operation prerequisites, followed by an analysis of their technical flexibility potential [26]. Subsequently, the systems are evaluated for energy-flexible operation suitability within the production ecosystem, considering factors like control type, process relevance, and temporal decoupling.
- 2. In the second step of design and planning, the previously prioritized energy-flexibility measures are examined in detail, data (e.g. load profiles or individual plant parameters) are collected, and measurements (e.g, following [27]) are carried out if required. On this basis, the economic efficiency of individual energy-flexibility measures are determined. Subsequently, implementation planning is done for selected EFMs.
- 3. The third step is the actual implementation of promising energyflexibility measures and an energy-flexible operation of the factory. The technical enablement of individual systems must be carried out holistically, considering aspects of hardware, information and communications technology, and production planning and control. The energy-flexible operation is to be validated based on test runs, analyses, and safety checks.
- 4. After the successful realization of the energy-flexible operation, the operational marketing of the flexibility can be started in step four (cf. [24]).
- 5. During and after the implementation and marketing of flexibility, the controlling and monitoring phase analyzes the degree and quality of performance. This can be done with the help of energy management systems to identify anomalies and potential for improvement and to derive appropriate action steps.
- 6. With the help of the knowledge gained, further improvements in energy flexibility are possible in the step of operational optimization. Data-based analyses or simulations can further increase the

utilization of various energy-flexibility measures. The theoretical potential identified in this way can then be incorporated into a further iteration of the design and planning, thereby continuously optimizing energy-flexible operation.

Specific technical and market-related requirements are necessary to effectively and safely integrate flexibility into the future energy system based on this six-step process. A comprehensive implementation of energy flexibility requires automated marketing processes based on standardized procedures, often implemented through platform solutions at both company and market levels. In addition, intensive use of information and communication technologies is required, both within companies and across company boundaries. This makes it necessary to model the existing energy flexibility accordingly to ensure such an information exchange.

2.1.2 Modeling of Energy Flexibility

In order to describe energy flexibility, appropriate modeling is required. A method for modeling and description of energy flexibility is given by [14], which was published in 2019 and has been gradually developed since then [28, 29, 30]. This method allows a generic description of



Figure 2.2: Classes of the Energy Flexibility Data Model that span the flexibility space, which can contain various flexible load measures [14] (own figure).

energy flexibility and was developed while research of the "SynErgie" project. Thus, any technical system can theoretically be modeled accordingly. The advantage compared to other modeling approaches is the consideration of technical properties as well as market-side requirements for potential trading on the electricity market by corresponding suppliers. Since 2019, this original model has been further developed and adapted to achieve a higher degree of applicability in industry, especially using the json schema of the EFDM developed in [29]. This Energy Flexibility Data Model (EFDM) is composed of four classes that span the flexibility space, which can contain various flexible load measures (see Figure 2.2). The first class are the Flexible Load (FL).

Definition 2.5: Flexible Load [28]

A Flexible Load (FL) models a technical system or the interaction of different technical systems that have the potential to produce a change in performance.

This class flexible load is described by the key figures shown in Table 2.2. To increase the energy-flexibility potential, flexible loads can interact with one or more storages. For this purpose, storage facilities are defined as follows and the key figures are shown in Table 2.3.

Definition 2.6: Energy Storage [28]

An energy storage system is a technical system or the interaction of different technical systems that have the potential to store energy.

Key Figure	Description
$ID_{\rm Load}$	Flexible load ID
$T_{ m V}$	Validity
Р	Vector of power states
$T_{ m H}$	Holding duration
$T_{ m Rea}$	Reaction duration
$P_{ m S}$	Power state sequence
$ abla P_{ m Act}$	Activation gradient
$ abla P_{ m Mod}$	Modulation gradient
$ abla P_{ ext{Dea}}$	Deactivation gradient
$T_{ m Reg}$	Regeneration duration
$N_{\rm Use}$	Usage number
$N_{ m Mod}$	Modulation number
$c_{ m Load}$	Costs for usage of the flexible load
p	Price for the flexible load
$t_{ m OCD}$	Order conformation deadline
$L_{\rm Load}$	Metering point designation of flexible load

Table 2.2: Description of the key figures in the EFDM of class flexible loads, following [29].

Table 2.3: Description of the key figures in the EFDM of class storages, following [29].

Key Figure	Description
ID_{Stor}	Storage ID
$C_{ m S}$	Usable capacity
$E_{ m to}$	Initial energy content at the start time of the
	validity
E_{Tar}	Target energy content
$E_{\rm Loss}$	Energy loss
$S_{ m S}$	Suppliers
E_{Drain}	Energy drain from storage
$c_{ m Stor}$	Cost for operation of flexible storage

As shown in Figure 2.2 the third class of the Energy Flexibility Data Model (EFDM) are dependencies. These class decribes the interaction of multiple flexible loads and is defined as follows.

Definition 2.7: Dependency [28]

Dependencies model the constraints and dependencies for the interaction of multiple flexible loads.

This means, for example, the use of one flexible load can imply or exclude the use of another flexible load. Therefore those dependencies musst be taken into account while the realization of an energy-flexibility measure.

Key Figure	Description
ID_{Dep}	Dependency ID Twigger florida logid
ID_{Tri} ID_{Tar}	Target flexible load
Typ_{Log}	Logical type
Typ_{Temp}	Temporal type
$T_{ m App} = A_{ m C}$	Applicability duration Applicability conditions

Table 2.4: Description of the key figures in the EFDM of class dependencies, following [29].

For a realization of an energy-flexibility measure on a production system based on the EFDM, a more specific description in change of power and how long this variation should be accordingly to all given contraints of the flexible loads, storages and dependencies. Therefore the class flexible load measure (FLM) is defined. The necessary key figures are given in Table 2.5.

Definition 2.8: Flexible Load Measure [28]

A flexible load measure (FLM) defines a concrete power change within the flexibility space in the form of a load change profile, takes into account the dependencies, has no degrees of freedom itself and is time scheduled.

/ L J		
Key Figure	Description	
$ID_{\rm FLM}$	Flexible load measure ID	
ID_{Load}	ID of the flexible load to which the flexible load	
	measure is directed.	
$\mathbf{P}_{ ext{FLM}}$	Vector of load change profile	
r	Reward	

Table 2.5: Description of the key figures in the EFDM of a flexible load measure, based on [29].

Not all energy flexibility key figures are mandatory for any system. Therefore, their necessity was defined in Lindner *et al.* [29] for each key figure. The classes storage and dependency are optional by definition, see Appendix A.3.

As seen in Figure 2.3 there are multiple periods and times which are necessary to describe and to realize a flexible load measure but not all of them are a key figure. The first relevant time is $t_{\rm OCD}$ which means the point in time until orders are accepted at energy markets [31]. This is following by the validity $T_{\rm V}$ which is given by the refering FL and starts with the timestamp $t_{\rm V,start}$ and terminates at the timestamp $t_{\rm V,end}$. Within the validity lies the flexible load measure (FLM) which spans from starting time t_0 , there the system gets the initial signal to activate the FLM, to the end time $t_{\rm end}$. The regeneration duration $T_{\rm Rea}$ describes a delay which a system could have until the first load change is detectable or realizable. For trading and usage of the FLM by a grid operator the delivery duration $T_{\rm D}$ is given by

$$T_{\rm D} = T_{\rm Act} + T_{\rm H} \tag{2.1}$$





or respectivly by

$$T_{\rm D} = T_{\rm Mod} + T_{\rm H} \tag{2.2}$$

and describes a active change in state of power. In Figure 2.3 the first delivery durations $T_{\rm D}^1$ which is calculated by equation 2.1 and $T_{\rm D}^2$ following equation 2.2. A delivery duration do not include a deactivation gradient $\nabla P_{\rm Dea}$ or a regeneration duration $T_{\rm Reg}$.

After the relevance and methodological approaches to modeling and implementing energy flexibility in industrial systems have been presented, it becomes clear that their effective implementation and optimization require deep integration into the digital processes of modern manufacturing. The topic of digital manufacturing is therefore explained in more detail below.

2.2 Digital Manufacturing

Digital manufacturing, also known as Industry 4.0, is described by [32] as the intelligent networking of machines and processes for the industry with information and communication technology. The integration of digital manufacturing processes is crucial in the context of energy flexibility as it enables the seamless implementation of energy flexibility measures, which are vital for optimizing energy demand and enhancing operational efficiency in industrial systems. This section explores how digital manufacturing technologies facilitate the realization of energy flexibility and adjust production processes to an energy-flexible operation. Some of the possibilities of Industry 4.0 are [33]:

- Flexible production: Networking facilitates coordination and planning in production by optimizing the involvement of different companies.
- Changeable factory: Production lines of the future will be modular and quickly configured for specific tasks, enabling the efficient and economical production of individual products.
- Optimized logistics: Intelligent algorithms improve the flow of materials and delivery routes and automatically report material requirements.
- Use of data: Analyzing production data enables process optimization and new data-based business models.

These aspects are characterized by the objective of achieving both optimization and flexibility in production through Industry 4.0, while at the same time supporting sustainable and economic goals. One of the central models for structuring and implementing Industrie 4.0 concepts is the Reference Architecture Model Industry 4.0 (RAMI 4.0) as shown in Figure 2.4 [34]. The model is divided into three main axes.

The hierarchy axis represents the different levels within a company, from the product level to the company level, and thus integrate all levels of the automation pyramid. This enables end-to-end networking and control of production processes. At the same time, the life cycle and value chain axis consider the entire life cycle of a product or plant, enabling a holistic view and optimization. The six structured layers on the vertical axis describe the decomposition of a machine into its properties.

Based on this, the RAMI 4.0 architecture offers a suitable structure for implementing energy flexibility. In particular, the possibilities of bidirectional data and information exchange across company levels and company boundaries to the connected world, e.g., to energy or flexibility markets, enable the description and implementation of energy flexibility [35], as in the validation use case of this work in Chapters 6 and 7. In order to automate and standardize the entire process of energy flexibility trading from the machine to marketing, integrating energy-flexibility measures into the production planning and control of the company is essential [36]. The Energy Synchronization Platform [37] addresses precisely this goal. The Energy Synchronization



Figure 2.4: Reference Architecture Model Industry 4.0 (based on [34]).

Platform comprises framework conditions, interfaces, data models for describing energy flexibility (see Section 2.1.2), stakeholders, and security aspects and maps the entire process of automated energy flexibility trading from the machine to trading itself [38] (see Appendix A.1.2).

Modeling and simulation of machines and machine operation play a crucial role in digital manufacturing and implement energy-flexibility measures. They allow the virtual representation of production processes to test and validate optimizations in advance, for example, concerning energy efficiency or adaptability to fluctuating energy prices [39]. A simulation is defined as follows:

Definition 2.9: Simulation [40, p.3]

Simulation is the imitation of the operation of a real-world process or system over time. Simulation involves the generation of an artificial history of the system and the observation of that artificial history to draw inferences concerning the operating characteristics of the real system that is represented.

Using advanced software and algorithms, complex systems can be modeled and simulated in real-time, enabling precise prediction and planning of energy consumption and production [39]. Using those approaches different models are necessary. A model is generally defined as:

Definition 2.10: Model [41, p.13]

A model is defined as a simplification representation of a system for the purpose of studying that system.

The creation of such a model is called model specification and is defined as

Definition 2.11: Model Specification [42, p.3]

Model specification is the exercise of formally stating a model and involves formulating a statement about a set of parameters. In the context of this work and in particular for enabling energyflexible operation, the energy flexibility model described in Section 2.1.2 is such a representation and specification. The defined key figures to describe the energy flexibility must be filled with values to describe the energy flexibility of a system. This important step of describing one or more values for each variable quantity for each key figure of the model is called model parameterization and is defined as:

Definition 2.12: Model Parameterization [43]

Adaptation of a system, technical device, model, or software by specifying one or more values for each variable quantity.

This step for determining the key figures of the Energy Flexibility Data Model is described in Section 5.1.

Digital manufacturing refers to the comprehensive use of data in real-time to monitor and optimize the entire production process [44, pp. 181–183]. Data-based and data-driven approaches represent different degrees of integration and use of data in manufacturing. While data-based approaches use data at the beginning of a process to set parameters or workflows, data-driven approaches take a more dynamic role by continuously collecting and analyzing data to optimize processes continuously. A distinction must be made between the two approaches. Data-based is defined as:

Definition 2.13: Data-Based [45]

Data-based means the process is an open loop, and only the starting point uses data.

The term data-driven encompasses this term more broadly and is defined as:

Definition 2.14: Data-Driven [45]

Data-driven means that the process is a closed loop and its starting point and destination are both data. In the context of digital manufacturing, these approaches enable precise monitoring and control of machines and systems, leading to more efficient and flexible processes. By implementing closed-loop systems, it is possible to react automatically to changes in the production process and operate flexible [38]. Using real-time data analysis and adaptive systems, it is also possible to parameterize models of the production system automatically. This is possible with the method developed in in this work for the Energy Flexibility Data Model. Data analytics approaches and fundamentals, which are used while implement the method for DD-EFMod, are explained in the following.

2.3 Data Analytics and Machine Learning

In order to implement data-driven approaches, methods from the fields of data analytics and machine learning play a central role. The aspects relevant to this work are examined in more detail below. First, methods of correlation analysis are discussed, as these methods are used to determine correlations between variables, which is a key aspect of the parameterization of energy flexibility. Secondly, this section shows which different clustering algorithms, such as those implemented in the DD-EFMod method, exist and how they can be validated. This is followed by an explanation of the CRISP-ML(Q) methodology, whose individual steps are adapted during the development and implementation of the DD-EFMod.

2.3.1 Data Analytics

In particular, correlation analysis is a common tool for identifying and interpreting relationships between different variables within large data sets. By calculating the correlation coefficient, insights can be gained into the strength and direction of the relationship between two variables. For this purpose, the correlation between variables can be described using the correlation coefficient ρ . The greater the amount ρ of the correlation coefficient, the more dependent two measured variables are and is in gerally defined according to [46, p. 821]:

Definition 2.15: Pearson Correlation Coefficient [46, p. 823] Pearson's correlation coefficient

$$\rho_{X,Y} = \frac{\operatorname{cov}(X,Y)}{\sigma_X \sigma_Y} \tag{2.3}$$

with the covariance of the two variables cov(X, Y) divided by the product of their standard deviations σ .

A correlation coefficient of +1 means a perfect positive correlation, -1 a perfect negative correlation and 0 no correlation. This is followed by enhancing the informational value of the dataset. This involves deriving auxiliary variables and increasing the density of useful information within the data. In a more specific way for time series which are necessary for an energy-flexible operation, like power consumption or temperature data, correlation is calculated by

$$\rho = \frac{\sum_{i=1}^{n} (x_{1,i} - \mu(x_1)) (x_{2,i} - \mu(x_2))}{\sqrt{\sum_{i=1}^{n} (x_{1,i} - \mu(x_1))^2} \sqrt{\sum_{i=1}^{n} (x_{2,i} - \mu(x_2))^2}}$$
(2.4)

with $x_{1,i}$ and $x_{2,i}$ as values of the relvant signals $\mathbf{x}_1(t)$ and $\mathbf{x}_2(t)$ at the time *i* with *n* samples [47, p. 384]. In addition, for a deeper look, lagged correlation could be applied when there is a time delay in the relationship between two signals. The lagged correlation is calculated with a time shift τ for one of the signals by

$$\rho(\tau) = \frac{\sum_{i=1}^{n-\tau} \left(x_{1,i+\tau} - \mu(x_1) \right) \left(x_{2,i} - \mu(x_2) \right)}{\sqrt{\sum_{i=1}^{n-\tau} \left(x_{1,i+\tau} - \mu(x_1) \right)^2} \sqrt{\sum_{i=1}^{n} \left(x_{2,i} - \mu(x_2) \right)^2}}$$
(2.5)

by shifting the signal $\mathbf{x}_1(t)$ by τ points in time and then correlating it with $\mathbf{x}_2(t)$. A lagged correlation can be used to investigate whether there is a time lag in the relationship between two signals. A high lagged correlation value at a certain lag τ could indicate that one signal influences the other with this time delay.

However, in order to determine meaningful correlations, it is often necessary to standardize and normalize the data. Normalization for better scaling according to

$$x_{\operatorname{norm},i}(t) = \frac{x_i(t) - \min(\mathbf{x}(t))}{\max(\mathbf{x}(t)) - \min(\mathbf{x}(t))}$$
(2.6)

should be done [48, p. 110]. Normalization aims to scale the data so that it lies within a specific range, typically between 0 and 1. In addition signal standardization transforms the data so that they have a mean value of 0 and a standard deviation of 1, with the calculation following

$$x_{\text{std},i}(t) = \frac{x_i(t) - \mu(\mathbf{x}(t))}{\sigma(\mathbf{x}(t))}$$
(2.7)

with $\mu(\mathbf{x}(t))$ as mean of the input signal and $\sigma(\mathbf{x}(t))$ the standard deviation [48, p. 113]. These steps are crucial to make data sets comparable and to improve the efficiency of data processing, especially if the data comes from different sources or is available in different units.

Another important step in data analysis and the preparation of data sets for machine learning processes is quantization. This process reduces the complexity of the data by dividing continuous variables or time series into discrete intervals. Quantization not only simplifies the data structure, but can also help to reduce the risk of overfitting and increase the computational efficiency of algorithms [49, p. 115] [50, p. 290]. Let a data set D with N values and a value range from min(D)to max(D). The data set is to be divided into k bins, then according to [51, p. 332] the width W of each bin is given by

$$W = \frac{\max(D) - \min(D)}{k} \tag{2.8}$$

and its boundaries as

$$b_i = \min(D) + i \cdot W \tag{2.9}$$

for i = 0, 1, ..., k. Thus, the assignment of the data to the bins follows with

$$\operatorname{Bin}(x) = \left\lfloor \frac{x - \min(D)}{W} \right\rfloor$$
(2.10)

where $\lfloor \cdot \rfloor$ is the rounding function. Such a quantification method can be further used to calculate the moving average of a signal according to

$$\mathbf{x}_{\mathrm{MA}}(t_k) = \frac{1}{M} \sum_{i=k-\lfloor M/2 \rfloor}^{k+\lfloor M/2 \rfloor} \mathbf{x}(t_i)$$
(2.11)

with a window of size M|M < N where N is the number of signal samples under consideration [52].

Another option for data and signal analysis is Boolean logic analysis, which enables the mathematical treatment of binary values (1 for "true" and 0 for "false"). This analysis uses fundamental logical operations [53]:

- Conjunction: $A \wedge B$ is true if both A and B are true.
- Disjunction: A ∨ B is true if at least one of the statements A or B is true.
- Negation: $\neg A$ reverses the truth value of A; true becomes false and vice versa.

In data analysis and processing, Boolean logic enables data sets to be filtered using defined conditions that specify specific criteria for data selection or processing.

The presented data analysis methods are often used as pre-processing steps for other data processing algorithms, like the implementation of the DD-EFMod method. Such pre-processing steps are exceptionally fundamental for machine learning algorithms.



Figure 2.5: Subdivision of machine learning methods into online and offline and the subgroups incremental learning and batch learning based on [57] (own figure).

2.3.2 Machine Learning and Data Stream Mining

A machine learning algorithm is a process or method for recognizing patterns in data and learning from them [54]. It is essentially a process or set of rules applied to a data set to achieve a specific goal, such as classification or clustering. Examples of machine learning algorithms include Decision Trees, K-Nearest Neighbors, clustering algorithms such as K-means, and network algorithms such as Neural Networks [55, pp. 2–4]. In the context of production systems it is relevant to consider offline algorithms for historical or storaged data and also consider online algorithms for real-time data streams with only temporary availability to facilitate immediate operational adjustments and real-time decision making [56].

A machine learning model is the resulting product of a machine learning algorithm applied to a data set. After the algorithm has been trained, the model represents the knowledge learned, it is basically a specific representation of what has been learned from the data. The model can then be used to make predictions or decisions based on new, unknown data [51, p. 7].

Furthermore, a distinction must be made between offline and online

algorithms when using machine learning. The following options must be differentiated here, as shown in Figure 2.5 while following [57]:

- Online learning: Refers to learning processes in which the model is continuously updated as soon as new data is available. It is ideal for applications where data must be processed in real-time.
- Offline learning: Also known as batch learning, it refers to the situation where the learning model is trained with a fixed data set. New data leads to a retraining of the model.
- **Batch learning**: A special form of offline learning in which large amounts of data are processed in batches to train or update the model.
- **Incremental learning**: In contrast to batch learning, this method updates the model step by step by continuously processing smaller amounts of data.
- **Instance incremental**: A form of incremental learning where each individual instance is used to incrementally update the model.
- **Batch incremental**: This uses small batches of data to incrementally update the model instead of retraining it with a full set of data.
- Full batch learning: A method where the entire available dataset is processed in a single batch, often used in the initial training phase.
- Sequential learning: Focuses on sequential processing of data, where the order of the data matters and the model is updated based on this sequence.

If the data of a data set needs to be grouped, K-means clustering, which represents an offline full batch learning algorithm, can be used for this [58, pp. 45–63]. If $\mathbf{f}[k]$ as a feature vector with historical batch data was formed through preprocessing and data engineering, initial cluster centers \mathbf{C} must be defined during the initialization step. The cluster assignment is then carried out for each data point using a distance measure (e.g., using the Euclidean distance) according to

$$d(\mathbf{p}, \mathbf{C}) = \sqrt{\sum_{i=1}^{n} (p_i - c_i)^2}$$
(2.12)

while $\mathbf{p} = [p_1, \ldots, p_i]$ with $i \in \mathbb{N} | n > 0$ Datapoints and $\mathbf{C} = [c_1, \ldots, c_i]$ with $i \in \mathbb{N} | n > 0$ cluster centers. This is followed by the recalculation of the cluster centers \mathbf{c}' according to

$$\mathbf{c'}_j = \frac{1}{|A_j|} \sum_{\mathbf{p} \in A_j} \mathbf{p} \tag{2.13}$$

as the mean value of all data points assigned to the cluster, \mathbf{c}'_j is the center of the *j*-th cluster, and A_j is the set of all data points assigned to this cluster. The recalculation according to equations 2.12 and 2.13 is iterated until no significant changes in \mathbf{c}' occur [58, pp. 45–63].

An alternative to offline full batch learning with K-means is the batch incremental learning algorithm, mini-batch K-means [59]. This algorithm divides the data into smaller batches, enabling its use in offline or online applications. Such adaptability makes it suitable for a wide range of data processing scenarios. Each batch contains a selection of n data points. These are assigned to the existing cluster centers \mathbf{C} , and then the cluster centers are updated according to

$$\mathbf{c}_j = (1 - \eta) \cdot \mathbf{c}_j + \eta \cdot \mu(x_{\text{batch}})$$
(2.14)

which corresponds to weighted averaging, where c'_j is the center of the *j*th cluster, $\mu(x_{\text{batch}})$ is the mean of the data points of the batch that were assigned to the *j*th cluster, and η is the learning rate that determines how much the cluster center is shifted [59]. The advantage of the mini-batch K-means is its efficiency with large data sets and the ability to work in real time, as the entire data set does not have to be processed with each iteration.

In data stream mining, online clustering is central to meet the spe-

cific challenges associated with processing continuous and dynamic data streams [60, pp. 67–76] and is defined as:

Definition 2.1	6: Data	Stream	[56, r	$\mathbf{b}.\ 26$
----------------	---------	--------	--------	-------------------

Data stream are data generated in real-time and whose volume is potentially unlimited.

To analyze data streams, it requires approaches that are not only efficient, but also able to adapt to changes in the data streams [61, pp. 23–32] [56, pp. 87–105].

The CluStream algorithm, which represents an online learning instance incremental method, has been specifically developed for the online clustering of data streams [62]. The special feature is that the current structure of the data can be recorded in real-time, and historical data can also be taken into account [47]. The CluStream algorithm is based on the concept of micro-clusters, which summarize data streams [47, pp. 237–239]. A microcluster for a data stream is represented as c(t) = (N, LS, SS, LT, ST), where [56, pp. 177–179]:

- p is the number of data points in the cluster up to time t.
- LS is the linear sum vector of the data points in the cluster.
- SS is the squared sum vector of the data points in the cluster.
- LT is the sum of the timestamp of the last update of the cluster.
- *ST* is the sum of the squares of the timestamp of the last update of the cluster.

When a new data point x arrives at time t, the microcluster is updated:

$$p_{new} = p_{old} + 1 \tag{2.15}$$

$$LS_{new} = LS_{old} + x \tag{2.16}$$

$$SS_{new} = SS_{old} + x^2 \tag{2.17}$$

$$LT_{new} = LT_{old} + t \tag{2.18}$$

$$ST_{new} = ST_{old} + t^2 \tag{2.19}$$

The distance d between a data point x and the center of a micro-

cluster C is often calculated using the Euclidean distance [46, p. 252]. The center of a microcluster is determined by $\frac{LS}{N}$. The CluStream method is a robust and adaptive method that makes it possible to extract information from continuously generated data and effectively capture dynamic changes in the data that frequently occur in the production environment. Validation of clustering algorithms is required to ensure the reliability and accuracy of the clusters generated. Checking the results ensures that the algorithms create meaningful and valuable groups within the data. Validation also helps to adjust and optimize the machine learning models to improve performance in real-world application use cases, such as those in the validation Chapter 7 of this thesis.

2.3.3 Validation of Clustering Algorithms

There are various methods for evaluating the quality of clustering algorithms. Two of the most common are discussed below. The silhouette coefficient s [63], where a higher coefficient value suggests more distinctly defined clusters in the model. The silhouette coefficient, calculated for each sample, comprises two components:

- a This represents the average distance between a sample and all other points in its cluster.
- b This is the average distance between a sample and all other points in the nearest cluster that it is not a part of.

For an individual sample, the silhouette coefficient s is calculated by

$$s = \frac{b-a}{\max(a,b)} \quad \text{where} -1 \le s \le +1 \tag{2.20}$$

with the conditions:

- s = +1: Indicates an ideal, high-density clustering where each group is clearly separated from others.
- s = -1: Represents a completely incorrect or inappropriate clustering, with no meaningful grouping of the data.

• $s \approx 0$: The clusters overlap or are not clearly distinguished from one another.

On the other hand Davies-Bouldin-Index [64] can be used to evaluate the model to indicates the average similarity between clusters, where similarity is a measure that compares the distance between the clusters with the size of the clusters themselves. The index is defined as the average similarity between each cluster c_i for $i = 1, \ldots, k$ and its most similar one c_j . In the context of this index, similarity is defined as a measure R_{ij} that trades off

- s_i , the average distance between each point of cluster i and the centroid of that cluster also know as cluster diameter.
- d_{ij} , the distance between cluster centroids i and j.

A simple choice to construct R_{ij} so that it is nonnegative and symmetric is with

$$R_{ij} = \frac{s_i + s_j}{d_{ij}} \tag{2.21}$$

and the Davies-Bouldin index is defined by [65] as

$$DB = \frac{1}{k} \sum_{i=1}^{k} \max_{i \neq j} R_{ij}.$$
 (2.22)

Both the Silhouette Coefficient and the Davies-Bouldin Index are metrics for evaluating the cluster quality of an online and offline clustering algorithm. These metrics, independent of the clustering method used, assess how well data points have been assigned to clusters.

In this thesis, the DD-EFMod method is used to analyze data and derive key figures from the This is possible with the method developed in in this work for the Energy Flexibility Data Model. The machine learning algorithms and methods presented are used for this. A structured approach is required for the application and implementation of the presented algorithms. The CRISP-ML(Q) method is presented below for this purpose.

2.3.4 Implementation of Machine Learning Algorithms with CRISP-ML(Q)

The CRISP-ML(Q) model is an adaptation of the Cross-Industry Standard Process for Data Mining (CRISP-DM) framework [66] specifically for machine learning. It represents a structured process for planning, implementing, and evaluating machine learning projects [67]. It integrates a strong focus on quality assessment in every project phase, from conceptualization to implementation and operational deployment. The CRISP-ML(Q) model aims to balance technical feasibility, practical applicability, and the creation of added value [67]. The CRISP-ML(Q) approach is essential for the development and implementation of the DD-EFMod method for determining the key figures of the EFDM. The structured procedural model of CRISP-ML(Q) ensures that all aspects of the applied machine learning algorithms, from data preparation to model evaluation, are considered systematically and under constant quality control. This quality assurance is important as the



Figure 2.6: Phases of the CRISP-ML(Q) Process (own illustration adapted from [66] based on [67]).

accuracy and reliability of the identified energy flexibility key figures of the EFDM through DD-EFMod directly influence the effectiveness of the energy flexibility operation. Furthermore, the CRISP-ML(Q) supports the effective planning and implementation of machine learningbased projects through its structured approach. Adapting CRISP-ML(Q) in the development of DD-EFMod thus ensures that the identified key figures of the EFDM are robust, reliable, and directly implementable.

The CRISP-ML(Q) model is divided into several phases:

- Business & Data Understanding: This initial phase defines project goals and requirements from a business perspective. It lays the foundation for the entire project. The data sources are identified and examined in this phase. The aim is to deeply understand the data used during the project.
- **Data Preparation**: This phase includes all the steps required to prepare the data for analysis. This includes selecting, cleaning, and transforming the data.
- Modeling: In the core phase of the model, suitable machine learning algorithms are selected and applied. This phase also includes the adaptation of the algorithms to the specific requirements of the project as well as the evaluation and selection of the models based on their performance.
- **Evaluation**: The evaluation phase focuses on checking the models in terms of their effectiveness and accuracy. The technical aspects of model performance and the relevance of the results to the business objectives are considered.
- **Deployment**: In this phase of the CRISP-ML(Q) process, the selected model is transferred to the production environment. This phase also includes planning for monitoring and regular maintenance of the system to ensure its performance and reliability.
- Monitoring & Maintenance: This step includes the continuous monitoring of system performance and model accuracy to ensure that the models continue to function effectively and effi-

ciently under changing environmental conditions. It also includes maintenance work, where models are adjusted or retrained to maintain relevance and accuracy over time and respond to new data or requirements.

The CRISP-ML(Q) model thus supports a comprehensive and systematic approach to machine learning projects, from data preparation to model development and evaluation through to practical implementation. The inclusion of quality assessment ensures that the solutions developed are also effective and sustainable in practical application. It is, therefore, very effective for use in the context of energy flexibilization.

2.4 Interim Summary

This chapter provides a comprehensive basis for developing the method for DD-EFMod by focusing on key concepts and technologies that contribute to its effectiveness and feasibility. First, industrial energy systems in the context of production systems are discussed. This is followed by an introduction to demand-side management and the concepts of energy flexibility, as well as corresponding energy-flexibility measures. Furthermore, the core aspects of modeling energy flexibility is discussed. In particular, the Energy Flexibility Data Model (EFDM) and its classes, flexible loads, storage, their dependencies, and specific flexible load measures, are introduced. This is followed by an explanation of the importance of digital manufacturing and RAMI 4.0 in order to illustrate the implementation of energy-flexibility measures in modern production environments, such as in the validation use case of this work. The importance of modeling, simulation and data-driven approaches is emphasized, as they are essential for the development of the DD-EFMod method. Furthermore, the basics of data analytics and machine learning algorithms, especially clustering methods, are presented, which are central to the identification of key figures of the EFDM. In addition, the distinction between online and offline machine

learning algorithms is discussed in order to clarify their applicability in real-time applications and more static environments. Furthermore, the validation of clustering algorithms is shown. Finally, the CRISP-ML(Q) process is described in detail, providing a structured process for quality assurance and successful implementation of machine learning algorithms, which are essential for developing and implementing the DD-EFMod method.

3 State of the Art and Research

"To raise new questions, new possibilities, to regard old problems from a new angle, requires creative imagination and marks real advance in science."

Albert Einstein

The following chapter presents the current state of science and technology on modeling energy flexibility in production systems. This is conducted on the basis of a systematic literature review. This procedure and the methodology is explained in Section 3.1. Subsequently, Section 3.2 shows the results of the individual methodological steps and presents the relevant reports identified on the basis of the literature review. Finally, Section 3.3 discusses the identified need for research, which is addressed in this dissertation.

3.1 Procedure of Systematic Literature Review

The presented systematic literature review is based on the approach of the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) statement [68]. Following the PRISMA statement, a systematic literature review contains the overall parts of identification, screening, eligibility and including of studies and reports [69]. For clear understanding of what a report in this context is, the PRISMA statement gives these following explanations:

A report is a document (printed or electronic) supplying information about a particular study. It could be a journal article, preprint, conference abstract, study register entry, clinical study report, dissertation, unpublished manuscript, government report, or any other document providing relevant information [68]. As second, a study is an investigation, such as a clinical trial, that includes a defined group of participants and one or more interventions and outcomes. A study could contain multiple reports [68].

In addition to the PRISMA based systematic literature review, the part of concept and planning which is based on [70] and [71] is placed as the first. So my complete systematic literature review procedure is shown in Figure 3.1. The systematic literature review starts with the part concept and planning, which contains the following three steps:



Figure 3.1: Overall steps of the systematic literature review following the PRISMA statement (own figure based on [68, 73]).

In the first step **Define search strategy** the dimensions for the systematic literature review are defined. These dimensions, following [18] and [72] are objective, system boundary, approach and method. The objective of this work is to improve the enabling process for energyflexible factory operation. This results in the system boundary, the building of a factory with a production process. To realize and describe energy flexibility a model based approach is chosen. In addition, the modeling should be data-based to obtain a system description that is as real as possible. Therefore data from different sources like load profiles, electric or thermal energy consumption could used for dataanalysis or pattern detection. To evaluate and classify the relevant reports, I developed a classification scheme. The scheme is used to evaluate and classify the identified reports for research content under the aspects of the objective, the system boundary, the approach, and the methodology to the following topics:

- energy consumption data analysis or pattern detection
- energy flexibility
- industrial energy flexibility
- identification, description, usage of energy flexibility
- modeling of energy flexibility
- use of a data-based approaches to reach one of these targets
- use of an automated approach for modeling
- use a machine learning approach

As second step **choose database** the research data bases, including open access reports, "Web of Science" and "ScienceDirect" are defined. These data bases include scientific journals, publishers and conferences of related engineering communities and topics such as The International Academy for Production Engineering (CIRP) and Institute of Electrical and Electronics Engineers (IEEE).

In **define query** the search queries for browsing the databases are defined. Therefore, for each dimension resulted from step one analogously and closely related terms are considered and systematically linked by

	Demension	Search string	
	objective	"energy flexibility" OR "demand response"	
		OR "demand side management"	
ANI	aveter boundary	"production" OR "factory OR "manufacturing" OR	
	system boundary	"machine tool" OR "industry"	
	approach	"model" OR "simulation"	
	methode	"data-based" OR "data stream mining" OR	
		"data analysis" OR "pattern detection"	

Table 3.1: Search strings for each dimension of the systematic literature review (own table).

logical operations. This results in the queries given in Table 3.1. In the next step **metadata search** these search strings are used to search the databases for each dimension starting with the dimension objective. Afterwards, the search string is successively extended by the search strings for the other dimensions with a logical AND operator. The second part of the systematic literature review is identification. This starts with the **metadata search** where the browsing is done, using the queries given in Table 3.1, and the results are collected. It follows by the **metadata analysis**. In this step, the results are interpreted and discussed. The results of my systematic literature review are discussed in Section 3.2. In addition to browsing the research databases, also the step **manual search** is done. The manual search is necessary to find relevant national standards (e.g., International Organization for Standardization (ISO) or VDI) and other previous dissertations on the topic which are not listed in the databases. In the next parts, screening and eligibility the **analysis and synthesis** of the report results from the identification part is done. Subsequently, as shown in Figure 3.1, I performed the review of the titles and abstracts of the reports one by one. For the reports, which fullfils the scheme of step one, the classification and review of the full text of the report was done. In the course of the literature synthesis, essential characteristics are recorded and summarized (see Section 3.2). Through screening of the literature references in the selected publications, additional relevant references are fed back into the literature search if suitable. Finally, I evaluate



Figure 3.2: Results of the metadata analysis for the search query combination of the dimensions objective, system boundary and approach (own Figure).

the reports to the developed scheme and identified the research gap based on the analyzed literature.

3.2 Results of the Literature Review

As the first result of the systematic literature review the analysis of the metadata search shows that 166 relevant reports result from browsing the databases for all dimensions (see Table 3.2). In comparison, the objective query still yields 10677 reports. The analysis of the temporal development of the relevant publications clearly shows the increased research activities in industrial energy flexibility and demand-side management. This evolution is shown in Figure 3.2 for the third level of dimensions (objective, system boundary, and approach). Also, 20 reports related to the topic could I identify while doing the manual search. So overall I start the part of screening for 184 reports. In the title review, only 84 were left, and while reviewing the abstracts, 36 reports were identified as relevant. Finally, 26 reports result from the full text review to consider. These 26 reports are discussed shortly and classified

by the scheme in the following. The results of the classification is given in Table 3.3.

Akhavan-Hejazi *et al.* present various possible applications of big data analytics in energy systems. Among other things, real-time analysis and monitoring, monitoring of decentralized energy networks, but also the improvement of demand response are listed. The necessary basis for this is big data to enable robust analyses and thus extract new information [74].

Li *et al.* show that needed data for machine learning models are difficult to collect. The most important barriers are cost, technical constraints or privacy. To address these problems they propose a data generation model to synthesize energy consumption time series. These synthetic data a used to train and validate classification models. In result the time series have satisfactory similarity with real data in terms of data distribution, pattern and performance [75].

Calikus *et al.* propose a data-driven approach for pattern detection methods for analyze power consumption data. The automatic analysis of heat load patterns of district heating networks requiring no prior knowledge. The given approach clusters the customer power load profiles into different groups and extracts their representative patterns [76].

Corsetti *et al.* give a modeling framework and an associated optimization tool to support short-term operational planning and electricity market service trading. The focus is to to deploy flexibility with specific

Table 3.2: Results of the metadata search of the systematic literature review (own Table).

Combination of demensions	Number of Reports
objective	10677
objective AND system boundary	3976
objective AND system boundary AND approach	3238
objective AND system boundary AND approach AND methode	166
application to provision of frequency control ancillary services. Therefore they discribe caracteristics of flexibility, based on energy grid time series and model it with mathematical features to use mixed-integer linear programming optimization of the flexibilities [77].

Sossenheimer gives in his work, a hybrid energy metering point concept for the implementation in industrial environments. The availability of energy demand information is the basic prerequisite for the identification, evaluation and implementation of energy efficiency and energy-flexibility measures. Accordingly, end-to-end energy monitoring at all production system levels is essential for enabling holistic energy efficient and energy-flexible factory operations. The primary objective of this work is to develop a continuous hybrid energy monitoring concept taking into account all production system levels. [27, pp. 68–76]

Tsay *et al.* provide a method, using time-series analysis, to facilitate online scheduling computations. Optimal demand-response scheduling of industrial air separation units should account for process dynamics and control to ensure that schedules are dynamically feasible. Their work presents a data-driven strategy to create machin learning models for an industrial process from routine operational data. They find that time-series models can accurately represent process dynamics over several days, but that their accuracy can suffer over time. To handle this challange they show that it is possible to online update the models, which significantly improve demand and response planning models in the future [78].

Ulbig *et al.* quantify and visualize the technically available operational flexibility of individual power system units. Necessary metrics for defining power system operational flexibility are presented. The flexibility properties of different power system unit types are qualitatively analyzed and compared to each other. The contributions of this paper are the presented modeling and analysis techniques in electric power systems. The calculation of the remaining operational flexibility in a power system after having subtracted the needed from the originally available operational flexibility was shown [79]. **Zhao** *et al.* discuss ensemble clustering for power data while reducing the dimensions of the data. The relationship between the data dimensions and the clustering performance were determined. Specific characteristics and practical significance for different mode were determined. Furthermore, the users' electricity consumption behavior in one week based on these modes were analyzed. This paper proves that ensemble clustering can be successfully applied for power data analysis [80].

DIN SPEC 91366 specifies a reference model that systematically shows companies which aspects must be considered and investigated when identifying, evaluating and using energy flexibility. The reference model supports the communication of energy information and knowledge as well as an evaluation of energy consumption behavior with regard to energy flexibility in order to create a basis for participation in the intelligent electricity supply system. Relevant control and planning variables in energy and load management or in production planning and control are included in the evaluation [81].

DIN SPEC 91410-2 specifies requirements for the method of identifying and evaluating the temporal flexibility of energy conversion plants in buildings and districts. The method makes it possible to identify these plants with regard to flexibility options, to characterize and evaluate them according to technical, organizational, systemic and informational criteria. Different fields of application for flexibility in the electrical energy market, grid and system are considered. Capabilities of electrical systems to dynamically respond to corresponding flexibility needs are evaluated [82].

VDI 5207 Part 1 focus on energy-flexible factories. The standard describes the process of identifying and marketing energy flexibility and defines pertinent terms. The standard is applicable to different forms of energy but focuses on electrical energy [24].

VDI 5207 Part 2 describes in detail the process of identifying and marketing options of energy flexibility for industrial companies. Also challanges and a guideline to solve them are given [26].

Lee *et al.* show a method to evaluate the demand respose potential of industrial loads according to a process of related load-characteristic data analysis. The proposed potential-estimation model considers frequency, consistency, and demand response event operation scores during designated ramp-up and ramp-down time intervals separately [83].

Sadeghianpourhamami *et al.* focus on the validation of flexibility in households. Analyses of customer load patterns show high uncertainty about customer habits in offering flexibility. Therefore, a quantitative specification of flexibility was done. To address this, they give two systematic methodologies for modeling individual customer behavior, evaluate the proposed models, and provide a fundamental analysis of factors influencing flexibility behavior based on statistical tests. [84].

Schraml shows a method which makes it possible to optimize the electrical load profile of metal-cutting machine tools, in particular by synchronizing the activities of auxiliary units over time. The results enable operators and manufacturers of machine tools to estimate and evaluate the load reduction potential of metal-cutting machine tools and, if the evaluation is positive, to realize it [85, pp. 49–54].

Schulze et al. present an approach for flexibilization of energy intensive system elements. Against this background, technical capabilities and operational strategies for energy-flexible management of industrial cooling towers are analyzed, basing upon empiric data from a plant. A synergetic approach of data-driven analysis and scenario-based simulation is applied to demonstrate benefits of energy-flexible technical building services management [86].

Schott *et al.* present a descriptive model for industrial flexibility with respect to power consumption. The advancing digitization in the energy sector opens up new possibilities for utilizing and automatizing the marketing of flexibility potentials and therefore facilitates a more advanced energy management. This requires a standardized description and modeling of power-related flexibility [14]. A detailed description of this Energy Flexibility Data Model is given in Section 2.1.2. **Bahmani** *et al.* propose a mathematical optimization model that uses the generic data model for flexibility description based on [14]. The optimization model supports industrial companies to select when, where, and how to market their flexibility potential to optimize profit. The results of the optimization model evaluation suggest the model can fulfill its purpose under different use cases even with complex use cases such as various loads and storages [87].

In **Bank** *et al.* we present an approach to integrate Energy Flexibility (EF) into Production Planning and Control (PPC). Based on the energy-oriented PPC, the approach identifies and model EF of processes in a generic Energy Flexibility Data Model based on [14], which is subsequently integrated in the energy-oriented production plan and further optimised on the energy market side [36].

Unterberger *et al.* model energy storages, energy-flexible machines and smart controls. To realize smart control, electrical energy must be included into production planning and control as resource. Therefore an energy-flexible production control is realized by using the modeling language SysML [88].

Degefa *et al.* propose unified characterizing terms for flexibility resources. Furthermore, a taxonomy method which is applied to classify flexibility resources is presented. The taxonomy method clears the confusion on "what-is-what" under the concept of flexibility. This paper also presents the benefits of unified characterizing terms in mapping flexibility resources to ancillary services [89].

In **Lindner** *et al.* we show an approach to modeling and aggregation of flexibility in complex manufacturing processes. To overcome this challenge, a method for the aggregation of energy flexibilities that is based on the generic Energy Flexibility Data Model given by [14]. The method proposes a two-step approach to aggregate flexibilities cost efficiently and considers manufacturing specific limitations [90].

Strobel develops a software framework to realize the demand response measures store energy inherently given in [24]. Therefore the energy storages are modeled using the key values given in [14]. For the application the energy flexibility model is used for an optimization [72, pp. 74–104].

In **Buhl** *et al.* we present an advanced model of [14] for the generic and standardized description and modeling of energy flexibility. The data model enables a (partially) automated information-technical processing of the most diverse flexibility. The aim is to develop a comprehensive data model to represent flexibility in a flexibility space and concrete flexibility measures (see Section 2.1.2). The focus is on the mapping of technically and energetically relevant information in a granularity that enables the communication and of flexibility between industrial companies and energy markets [28].

Sauer et al. show how an energy synchronization platform enables the entire process of automated energy flexibility trading from the machine of a production system to energy flexibility marketing services. It thus represents the overall concept of a digital ecosystem. The processes are illustrated by means of use cases. In addition, we present the further developed Energy Flexibility Data Model [28] and show the application [5, p.249-260].

In Lindner *et al.* we examine the practicality of a methodical approach for energy-flexible and energy-optimal operation in the field of metal-cutting production. The analysis focus on a grinding machine and its central cooling system. An Energy Flexibility Data Model is built for each subsystem, which describes energy flexibility potentials generically. This is then extended to enable combined energy cost-optimal production planning. As a basis for the links between the data model representations, the cooling between the subsystems are modeled using parameter-estimation methods. Based on the presented approach, the results successfully validate the possibility of energy-flexible cost-optimal and sensor-reduced production planning [30].

3.3 Conclusion on the State of the Art and Research Deficit

The systematic literature review conducted on the state of the art and research in the field of energy consumption data analysis and the topic of energy flexibility shows a strong increase, especially since 2016 in this field (see Figure 3.2). Based on the classification scheme (see Table 3.3), I could show that the work from the energy and consumption data analysis field uses data-based tools and methods. In addition to statistical methods, machine learning approaches, e.g., for pattern recognition or parameter estimation, are also used for this purpose.

On the other hand, some research work and standards deal with identifying and describing energy flexibility, especially in the complex environment of industrial processes. The analyses of the systematic literature review show that about one-third of the reports or studies on this topic are within the SynErgie project. The identified relevant reports or studies often focus on identifying or describing industrial energy flexibility in specific use cases. In these cases, the EFDM presented in Section 2.1.2 is used as a basis. As it turns out, no report has been identified that provides automated data-based or data-driven modeling of industrial energy flexibility through load profile analysis. This research gap is to be filled by the present work.

(own table).										
Report	Consumption analysis	Energy Flexibility (EF)	Industrial EF	EF Identification	EF Description	EF Use	EF Model	Data-Based	Automated	Machine Learning
Akhavan-Hejazi	•							•		
$et \ al. \ [74]$										
Li et al. [75] Calikus et al. [76]									•	
Corsetti $et al.$ [77]	ĕ	•			•		•	ě		•
Sossenheimer [27]	•	•			-		-	•	\bullet	•
Tsay et al. [78]	\bullet	\bullet	\bullet	\bullet				\bullet	\bullet	•
Ulbig <i>et al.</i> [79]	\bullet			\bullet			\bullet	\bullet		
Zhao <i>et al.</i> [80]	\bullet	-	_	-	-			•		•
DIN SPEC 91366 [81]		•	•	•	•					
DIN SPEC 91410-2 [82]				•						
VDI 5207 Blatt 1 [24] VDI 5207 Blatt 2 [26]					•					
VDI 5207 Diatt 2 [20] Lee et al [83]										
Sadeghianpourhamami		•	•	ŏ				•	•	•
et al. [84]				-				-	-	-
Lee <i>et al.</i> [83]		\bullet	\bullet	\bullet						
Schraml [85]		\bullet	\bullet	\bullet						
Schulze et al. [86]		\bullet	\bullet	\bullet	\bullet	\bullet				
Schott et al. [14]		•	•		•	•	•			
Bahmani <i>et al.</i> [87]		•	•		•	•	•			
Bank et al. [36]					•					
Unterberger <i>et al.</i> [88]		•								
Degeta et al. [89]										
Strobel [72]	•				ž		-			
Buhl et al $[28]$	-	ě	ě	ě	ě	ě	ě			
Sauer $et al.$ [5]		ě	ē	ē	ě	ē	ē			
Lindner $et al.$ [30]		ē	ē	ē	ě	ē	Ĭ	ullet		•

Table 3.3: Relevant reports as results of the systematic literature review (own table).

4 Research Gap and Concept

"A new idea must not be judged by its immediate results."

Nikola Tesla

The previous chapters showed the necessity of modeling industrial energy flexibility. In addition, a need for research in automated parameterization of such models was identified. On this basis, the research objective, the resulting research question and the derived research hypotheses are formulated in this chapter. This chapter concludes by deriving the requirements and defining the assumptions and prerequisites for the functionalities of a method that can be applied to achieve the research objective.

4.1 Research Objective

The central research objective of this thesis is derived from the framework outlined in Chapter 1 and 2 and the research gap identified in Chapter 3. To enable a factory to become energy-flexible, it is necessary to describe the existing energy flexibility. This description is possible with the help of the data model shown in Section 2.1.2. Due to the complexity of production processes and their supply systems, the correct modeling of existing energy flexibilities is not trivial. However, this central step of the enabling process should be as simple as possible for plant operators or energy managers of a company. In summary, the research objective can be formulated as follows:

Research Objective

The primary goal of this work is to simplify the enabling process of a factory to energy-flexible operation. For this purpose, a data-driven method should be developed, which reduces the modeling effort of energy flexibility, based on information and data from the production system.

In order to achieve this research goal in a well targeted manner, the research questions and, from these, the research hypotheses are formulated.

4.2 Research Questions and Hypotheses

From the research objective, the central research question is derived, which focuses on the modeling and parameterization of the EFDM:

Research Question

Is it possible to develop a method for automated parameterization of an energy flexibility model, based on data from a production system, to describe the energy flexibility of the production system?

This given central research question leads to further sub-research questions, which serve to answer it and reach the objective. These subresearch question focus on specific parts to reach the research objective.

Sub-Research Questions

- 1. What methods can enable automatic parameterization of a model to describe energy flexibility?
- 2. Which data and which information of a production system are necessary for modeling energy flexibility?
- 3. Can an automated parameterized model enable an energy-flexible production system operation?

The research hypotheses can be derived from these research questions. Accordingly, under the assumption that an automated parameterization of energy flexibility models is possible, it follows that these also fulfill the requirements to implement an energy flexibility measure. The automated parameterized models should fulfill the same requirements as a manual modeling of energy flexibility. This results in my first research hypothesis.

Research Hypothesis : 1

Data-driven automatically parameterized energy flexibility models of machines in production infrastructure can be used to enable energy-flexible operations.

Growing digitization and data availability mean that more and more data-based methods can be used to obtain information (see Chapter 2). The systematic literature review shows that machine learning and databased or data-driven methods have the potential to be used for modeling, simulation and data analysis in the context of consumption analysis and energy flexibility. Furthermore, the developed method should also be applicable in the brownfield and systems with low amount of historical data availability, e.g. through temporary or streaming data acquisition. This leads, based on the research question, to my second research hypothesis:

```
Research Hypothesis : 2
```

Machine learning and data stream mining algorithms can be used to automate the parameterization of energy flexibility models of machines from the production infrastructure.

4.3 Requirements and Assumptions

The approach devised to achieve the research objective and address the research questions must satisfy specific functional and quality requirements. Additionally, boundary conditions and prerequisites are essential for efficiently implementing this method and its transfer to industrial applications [91]. These requirements, assumptions, and prerequisites are detailed and discussed below.

The following functional requirements on the method (FR-M) focus on the development of the method itself and are essential for the approach:

- **FR-M 1: Applicability:** The method must be designed to be applicable for real industrial production systems and scenarios, as explained in Chapter 2.
- **FR-M 2: Modularity:** The method is structured so that individual steps of the method can be applied separately if required.
- FR-M 3: Transferability: The method is designed in such a way that different energy-flexibility measures (see Table 2.1) can be realized and should be transferable to different machine types.
- **FR-M 4: Reproducibility**: Repeated application of the method under the same conditions leads to the same or very similar results.

In addition, with a focus on applicability, requirements are needed for the energy flexibility to be modeled. These functional requirements for energy flexibility modeling (FR-EF) are:

- **FR-EF 1: Feasibility:** The parameterized model describing the existing energy flexibility can be used to realize it.
- **FR-EF 2: Accuracy**: The energy flexibility models parameterized with the help of the method should be sufficiently accurate to enable an application.
- **FR-EF 3: Maximization:** The calculated energy flexibility model should describe the maximum realizable energy-flexibility potential.
- **FR-EF 4: Explainability**: The calculated key figures of the energy flexibility model should be as comprehensible as possible.
- FR-EF 5: Safety: The parameterization of the energy flexibility key figures must take the related machines' technical boundaries into account.

The above FR-EF lead to requirements for the Machine Learning (ML) algorithms to be used for the energy flexibility modeling. The for the

DD-EFMod method relevant quality requirements (QR) for machine learning models, which were derived from the de facto standard of machine learning applications, the CRISP-ML(Q) Studer *et al.* [67] explained in Chapter 2, are as follows:

- **QR 1: Quality:** The ML model's quality should fit the application requirements and should validated by accuracy and performance measures.
- **QR 2: Robustness**: The ML algorithms should have the capability to maintain its performance level despite certain circumstances.
- **QR 3: Explainability:** The ML model should directly or posthoc explainable.
- **QR 4: Scalability**: If necessary, the ML model should also be able to work reliably with more significant amounts of data.
- **QR 5: Real-time capability**: The ML model should work in a application defined time.

The application and the development of the method based on prerequisites that are assumed to be fulfilled within the context of this work. These prerequisites (PR) are:

- **PR 1: Data Availability:** All data and information (see Chapter 7, e.g., power value, operating condition) necessary to calculate and parameterize key figures of the EFDM are available and of sufficient quality.
- PR 2: Information and Communication Technology (ICT)-Availability: The existing ICT infrastructure ensures sufficiently fast and secure transmission of the required data to the corresponding data processing devices.
- **PR 3: Energy Flexibility Identified**: The relevant systems for which an energy flexibility model should be parametrized and the corresponding energy-flexibility measure are already known through methods of energy flexibility identi-

fication following [26] with tools like the E-FLEX-Scanner or the Energy-Flexibility-Audit [5, pp. 167–169].

PR 4: Technichal Scope: The method focuses on the determination of technical key figures for the modeling of energy flexibility.

The derived requirements and prerequisites form the framework for developing the methodology for a DD-EFMod. The development of this methodology is presented in Chapter 5.

Type	Identifier	Requirement & Prerequisites
Functional Requirements	FR-M 1 FR-M 2 FR-M 3 FR-M 4	Applicability Modularity Transferability Reproducibility
	FR-EF 1 FR-EF 2 FR-EF 3 FR-EF 4 FR-EF 5	Feasibility Accuracy Maximization Explainability Safty
Quality Re- quirements	QR 1 QR 2 QR 3 QR 4 QR 5	Quality Robustness Explainability Scalability Real-time capability
Pre- requisites	PR 1 PR 2 PR 3 PR 4	Data Availability ICT-Availability EF Identified Technical Scope

Table 4.1: Summary of requirements and prerequisites (own table).

4.4 Interim Summary

Based on the previous systematic literature review from Chapter 3, this Chapter derives the central research objective and the research questions. Subsequently, the research hypotheses are formulated. Finally, the functional requirements and prerequisites, summarized in Table 4.1, for developing a method to automate data-driven parameterization of energy flexibility models are derived.

5 Method Development

"The true Logic for this world is the Calculus of Probabilities, which takes account of the magnitude of the probability."

James Clerk Maxwell

To fulfill the research question and research hypotheses defined in Chapter 4, this chapter shows the development of the DD-EFMod method. First, Section 5.1 describes the approach and structure of the method, with detailed explanations of the steps and how the automated parameterization of energy flexibility models are realized within the DD-EFMod method. In Section 5.2 afterwards, the software concept for the method and its components is discussed.

5.1 Approach and Structure of the DD-EFMod Method

The research goal of simplifying the enabling process of energy flexibility in factories (cf. Figure 1.2) is the starting point. As described in Section 2.1.1, the third step of the process requires the implementation of identified possible energy-flexibility measures while taking into account the technical environment. As the systematic literature review shows, the realization requires modeling the identified energy flexibilities (e.g. Lindner *et al.* [30]). This can be done with the Energy Flexibility Data Model (EFDM) described in Section 2.1.2.

With the goal, that the parameterization of the EFDM should be as simplified and automated as possible (see Section 4.3) and as a result conducted from the systematic literature review (see Table 3.3), machine learning and data analytic algorithms can be used to achieve this.

Due to its structured approach to the planning, implementation and evaluation of machine learning projects, the application of the CRISP-ML(Q) process [67] (see Chapter 2) in this thesis provides a good basis for the realization of a method for data-driven automated parameterization of energy flexibility key figures of the EFDM in the context of this thesis. The CRISP-ML(Q) process is particularly well suited to the complexity of energy flexibility in factories, as it takes quality assurance into account at every stage of the project. This focus on quality is



Figure 5.1: Flow chart of the DD-EFMod method for data-driven automated parameterization of energy flexibility models (own figure based on [13] and [67]).

essential for the parameterization of EFDMs, as energy flexibility needs to be modeled based on multiple data sources and variable conditions as they occur in production systems. In addition, the CRISP-ML(Q) process supports the use of advanced machine learning and data analytics algorithms, which are required to automate the parameterization of EFDMs. By adapting the CRISP-ML(Q) process steps, I can systematically address the challenges associated with accurate modeling of energy flexibility by ensuring that each step, from data verification over machine learning model deployment to energy flexibility model validation, is executed precisely. This structured methodology is essential to achieve the quality required for the modeling and parameterization of EFDMs.

The adaption of the CRISP-ML(Q) process to the automated determination of the key figures of the EFDM for the description and modeling of energy flexibility by applying machine learning or data analytic algorithms represents the DD-EFMod method. The overall consideration for developing the DD-EFMod method is divided into the three layers: **energy-flexible system layer** with focus on the complete energy-flexible system, the **application and implementation layer** with focus on the handling and implementation of the complete EFDM, and the **energy flexibility modeling layer** in which the automated parameterization of each relevant energy flexibility key figure of the EFDM is realized by the application of machine learning or data analytics algorithms. The overall methodology and the interaction between this three layers is shown in Figure 5.1 and discussed in detail in the following.

5.1.1 Layer One - Energy-Flexible System

This layer is based on the VDI process (see Figure 1.2) and defines the starting point and the process or system to be enabled in the context of the DD-EFMod method. As described in Section 2.1.1, this layer focuses on the holistic identification, enablement, implementation, de-

ployment, and commercialization of energy flexibilities with the following steps:

- 1. **Potential analysis:** Initially, the energy-flexibility potential is evaluated both technically and economically. The input data, such as the power consumption, operating limits or system dependencies of a system, are collected with the help of an external software tool, the E-Flex Scanner. This tool supports those responsible in selecting relevant machines and systems for energy-flexible operation and helps to define company objectives for the use of energy flexibility [5, pp. 168–169].
- Concept and planning: Detailed analysis and preparation for the implementation of selected energy-flexibility measures. In this step, the external software tool the Energy-Flexibility Audit can be used to systematically identify energy-flexibility measures and assess their potential, which is then documented for further processing [5, pp. 76–83].
- 3. **Application and implementation:** Technical realization of the energy-flexibility measures defining the corresponding EFDM and validation through test runs. This step is the primary focus of this thesis, representing the central step where theoretical energy-flexibility measures could be transformed into practical application-able load change profiles based on the EFDM to enhance energy-flexible operation.
- 4. *Operative marketing*: Start of marketing the achieved energy flexibility.
- 5. *Controlling and monitoring*: Monitoring and analysis of performance to identify potential for improvement.
- 6. **Optimization:** Enhancement of energy flexibility through continuous improvement.

The first two steps set the basis for successfully applying the DD-EFMod method. In the first step, the potential analysis, the energyflexibility potential is determined to identify which possibilities of energyflexible operation are technically and economically feasible. This is crucial as it determines which systems and processes are eligible. The second step builds on this by defining how and which energy-flexibility measures can be implemented. This detailed planning and preparation is essential and can be done with the help of the E-FLEX-Scanner, or the Energy-Flexibility-Audit [5, pp. 76–83]. In the context of the CRISP-ML(Q) process, these results are part of the business and data understanding. Both steps, potential analysis and concept and planning, are assumed as fulfilled as defined as the prerequisite number three in Chapter 4. The data collected in these two steps can be used in layer two of the DD-EFMod, the application and implementation layer, and further processed and analyzed to determine the key figures of the EFDM.

5.1.2 Layer Two - Application and Implementation

The focus of the DD-EFMod method is on the actual enabling, modeling, and implementation of energy flexibility measures, which are described by the EFDM. Therefore, the second layer of the DD-EF-Mod looks at the EFDM of an energy-flexible system as a whole and describes how it can be deployed. For this purpose, as it contains the modeling step of energy flexibility, the CRISP-ML(Q) process, with the steps data preparation, modeling, evaluation, deployment, and monitoring and maintenance is adapted and applied. This adaptation makes it possible to provide a methodological structure that covers the specific steps for implementing and operating an energy-flexible production. The DD-EFMod method uses this structure to effectively manage the complexity of energy consumption data and the dynamics of production systems. Data which are relevant for these steps can be taken from the previous steps of *potential analysis* and *concept and planning*, and further relevant data for the correct EFDM modeling can be determined through hardware enablement and data verification. Within the application and implementation layer, the second layer of the DD-EFMod method, various sub-steps are executed that are derived directly from

the CRISP-ML(Q) process (see Chapter 2):

- *Hardware Enabling*: Preparing and adapting the physical and technical infrastructure to collect relevant data and information to model energy-flexibility measures.
- **Data Verification and Data Preparation**: Verify and prepare the collected data to ensure its quality and usability for the subsequent modeling processes.
- Energy Flexibility Modeling and Energy Flexibility Model Validation: Modeling of the EFDM and its key figures by automated parameterization with the usage of machine learning and data analytics algorithms. The validation of the energy flexibility models applied in order to proof and quantify the automated parameterized EFDM in the energy-flexible system context.
- Energy Flexibility Model Deployment and Monitoring and Maintenance: Implement the EFDM in the production environment and continuously monitor their performance.

Between these steps, dynamic feedback loops enable continuous feedback and adaptation of the models. These loops are for iteratively improving the energy flexibility models and adapting them to changing conditions. They enable an agile response to new findings and challenges during operation. As shown in Figure 5.1, corrective feedback between the following sub-steps are considered:

- from energy flexibility model validation to energy flexibility modeling and data verification
- from energy flexibility model deployment back to data verification.

These individual steps of the application and implementation layer are described below in their respective functions.

The *Hardware Enabling* stands at the beginning of creating a model for the description of energy flexibility. It contains the enabling of relevant hardware to ensure sufficient data availability and the associated information density. Various approaches and solution can be

used for this purpose:

- data acquisition using hybrid measuring points according to [27]
- centralized or decentralized energy management systems [92]
- collecting raw data via temporary measuring technology
- readout of data points of the building automation [93]

Depending on the application, the requirements (e.g., sampling rate, consistency, and accuracy) must be defined individually, and the hard-ware must be selected accordingly. For more information, I refer to our previous work in Fuhrländer-Völker *et al.* [93], and Ahrens *et al.* [38]. As well as to the work of Lodwig *et al.* [92] with a focus on decentralized systems. In the context of this work, as defined in Section 4.3 as prerequisites number one and two, this step is assumed to be fulfilled.

The **Data Verification** step forms the basis for the subsequent steps of modeling energy flexibility. In this sub-step, data from various systems, including Enterprise-Resource-Planning (ERP), Energy Management System (EMS), machine data, data sheets, and product information, is verified. The data verification step is based on the CRISP-ML(Q) data quality verification sub-step, which sets requirements for data quality. The aim is to ensure the accuracy, consistency, and completeness of the data, which is a central condition for all subsequent steps of usage machine learning or data analytics algorithms for the automated determination of the energy flexibility key figures (EFKF) for parameterize the EFDM.

By reusing data from previous steps, duplication of work is avoided, and a consistent database is ensured. This is particularly important as *data verification* not only identifies errors, but also ensures that all relevant data and information for modeling energy flexibility is collected and interpreted correctly. Successful *data verification* makes it possible to make informed decisions about the data analytics and machine learning algorithms to be used. During data verification relevant metrics are defined. While non-technical metrics can be defined directly, technical metrics require further steps, especially those that are not trivial to determine. In this work, the aim is to carry out automated parameterization of EFDMs to increase the depth and accuracy of energy flexibility models further. Therefore for a system under consideration, let the flexibility space be \mathcal{F} . This is described by

$$\mathcal{F} = f(\mathcal{L}, \mathcal{S}, \mathcal{D}) \tag{5.1}$$

as a functional combination of the spaces of flexible loads \mathcal{L} , storages \mathcal{S} , and dependencies \mathcal{D} . Each of these spaces is spanned by the energy flexibility key figures defined in the EFDM (see Chapter 2 and Appendix A.1).

For the considered system, from the relevant defined EFKFs, it follows for the *data verification* step that $\mathbf{K} = [\mathcal{K}_1, \ldots, \mathcal{K}_i]$ with $i \in \mathbb{N}$ represents the vector of the EFKF to be determined. Data-driven modeling of the key figures is carried out based on time-variant measurement data as input variables $\mathbf{X}(t) = [x_1(t), ..., x_n(t)]$ with $n \in \mathbb{N}$. Accordingly, for

$$\mathcal{K}_i = f_i\left(x_1(t), \dots, x_n(t)\right) \tag{5.2}$$

and thus

$$\mathbf{K} = [f_1(x_1(t), \dots, x_n(t)), \dots, f_i(x_1(t), \dots, x_n(t))]$$
(5.3)

apply. Taking the example of the flexibility space of the flexible load, it follows

$$\mathcal{L} = f(\mathbf{K}(\mathbf{X}(t))) \tag{5.4}$$

which means the modeling of the flexible load depends on the corresponding energy flexibility key figure, which depends on the input variables. This is equivalent for the spaces S and D. By determining the relevant measurement data for the respective \mathcal{K}_i , relevant features required for automated key figure modeling can be derived in the next sub-step.

The *Data Preparation* forms a central element of the entire method.



Figure 5.2: Data preparation step (own figure).

It significantly influences the quality of the machine learning or data analytics algorithm procedures used and, thus influences the accuracy of the identified EFKFs and the resulting EFDMs [67]. It is an iterative step due to feedback loops from subsequent steps (as shown in Figure 5.1). This ensures continuous improvement of both the machine learning or data analytics algorithms and the EFDM. The *data preparation* includes the sub-steps *data selection*, *data cleaning*, *data construction*, and *data standardization* as shown in Figure 5.2

The sub-step data selection includes the actual selection of relevant data samples and feature selection. In this phase, according to the CRISP-ML(Q) methodology, raw input signals or datasets $\mathbf{X}(t)$ are formed that are suitable for advanced analytical processing. This step involves several key transformations like moving average, normalization or standardization (see Chapter 2). The selection and refinement of the raw data $\mathbf{X}(t)$ leads to a more focused dataset $\mathbf{x}(t)$. This process involves identifying and extracting the most relevant data features (e.g., power values, machine states, or energy consumption) that are crucial for the energy flexibility modeling. Data and information that provide only little added value for achieving the target variable will be excluded. This avoids overfitting and reduces the computational effort of the resampled application [67].

This is followed by cleaning and smoothing the data values of $\mathbf{x}(t)$. During this stage, tasks such as outlier removal (e.g., IQR-Method) and denoising (e.g., Gaussian low-pass filter) are done to create more accurate and reliable data. Unwanted interfering signals, outliers, and missing data points are filtered out. Thus, an increase in data quality takes place [67]. These pre-processing steps are essential when the data has different scales, typically for modeling EFKF based on different raw data. This transformations resulting in $\mathbf{x}^*(t^*)$ as the cleaned dataset. This step ensures the data is free of anomalies and noise, which could otherwise skew the analysis.

In the *data construction* sub-step, the feature engineering occurs. In this process, new features can be derived based on expert knowledge, data, or existing information. This can be done, for example, by generating new feature spaces or transformations (e.g., from the time domain to the frequency domain). Overall this leads to the contruction data signal $\tilde{\mathbf{x}}^*[t]$ which is processed further as shown in Figure 5.2 by using standardized data types, formats (JavaScript Object Notation (JSON), Comma-separated values (CSV)) and sampling. Finally this leads to the feature vector $\mathbf{f}[k]$ as the result of the *data preparation* step. Each of these sub-steps plays a relevant role in ensuring that the raw data is not only clean and reliable but also rich in information and structured in a way that machine learning and data analysis techniques are usable to automate the EFKF determination.

The *Energy Flexibility Modeling* step represents the third layer of the DD-EFMod. The automated modeling of the energy flexibility is done by the application of machine learning and data analytics algorithms for determination of the relevant EFKF for each energy-flexible system. Therefore the input of this step are the preprocessed data f[k]and the output are the EFDM with the determined EFKFs. The exact procedure and the individual steps, cf. Figure 5.1, are described in the following Section 5.1.3.

The *Energy Flexibility Model Validation* is the step where the automated parameterized EFDM is validated. This step is a part of the energy flexibility model setup sub-process, as shown in Figure 5.3. The results of the determined energy flexibility key figure (EFKF) must be evaluated for correctness regarding the complete EFDM into account. On the one hand, this can be done manually by experts with sufficient knowledge of the domain and the system. On the other hand, it is possible to perform this automatically using validation procedures. The



Figure 5.3: Energy flexibility model setup sub-process (own figure).

validation of the EFDM can be done by simulations as well as on real systems. Various conditions have to be evaluated [67]:

- Accuracy: Are the EFKF of the energy flexibility model correct, and do they meet the success criteria?
- Feasibility: Are the process parameters and technical limits met?
- Robustness: Which process and system tolerances are considered to react to disturbances and process deviations?
- Explainability: Are the EFDM parameterization results transparent?

Findings and possible improvements identified in this step are fed back to the previous *data verification* and *energy flexibility modeling* steps via feedback loops. This ensures the successive improvement of the EFDM quality by adjusting the EFKF. If the validation is successful, a deployment of the energy flexibility model must be tested.

The *Energy Flexibility Model Deployment* is the application step, where the previously automated parameterized EFDM is applied. For this purpose, the EFDM is transferred to operational mode. The parameterized EFDM must be checked for actual feasibility and acceptance by the user. A deployment strategy should be applied. This means that it is defined on which instance the EFDM are stored, how the resulting flexible load measure is to be transferred to the systems for energy-flexible operation, and how the flexible load measure is implemented by the energy-flexible system. This can be done manually by the machine operator or automatically via the machine's Programmable Logic Controller (PLC). Identified obstacles, system behavior, or errors that have not yet been considered should be fed back to the previous *data verification* step to adapt the key figures of the EFDM accordingly. In addition, considering an automated fallback mechanism, as we shown in [93], should safeguard the implementation. This ensures safe system operation by automatically switching off the flexible load measure to the reference case.

The *Monitoring and Maintenance* step is essential for maintaining the correctness of the EFDM and the automated determined corresponding EFKFs. Monitoring and updating the EFDM is necessary due to process changes, adjustments to the operating strategy, incorrect data transfers, or wear and tear. This step is not limited to monitoring the EFKF of the EFDM but is an extension of the *controlling and monitoring* step in the layer one. The update, to take account of identified changes, must be carried out while running through the other layer one steps, including the *optimization* step. This is the only way to identify and improve higher-level systematic changes.

5.1.3 Layer Three - Energy Flexibility Modeling

The third layer, the *Energy Flexibility Modeling layer* of the DD-EFMod method describes the process of how individual energy flexibility key figures (EFKF) can be automatically determined and so the energy flexibility model is parameterized. This layer is divided in two sub-processes, the *energy flexibility key figure determination* (Figure 5.4) and the *energy flexibility model synthesis* (Figure 5.5). For this purpose based on the input from the previous *data-preparation* step in layer two, a distinction is made between the available data type, streaming data or batch data, in the *algorithm selection* sub-step. Based



Figure 5.4: Energy flexibility key figure determination sub-process for automated determination of a single energy flexibility key figure of the EFDM (own figure).

on this, an appropriate machine learning or data analytics algorithm is applied to automatically determine the relevant key figures of the EFDM. Adapted from the CRISP-ML(Q) process, the quality requirements (defined in Section 4.3) of the machine learning algorithm must be evaluated and, if necessary, improved by adjusting the algorithm parameters. If the machine learning model is of sufficient quality, the EFKF is parameterized and evaluated. If requirements meet the actual EFKF correctly, the entire energy flexibility model can build in the step *energy flexibility model building* which is then transferred back as the output to the second layer. If the EFKF identified by the machine learning or data analytics algorithm is incorrect, a manual EFKF calculation should be performed. If this is also not possible, e.g., due to missing information or data, the process of this layer three *energy flexibility modeling* must be started again with new input data. Each sub-step of layer three is explained in detail in the following.

Algorithm Selection is based on the previous steps of data verification and data preparation, the type of machine learning or data analytics algorithm \mathcal{A} to be used for the automated determination of the EFKF is selected. Conventional offline batch machine learning or data analytics algorithms can be used if sufficient historical data are available. In the other case, where only temporary measurement data were available, online data stream mining algorithms can be selected (see Chapter 2). In addition, existing framework conditions such as computing capacity and robustness must be considered when selecting an algorithm.

Algorithm Parameter Selection is the sup-step, which defines the measured data and features for the selected algorithm. In addition, quality criteria for the evaluation of the selected machine learning or data analytics algorithms must be selected (cf. Chapter 2). In this step the selected algorithm \mathcal{A} results in a specific model \mathcal{M} .

Model Deployment includes training and applying the selected machine learning or data analytics model \mathcal{M} . Depending on the energy-flexible system and the EFKF to be determined, various combinations of algorithms may be possible. This step is based on the deployment according to [67]. If the internal algorithm finishes the calculation and iterations, the specific machine learning or data analytics model \mathcal{M}' results. With this model \mathcal{M}' the automated determination of the EFKF can derived.

Model Quality Sufficient is used as the sub-step to evaluate the deployed machine learning or data analytics \mathcal{M}' and thus decide whether the results of the algorithms are usable. According to [67], the quality criteria of reproducibility, performance, robustness, accuracy, and explainability can be included in this model evaluation. If all criteria and requirements are met, the next step is to determine the corresponding EFKF from the output **Y**. Otherwise, if the model quality does not meet the required conditions, a new training with adjusted algorithm parameter selection must be carried out.

The sub-process energy flexibility model synthesis (see Figure 5.5) starts with the step **EFKF** Parameterization. This step evaluates the result **Y** of previous data-driven automated *EFKF* determination sub-process based on the selected machine learning or data analytics algorithms for a single EFKF. For this purpose, the output variable of the *EFKF* determination sub-process **Y** is assigned to the corresponding energy flexibility key figure \mathcal{K}_i and its attributes according to the energy flexibility model, in the case of this work the EFDM [29] as



Figure 5.5: Energy flexibility model synthesis sub-process to assign the machine learning or data analytics model output \mathbf{Y} to the corresponding energy flexibility key figure \mathcal{K}_i , validate them, and parameterize the EFDM (own figure).

shown in Appendix A.3, for further processing.

The step of the **EFKF** validation is to check the respective specific key figures or their attributes for correctness. The evaluation is based on the defined criteria (see Chapter 4). The validation can be found on expert knowledge, simulations, or manual calculations. The main part is to check that the operating limits b_1 as the lower limit and b_2 as the upper limit are met for a determined key figure \mathcal{K}_i . This can be done using

$$b_1 \le \mathcal{K}_i \le b_2. \tag{5.5}$$

In the case that several operating conditions $\mathbf{B} = \{b_1, b_2, \dots, b_n\}$ are to be fulfilled, a logical operation

$$(\mathcal{K}_i > b_1) \land (\mathcal{K}_i < b_2) \land \dots \land (\mathcal{K}_i > b_n)$$
(5.6)

can be used for validation. In the event of successful validation, ${f K}$ can

be used as a vector with the individual key figures in the next sub-step of the synthesis process, see Figure 5.5.

The *Energy Flexibility Model Building* step is to span the flexibility space \mathcal{F} . In this process, this \mathcal{F} is converted into the corresponding standardized data format. In this work, the JSON schema format published in Lindner *et al.* [29] describes the energy flexibility space as EFDM. Therefore our developed external service the EFDM GUI, can be used [94] (see Appendix A.1.1).

If, on the contrary, the validation of a K_i cannot be carried out successfully, it must be checked whether a **manual calculation** of the corresponding key figure is possible. If not, the modeling process must be restarted as shown in Figure 5.1, starting with the *data preparation* step. If a manual determination of the key figure is possible, as shown in our previous work [30], the flexibility space \mathcal{F} can be spanned on this basis.

After completion of the *energy flexibility model building* sub-step, the modeling of the energy flexibility is completed, and the automated parameterized EFDM is transferred to the *application and implementation* (see Figure 5.1). The EFDM is transferred to the second layer *energy flexibility model validation* step (see Section 5.1.3).

The developed structure of the DD-EFMod method enables a procedure to realize the automated parameterization of the Energy Flexibility Data Model by determine the relevant energy flexibility key figures of the Energy Flexibility Data Model with the application of machine learning or data analytics algorithms. A software concept is presented in the following section to demonstrate its feasibility and prototypical implementation.

5.2 Software Concept

To ensure the applicability of the DD-EFMod method, I developed a corresponding software concept. This is visualized in the Unified Modeling Language (UML) component diagram in Figure 5.6. The



Figure 5.6: Software concept for prototype implementation as UML component diagram (own figure).

software concept corresponds to a prototypical implementation of the developed DD-EFMod method. The focus here is on demonstrating that implementing the method can be realized under consideration of the functional requirements defined in Chapter 4. The software concept is structured around several key components:

- Energy-flexible system, referred to as the *EF System* component, which serves as the main data source. Data and information are transmitted to a central database via corresponding interfaces (e.g., Open Platform Communications Unified Architecture (OPC UA)).
- The Database allows various data formats, such as time series (CSV), status data (CSV), machine data (Portable Document Format (PDF)), and the Energy Flexibility Data Model (EFDM)

as JSON format to be stored and managed.

- Data and information can be transferred via interfaces from the database and the energy-flexible system to the DD-EFMod structure (realized in python programming language (see Section 6.2)). This is made up of the main components:
 - data verification
 - data preparation
 - energy flexibility modeling
 - energy flexibility model synthesis

Each sub-component corresponds to the DD-EFMod method steps described above in Section 5.1. Furthermore, an interface to the central database is provided for the setup component to save the determined EFDMs.

- Interaction with the external service EFDM Graphical user interface (GUI) can transform the determined key figures of the EFDM in the standardized JSON structure. This external service is explained more in detail in Appendix A.1.1.
- A defined flexible load measure can be transmitted via an interface to the energy flexible system PLC. This means that the system can implement a flexible load measure derived from the EFDM.

By using appropriate interfaces (like our previuos work [95]), the database can also be enriched with information from external services, including EMS (energy management system), ERP (enterprise resource planning), or the energy market. This information can be used in the automated determination of the key figures of the EFDM.

5.3 Interim Summary

The primary objective of this work is to simplify and automate the enabling process for energy-flexible operation of production infrastructure (see Chapter 4). To achieve this, the DD-EFMod method, as shown in Figure 5.1, has been developed to facilitate the necessary modeling of energy flexibility through data-driven approaches. This method integrates and adapts the CRISP-ML(Q) process, extending the VDI process of enable factories to energy-flexible operation during the application and implementation step. The application and implementation layer of the DD-EFMod method specifically use the different steps data verification, data preparation, and the energy flexibility *model setup* to describe energy flexibility, reducing the need for manual intervention. The energy flexibility modeling layer further automates the process by determining the individual energy flexibility key figures (EFKF) of the Energy Flexibility Data Model using suitable machine learning and data analytics algorithms. The DD-EFMod method provides structured steps to automate and simplify the modeling of energy flexibility. The DD-EFMod method provides structured steps to automate and simplify the modeling of energy flexibility, ensuring efficient and effective implementation of energy flexibility. Based on the developed method with the individual steps, I derived a software concept for feasibility. This software concept consists of modular subcomponents and has software and hardware interfaces to enable interactivity between external software services and the energy-flexible system under consideration. The method development and the software concept development were carried out considering the requirements and quality criteria (see Section 4.3). The following describes how these functional requirements on the method (FR-M), functional requirements for energy flexibility modeling (FR-EF), and quality requirements (QR) are addressed:

- The applicability of the methodology (FR-M 1) is addressed by the software concept, using standardized interfaces and programming languages. In addition, the applicability is also supplemented by the fulfillment of the other criteria.
- The chosen division into layers, sub-steps and into corresponding software components and steps ensures modularity (FR-M 2) so

that only sub-steps can be executed if required.

- The requirement of transferability (FR-M 3) is met, as the method can be used to calculate the key figures of a model to describe energy flexibility, which in turn can represent individual energy-flexibility measures.
- The reproducibility of the results when applying the method (FR-M 4) is ensured by iteratively running through the individual sub-steps and saving the output data and results in the database.
- The feasibility (FR-EF 1) and accuracy (FR-EF 2) of the automated determined EFKF and the parameterized energy flexibility model can be ensured by the sub-process of the *energy flexibility model setup* through validation, deployment, and monitoring of the energy flexibility model.
- The *EFKF validation* step built into the method ensures that the energy flexibility key figures consider the possible identified maximum limits (FR-EF 3 and FR-EF 5) and are also explainable and comprehensible (FR-EF 4).
- The defined criteria for the machine learning and data analytics algorithms used (QR 1-5) to determine automatically the EFKF can be considered in the *algorithm selection* step and checked for their quality as part of the *model deployment* step and machine learning and data analytics *model quality sufficency* evaluation.

The validation of the DD-EFMod method is carried out in the following chapters to check the applicability of the DD-EFMod method and the associated software concept. I will first discuss the use case in the next chapter, including the machines and components involved (Chapter 6). This is followed in Chapter 7 by the implementation and validation under consideration of the results.
6 Use Case and Application at the ETA Research Factory

"Almost everything is like a machine."

Ray Dalio

This chapter describes the use case after showing the development of the method in the previous chapter. For this purpose, I first describe the machines used in the use case in Section 6.1 and discuss their systematic technical relationship. Then, in Section 6.2, I describe how the machines are connected to aggregate all relevant data from the different systems. This is followed by the description of four scenarios which focus on validation of the method in Section 6.3.

6.1 Process and Machines of the Use Case

To ensure the practical applicability of the developed DD-EFMod method for data-driven energy flexibility modeling, I evaluated the approach at the ETA Research Factory. The ETA Research Factory is located at the Technical University of Darmstadt and represent a real-scale research factory [96]. At the ETA Research Factory a holistic approach for an energy-efficient factory is realized. Energy efficiency is achieved by integrating the production machines, the technical building service systems, and the building shell in a thermal network. This factory setup aligns well with the RAMI 4.0 framework, which emphasizes the integration of digital and physical processes across different layers of a manufacturing system. The research factory is equipped with an exemplary production process chain for metal processing and the required technical building service systems. This production process contains the steps, lathing, cleaning, heat treatment, grinding, and final cleaning as shown in Figure 6.1.

For the validation use case of the developed DD-EFMod, I focused on



Figure 6.1: Production process chain of the ETA Research Factory (own figure adapted from [96]).

[97].



Figure 6.2: Flexible loads of the use case (own figure).

VLC-100GT [98].

the grinding sub-process. I have chosen this sub-process for validation to examine a system with several flexible loads of different types, storages, and dependencies. This allows a validation of the applicability of the DD-EFMod method within a RAMI 4.0 structured environment under consideration of the defined functional requirements on the method (see Section 4.3).

The central system of the grinding process is the EMAG VLC-100GT [98] machine tool (see Figure 6.2b), which is integrated into the higherlevel production process. In the context of the production process, the machine is charged with workpieces from a material storage via support automation and grinded in the machining area. The finished component is then forwarded to the product storage, where it is stored and transferred to the next production step.

To ensure process stability and take energy efficiency aspects into account, the machine is supplied with cooling via the factory's central heating and cooling system. In the application case, central cooling is provided by the eChiller45-II cooling machine from Efficient Energy GmbH [97, 99] (see Figure 6.2a). Both machines are electrical consumers must be cooled while operating, and therefore they are connected to the building's central cooling supply system, which allows the buffer storage of the central cooling supply system to be considered



Figure 6.3: Use case process and systems their connections, information, and data flows between the sub-systems (own figure).

[100]. As mentioned in Section 4.3, a central prerequisite for the application of the DD-EFMod method is that all relevant components and systems for which an energy-flexibility measure is to be created provide the relevant data and information with the help of ICT systems (see prerequisite number two in Section 4.3). Therefore in the ETA Research Factory we implement a framework which we developed over the last years [95] to enable different machine types, various databases and monitoring systems implemented in the ETA Research Factory. This framework facilitates the data integration and management required for effective energy flexibility modeling, reflecting the hierarchical and functional structuring proposed by RAMI 4.0 (see Chapter 2). All relevant components and their energy supplies and data connections of my validation use case are shown in Figure 6.3.

6.2 Hardware and Software Enabling

As mentioned in Section 4.3, this work focuses on the parameterization of energy flexibility models and the automated determination of technical energy flexibility key figures. This is possible by integrating the software concept presented in Section 5.2. Therefore, the hardware capability of all relevant systems is assumed to be fulfilled (see prerequisites 1-4 in Chapter 4). We allready presented the hardware enablement in Fuhrländer-Völker *et al.* [93]. Also refer to Bauer *et al.* [101] as well as to the results of Sossenheimer [27] and Fuhrländer-Völker [102]. To address the functional requirement on the method number one for the applicability of the DD-EFMod method as part of the implementation of my software concept, I used the open-source standard libraries listed in Table 6.1. The method was implemented in

- 、	,		
Software	Version	Reference	
Python	3.10.11	[103]	
scikit-learn	1.3.2	[104]	
scipy	1.11.3	[105]	
numpy	1.26.2	[106]	
pandas	2.0.3	[107][108]	
river	0.21.0	[109]	
Hardware			
System	Lenovo 20WE		
Operating system	10.0 Pro 64 Bit		
Processor	Intel(R) Core(TM) i7 2.80GHz		
Memory	1	.6 GB	

Table 6.1: Software and hardware used for implementation of the software concept (own table).

the specified open-source Python libraries to enable the broadest possible applicability without the vendor lock effect of software licenses. The implementation was carried out using the hardware also set in Table 6.1. This is a standard workstation, which also ensures simple implementation and transferability. There is no need for special highperformance computers or cloud applications, as at least the prototype implementation in the context of this work does not require intensive computing power. This work focuses on determining and calculating the technical key figures describing energy flexibility. Four different scenarios are discussed below.

6.3 Scenario Definition

All relevant components, energy supplies, and data connections of my validation use case are shown in Figure 6.3. As shown, the processes and systems that should operate energy-flexible are the grinding machine tool and the cooling machine, which extend their flexible potential by considering product storage and cold storage. These systems and their connections result in the following scenarios, which are used to validate developed DD-EFMod method:

- Scenario 1: The focus is on the EMAG VLC-100GT machine tool to show that the method works for a machine modeled as the EFDM class flexible load. I will also validate the deployment of batch and data stream machine learning algorithms for automated determination of the EFKF.
- Scenario 2: This extends scenario one by considering the machine tool EMAG VLC-100GT combined with the product storage to show the DD-EFMod methods application for the EFDM class storages.
- Scenario 3: Validation of the transferability of the method is demonstrated with this third scenario. For this pur-

pose, the DD-EFMod method is transferred to another machine type, the cooling machine eChiller45-II, and the cold storage.

Scenario 4: The focus here is on two flexible loads, the machine tool EMAG VLC-100GT together with product storage and considering the cooling machine eChiller45-II with the cold storage to validate the DD-EFMod capability for automated determination of the EFKF of the EFDM class dependency.

In addition to the four scenarios mentioned above, an evaluation of the potential energy and cost savings that are possible through the use of energy-flexibility measures defined as flexible load measures, based on the automated parameterized EFDM, on different energy markets is carried out. The implementation and application, as well as the discussion of the results, are discussed in the next Chapter 7.

6.4 Interim Summary

This chapter first describes the use case for the validation of the DD-EFMod method I developed as part of this thesis. For this purpose, the process chain of the ETA Research Factory was described, and the individual machines, the machine tool EMAG VLC-100GT, and the cooling machine eChiller45 are relevant. I also listed the appropriate software that I used while implementing the software concept of the DD-EFMod method. This was followed by defining the four scenarios used to validate the method.

7 Implementation and Validation

"An experiment disproving a prediction is discovery."

Enrico Fermi

This chapter shows the validation of the developed DD-EFMod method, which was applied to the use case and four different scenarios (see section 6.3). A separate system boundary is considered for each scenario in order to test different applications of the method. For this purpose, each scenario is divided into three sub-steps. First, the framework conditions of the scenario and the energy flexibility key figures to be determined are discussed. This is followed by an explanation of the implementation based on the sub-steps using the DD-EFMod method. Finally, the results of the individual sub-steps for each scenario are presented and discussed. Following the four scenarios, the implementation of the energy flexibility measures change processing sequence and shift start of job defined as flexible load measures based on the automated parameterized Energy Flexibility Data Models of the four scenarios is validated. This additionally evaluates the implementation of the DD-EFMod method in practice. The chapter concludes with a summarizing interim conclusion. To ensure sufficient traceability of the validation of this work, all measurement data, implemented algorithms, and results used in this work are documented and made available as a publicly accessible TUdatalib repository [110].

7.1 Scenario 1 - EMAG GT

The focus of this scenario is on the EMAG VLC-100GT machine tool (see Chapter 6) to show that the method works for the EFDM class flexible load. In this scenario, I will also investigate the use of offline and online streaming machine learning algorithms to calculate the corresponding EFKF.

Based on the method (see Figure 5.1), the first step is to determine which energy-flexible system is to be examined (layer 1). The system boundary of this scenario in the use case presented in the previous chapter is shown in Figure 7.1 and is defined around the machine tool EMAG VLC-100GT itself. This means that only the energy supply, particularly the electrical power consumption, is considered as a sim-



Figure 7.1: System of scenario 1 (own figure).

plification in this scenario to analyze and determine energy flexibility. The next step is to analyze and select the data required to determine the relevant technical EFKF for describing the energy flexibility of the machine. For this purpose, an exemplary electrical power consumption of the machine, shown in Figure 7.2 recorded on 31.08.2023, is examined as a starting point and reference case operation load profile.

7.1.1 Key Figure Definition

The *data verification* begins with the definition of the relevant EFKF. Assuming that the EFDM is used to describe energy flexibility, as described in Chapter 2.1.2. In the following, it is assumed that the EFDM according to [29] is used to describe the existing energy flexibility. In the case of the EFDM class flexible load, the following selected submodel of technical key figures is shown in Table 7.1.

As discussed in the fundamentals chapter, not all key figures are always



Figure 7.2: Electrical power consumption of EMAG GT (own figure).

necessary. For this scenario, determining the power states is defined as the minimum criterion, and the other technical key figures are considered optional. This results in the vector of relevant EFKF

$$\mathbf{K} = [\mathbf{P}, T_{\text{Rea}}, T_{\text{V}}, T_{\text{H}}, \nabla P_{\text{Act}}, \nabla P_{\text{Mod}}, \nabla P_{\text{Dea}}, T_{\text{Reg}}].$$
(7.1)

I examined in the *data verification* step the measured variables and input data listed in Table 7.2 as the data basis for the next steps. In the first step, I examined the measurement variables mentioned based on system knowledge. The aforementioned prioritization could be carried out based on these. Based on this prioritization, the definition of the input variables follows

$$\mathbf{X}(t) = [P(t), s_{\mathrm{st}}, s_{\mathrm{wk}}, s_{\mathrm{op}}, s_{\mathrm{da}}, t_{\mathrm{pro}}]$$
(7.2)

to determine the flexibility space \mathcal{L} .

Algorithm 1 Data	Verification	and Preparation
------------------	--------------	-----------------

Input: Dataset of measured data **X Output**: Feature vector $\mathbf{f}[k]$

- 1: Load dataset \mathbf{X} ;
- 2: Analyze power values P(t):
- 3: Calculate min, max, and mean of P(t);
- 4: Apply IQR method to detect and filter outliers in P(t);
- 5: Compute Moving Average \mathbf{x}_{MA} of P(t) with window size M (Eq.: 2.11);

7.1.2 Implementation

The next step in layer 2 is *data preparation*. Here I have processed the input data from $\mathbf{X}(t)$ using standard methods (e.g., normalization, moving average, see Chapter 2), from which the respective feature vector $\mathbf{f}[k]$ is derived.

The example code of the implementation for this step is shown in Algorithm 1 as an example for $\mathbf{X}(t) = [P(t)]$. The results are shown and discussed in the next section. This is followed by the step of *energy flexibility modeling*, starting with the *algorithm selection* sub-step in

Class	Key Figure	Description	Feasibility Level
	Р	Power states	1
e	$T_{ m Rea}$	Reaction duration	2
	$T_{\rm V}$	Validity	2
cib] ad	$T_{ m H}$	Holding duration	2
Fley Lo	$\nabla P_{ m Act}$	Activation gradient	2
	∇P_{Mod}	Modulation gradient	2
	∇P_{Dea}	Deactivation gradient	2
	$T_{ m Reg}$	Regeneration duration	2

Table 7.1: Technical key figures in the EFDM for describing the flexibility space and their feasibility levels of scenario 1 following Lindner *et al.* [29] (own table).

()			
Measured Value	Symbol	Unit	Priority
Electric Power	P	W	•
Machine state - standby	$s_{ m st}$	binary	•
Machine state - working	$s_{ m wk}$	binary	•
Machine state - operational	$s_{ m op}$	binary	•
Machine state - disabled	$s_{ m da}$	binary	•
Processing time	$t_{\rm pro}$	s	•
Electric voltage	\overline{U}	V	\bullet
Electric current	Ι	А	\bullet
Frequency	f	Hz	٢
Machine temperatur	$T_{\rm pro}$	$^{\circ}\mathrm{C}$	Ο

Table 7.2: Priority of measured values of the EMAG GT for scenario 1 (own table).

layer three of the DD-EFMod. In this scenario, the focus is initially on automatically determining the power states of the EMAG GT, as this is the minimum requirement (see Table 7.1). For this purpose, a offline batch clustering algorithm is first selected to quantify the possible machine states and their associated power states based on historical data. As a clustering algorithm, I first select a batch K-means cluster algorithm (see Chapter 2). This was implemented on the basis of [58] in Algorithm 2 using [111]. The deployment of the algorithm is followed by the *model quality sufficient* validation step of the machine learning algorithm (see Figure 5.4). I have used the Algorithm 3 for this purpose. If the quality measures are sufficient, the *energy flexibility* model synthesis sub-process follows (see Figure 5.5). Thus, based on the clustering results, the *EFKF parameterization* of the power states **P** follows. One cluster c_i represents one power state. Each of these cluster centers is associated with a power state P_i , where $P_{i,\min}$ and $P_{i,\max}$ represent the corresponding minimum and maximum power values of the power state. This applies

$$P_i = [P_{i,\min}, P_{i,\max}] \tag{7.3}$$

Algorithm 2 Deployment: K-Means Clustering

Input: \mathbf{x}_{MA} with **C** clusters

Output: Custercenters \mathbf{C} , Clusterboundries \mathbf{C}_{B}

- 1: Apply K-means clustering with c_i clusters;
- 2: repeat
- 3: Assignment of the data from \mathbf{x}_{MA} to the respective cluster with the next cluster center (Eq.: 2.12);
- 4: Recalculation of the c_i cluster centers as centers of the objects of the individual clusters (Eq.: 2.13);
- 5: until Cluster partition remains unchanged;
- 6: Initialize indentify cluster boundaries;
- 7: for each data and center in \mathbf{C} do
- 8: Calculate lower boundary as 10% below the center as $c_{i,\min}$;
- 9: Calculate upper boundary as 10% above the center as $c_{i,\max}$;

10: end for

where

$$P_{i,\min} = c_{i,\min} \tag{7.4}$$

$$P_{i,\max} = c_{i,\max} \tag{7.5}$$

for all clusters and power states. These must then be checked according to the *energy flexibility model synthesis* and saved in the corresponding EFDM data format. Once the power states have been calculated, the minimum requirement for determining the power states is fulfilled (see Table 7.1).

To determine the other EFKFs, the power states already identified were the basis. For this purpose, the sub-processes of EFKF determination

Alg	gorithm 3 Model Quality: Clustering Quality
	Input: C' clusters
	Output : Cluster Performance Values
1:	Calculate the Silhouette Score for the data and clusters (Eq.: 2.20);
2:	Calculate the Davies-Bouldin Score for the data and clusters (Eq.:
	2.22);

Alg	gorithm 4 Deployment: Power State Duaration Analysis
	Input : Power Data $P(t)$, Operation states S
	Output: Intervals of Power States I, Power State Durations D,
1:	Allocation machine state temporal intervals \mathbf{I} ;
2:	Define power states \mathbf{S} ;
3:	for each mode s_i in S do
4:	Identify continuous intervals I_s where
	machine state s is constant;
5:	Record start and end indices of each interval I_s ;
6:	$\mathbf{I}[s_i] \leftarrow I_s;$
7:	end for
8:	Analyze Durations of I:
9:	Calculate duration for each interval in \mathbf{I} ;
10:	Identify unique power states in D .
11:	Determine minimum and maximum duration $T_{\rm H}$
	for each unique power state.

(see Figure 5.4) and energy flexibility model synthesis (see Figure 5.5) are rerun for each EFKF to be determined (see Equation 7.1). To identify the key figure power state duration $T_{\rm H}$ of the grinding machine, the different machine operation states are defined as **S** with $s_i \in \{Working, Operation, Standby, Disabled\}$.

The automated determination of the key figures holding duration is done by deployment of Algorithm 4. Here, **D** describes an unspecific data vector with all holding durations $T_{\rm H}$.

To determine the key figures power gradients $(\nabla P_{\text{Act}}, \nabla P_{\text{Dea}}, \nabla P_{\text{Mod}})$ for each power state change, the gradients are calculated with

$$\nabla \mathbf{P} = \frac{(P_{end} - P_{start})}{t_{end} - t_{start}}.$$
(7.6)

The identification of the power gradients is done by Algorithm 5. Thereby \mathbf{T} is an unspecific data vector for all transitions.

To automatically determine the key figure reaction duration T_{Rea} , it is necessary to analyze the processing time of a workpiece. This is based on the assumption that the shortest time horizon of a energy-flexible

Alg	gorithm 5 Deployment: Gradient Analysis
	Input : Power Data $P(t)$, Power States Transitions T
	Output : Average Gradients for Transitions
1:	Determine current power state gradient for value:
2:	for each transition pair (fromState, toState) in \mathbf{T} do
3:	for each Data Value in $P(t)$ do
4:	Check if power state transition occurs;
5:	if transition starts then
6:	Record start power and time;
7:	else if transition ends then
8:	Record end power and time;
9:	Calculate gradient (Eq.: 7.6);
10:	end if
11:	end for
12:	Calculate average gradient $\nabla \mathbf{P}$ for the transitions;
13:	end for

operation is based on the electricity market. For the Intraday energy market, this is divided into $T_{\text{Intraday}} = 15 \text{ min sections [112]}$. This means that the cycle time corresponds to $t_{\text{cycle}} = 15 \text{ min}$. From the analysis of the load profiles of the machine tool (see Figure 7.2) and the process times t_{pro} , the following relationship emerges

$$t_{\rm pro} = \frac{t_{\rm cycle}}{n}$$
 with $t_{\rm cycle} = T_{\rm Intraday}$ (7.7)

with \boldsymbol{n} number of processed parts, which directly results in the maximum reaction duration

$$T_{\text{Rea}} = t_{\text{pro}}.\tag{7.8}$$

The EFKF for regeneration duration T_{Reg} is also use case or scenario specific. For the machine tool of the use case, assuming that the machine goes into regular operation after an energy-flexible operation or is in standby mode, then

$$T_{\text{Reg}} = \max(t_{\text{pro}}) \tag{7.9}$$

if a machining process has to be terminated.

The EFKF validity $T_{\rm V}$ is defined manual. I choose this approach because it is a very benefit-specific key figure. This primarily influences the period the energy-flexible system under consideration can be switched to energy-flexible operation (see Appendix A.3). This can be, for example, the next hour, the duration of a shift, a specific day, or even several days.

As described in the previous chapter (see Chapter 5) and shown in the software concept, the building and transfer of the key figures, which are automated determined with the mentioned machine learning or data analytics algorithms, into the standardized EFDM format takes place in the step *energy flexibility model building*. For this work, with a focus on the DD-EFMod, I use the external software service EFDM GUI [94] for building the standardized EFDM in JSON schema (see Appendix A.1.1). Flexible load measures can then be derived and further processed from the defined flexibility space.

Alternative Implementation with Data Streaming Algorithms

To address the second research hypothesis (see Chapter 4), I also investigate the applicability and feasibility of online machine learning

Algorithm 6 Deployment: Mini-Batch K-Means Clustering			
Input : Data Stream $\mathbf{X}_{\text{scaled}}$, number of clusters k, batch size n			
Output: Cluster centers C, Predicted clusters P			
1: Initialize mini-batch K-means with k clusters and batch size n			
2: for $i = 0$ to len(\mathbf{X}_{scaled}) step n do			
3: batch $\leftarrow \mathbf{X}_{\text{scaled}}[i:i+n]$			
4: Perform partial fit of mini-batch K-means on batch (Eq.: 2.14			
5: end for			
6: Predict clusters for all data in $\mathbf{X}_{\text{scaled}}$ to get \mathbf{P}			
7: Retrieve cluster centers \mathbf{C} from mini-batch K-means			
8: Inverse transform cluster centers if necessary			
 7: Retrieve cluster centers C from mini-batch K-means 8: Inverse transform cluster centers if necessary 			

Algorithm 7 Deployment: CluStream Clustering **Input**: Data stream \mathbf{X} , window size wOutput: Cluster centers C, Predicted clusters P 1: Load dataset X 2: Compute moving average of **X** with window size w3: Initialize CluStream 4: for each value x in the moving average do Update CluStream with x (Eq.: 2.15 - 2.19) 5: 6: end for 7: Simulate data stream 8: for each data point x in X do 9: Assign x to the nearest cluster center (Eq.: 2.12) 10: end for 11: Compute cluster centers \mathbf{C} and predicted clusters \mathbf{P}

methods as part of the first scenario. Since the automated determination of the EFKF of the power states is the minimum requirement, the focus of testing data stream mining algorithms is on this key figure. The batch incremental learning clustering method mini-batch K-means (see Algorithm 6) [59] and the instance incremental method CluStream (see Algorithm 7) are selected as standard clustering algorithms for data stream mining problems [56]. The selection of these two methods was based on the fact that they are proven standard methods from data stream analysis and on the open-source availability of the algorithms as well. Furthermore, the basic principles of the two methods are similar to the batch K-means clustering algorithms wich performed well for the automated parameterization of the key figure power states in the context of the DD-EFMod method in scenario 1. For this very reason, the three methods are comparable. The respective implementation of the selected data stream algorithms is part of the DD-EFMod within the *EFKF determination* sub-process (see Figure 5.4). Due to technical limitations (network failures, excessive latencies), the three algorithms were compared using two data sets: a measurement data set and a simulated data set with significantly increased measurement points. The results are listed and evaluated in the next section.

7.1.3 Evaluation and Results

The results and their evaluation of the implementation of the DD-EFMod method based on the software concept (see Figure 5.6) of the first scenario using the algorithms discussed in the previous section are presented in this section.

Based on the known reference case operation load profile of the EMAG GT (see Figure 7.2), the *data verification* and *data preparation* step (Algorithm 1) was applied. The data profile of the dataset from the load profile consists of 7595 samples with a sampling rate of 1 second, result-

Table 7.3: Results of data verification and preparation of scenario 1 (own table).

Value	Result	Unit
Minimum	-7718.63	W
Maximum	12681.04	W
Mean	1634.01	W
12.5		
10.0		
7.5		
≥ 5.0		
2.5 A	chichard hame	
_2.5		
-2.0		
-5.0		
-7.5		
0 1000	2000 3000 40 Time in	00 5000 6000 7000

Figure 7.3: Preprocessed electrical power consumption of EMAG GT (own figure).

Denoised Signal

Moving Average

Original Signal

ing in a dataset size of 0.370 MB. The data was sourced from recordings in the machine's PLC system and contained no missing values. The power consumption data exhibits strong transient behavior, which is typical for this machine's operation, as illustrated in Figure 7.2. This transient nature highlights the dynamic changes in power usage during the machining process, necessitating careful denoising and smoothing to obtain accurate characteristic values for subsequent analysis. The calculated characteristic values resulted from the *data verification* and *data preparation* step shown in Table 7.3 and the signal curves for the denoised power consumption and the power consumption signal with moving average (with window size M = 50) shown in Figure 7.3.

Figure 7.3 clearly shows the necessity of the *data verification* and *data preparation* step. The original measured load profile repeatedly shows power transients that are part of the normal operating behavior (e.g. spindle movements). However, these must be suppressed to enable the automated determination of the key figures like the power states with machine learning algorithms in a targeted manner. The resulting moving average $\mathbf{x}_{\mathbf{MA}}$ power signal achieves this because it is free of extreme outliers. Based on this, I could apply the Algorithm 2 as the *EFKF determination* sub-process. For this purpose, the batch K-means algorithm was executed with the signal $\mathbf{x}_{\mathbf{MA}}$ and the specification $\mathbf{C} = 3$. Here, $\mathbf{C} = 3$ is derived from the machine states relevant for the analysis (see Table 7.2). For this purpose, I have combined the states s_{st} and s_{op} as one common state. Thus

$$c_1 = s_{\rm st} \cup s_{\rm op} \tag{7.10}$$

$$c_2 = s_{\rm da} \tag{7.11}$$

$$c_3 = s_{\rm wk} \tag{7.12}$$

follows as individual clusters to be determined, representing the power states. The combination of c_1 is based on the operating behavior of the machine. It was shown that the EMAG GT automatically switches from mode $s_{\rm op}$ to $s_{\rm st}$ after 3min if there is no further manual state



Figure 7.4: K-means clustering results EMAG GT (own figure).

change. Thus, the outcome of this *model deployment* step is shown in Figure 7.4 results from implementing Algorithm 2 for the automated *EFKF determination* step for the key figure power states. Within the *model quality sufficient* step, the results of the batch K-means clustering algorithm are used to validate and analyze the statistical consistency and homogeneity of the clusters. This is shown by the values given in Table 7.4 and illustrated in the corresponding boxplot in Figure 7.5. As the figure and table show, the cluster centers determined for each

Table 7.4: EMAG GT K-means clustering statistical results (own table).

Cluster	Standby	Disabled	Working
Center	1172.91	222.93	2081.26
Min	704.66	214.03	1627.49
Q1	1010.07	214.24	1819.01
Median	1119.5	214.4	2016.22
Q3	1327.58	214.64	2328.25
Max	1626.45	689.32	2658.07



Figure 7.5: Boxplot of the clusters formed for the EMAG GT (own figure).

cluster are close to the median of the associated data point.

To further test the cluster quality, I determined the quality measures of the Silhouette Score and Davis-Bouldin Score (see Chapter 2). The implementation of the Algorithm 3 leads to the results in Table 7.5, which confirm the good quality of the clustering algorithm in this case. Thus, the automatically determined cluster centers from Table 7.6 can be used as the EFKF power states.

Table 7.5: Quality measures of batch K-means clustering (own table).

Quality measure	Value	
Silhouette Score Davis-Bouldin Score	$\begin{array}{c} 0.68\\ 0.37\end{array}$	

As part of the step *EFKF parameterization* it follows

$$P_{\text{standby}} = [1055, 1290] W$$
 (7.13)

$$P_{\text{disabled}} = [200, 245] W$$
 (7.14)

$$P_{\text{working}} = [1873, 2289] W \tag{7.15}$$

as the elementary key figure power states because these EFKFs are valid.

To further refine the Energy Flexibility Data Model and enhance its applicability, it is essential to extend the parameterization by incorporating optional key figures. This leads to revisit the steps of *EFKF determination* and *energy flexibility model synthesis*. Algorithm 4 is implemented in order to assign a possible holding duration to each power state, which results in the values given in Table 7.7.

Thereby $\max(T_H^{\text{standby}})$ is determined from the operating behavior of the machine tool, which automatically switches from the standby state to the P_{disabled} state at 30 min of inactivity. For $\min(T_H^{\text{working}})$, the 90 s corresponds to the processing time of a process step (see equation 7.7) of the machine tool EMAG GT.

The next rerun of the *EFKF* determination sub-process with implementing the Algorithm 5 for automated determining the EFKF of the power gradients results in the mean gradients as shown in Table 7.8. Analyzing this results as part of the energy flexibility model synthesis sub-process, no transition from P_{disabled} to P_{standby} could be measured

		0(1	
Value	Cluster Standby	Cluster Disabled	Cluster Working
Cluster Center Upper bound Cluster	1172.91 W 1290.20 W	222.93 W 245.23 W	2081.26 W 2289.39 W
Lower bound Cluster	$1055.62 \mathrm{~W}$	$200.64~\mathrm{W}$	$1873.14 \mathrm{~W}$

Table 7.6: Results of batch K-means clustering (own table).

Value	$T_H^{ m standby}$	$T_H^{\rm disabled}$	$T_H^{ m working}$
Minimum	66 s	0 s	90 s
Maximum	$1800\mathrm{s}$	inf	$1365\mathrm{s}$

Table 7.7: Identified holding durations of EMAG GT (own table).

Table 7.8: Identified power gradients of EMAG GT (own table).

Transition	Value
P_{disabled} to P_{standby}	$0.0 \frac{W}{s}$
P_{standby} to P_{working}	$320.0\frac{W}{s}$
$P_{\rm working}$ to $P_{\rm standby}$	$-13.8 \frac{W}{s}$
P_{standby} to P_{disabled}	$-804.5 \frac{W}{s}$

in this data set. This was due to the technical network problems that had already occurred. However, it can be deduced from all the tests carried out, and the other state changes that the state changes are instantaneous and, therefore, immediate. This is typical behavior for an electrical consumer. This can be seen, for example, in the activation P_{standby} to P_{working} with 320.0 $\frac{\text{W}}{\text{s}}$ or the deactivation of P_{standby} to P_{disabled} with $-804.5 \frac{\text{W}}{\text{s}}$.

Finally for this scenario, the EFDM is further extended by determine the key figures T_{Rea} and T_{Reg} which can be determined with the equation 7.8 and 7.9. This results in $T_{\text{Rea}} = 90$ s and $T_{\text{Reg}} = 90$ s.

Overall, this creates the flexibility space as EFDM of the EMAG GT with the automatically determined key figures as shown in Table 7.9. I manually defined the key figure validity as an example for a day from 06:00 to 22:00.

This resulting flexibility space is shown in Figure 7.6. The space of possibilities for the implementation of a possible flexible load measure, which can be realized by the EMAG GT, is limited as a set with the smallest ($P_{\text{disabled}} = 0.200 \,\text{kW}$) and largest power value ($P_{\text{working}} = 2.289 \,\text{kW}$), the smallest ($T_H^{\text{disabled}} = 0 \,\text{s}$) and largest holding duration

).	
Key Figure	Min	Max
Р		
$P_{\rm disabled}$	$0.200\mathrm{kW}$	$0.245\mathrm{kW}$
P_{standby}	$1.055\mathrm{kW}$	$1.290\mathrm{kW}$
$P_{\rm working}$	$1.873\mathrm{kW}$	$2.289\mathrm{kW}$
$T_{\rm Rea}$	-	90 s
$T_{\rm V}$	2023-08-31 06:00:00	2023-08-31 22:00:00
$T_{ m H}$		
T_H^{disabled}	$0\mathrm{s}$	inf
$T_H^{ m standby}$	$66\mathrm{s}$	$1800\mathrm{s}$
$T_H^{ m working}$	$90\mathrm{s}$	$1365\mathrm{s}$
$\nabla P_{ m Act}$	-	$0.320 \frac{\text{kW}}{\text{s}}$
$ abla P_{ m Mod}$	-	$-0.013\frac{kW}{s}$
∇P_{Dea}	-	$-0.804 \frac{\text{kW}}{\text{s}}$
$T_{\rm Reg}$	-	90 s

Table 7.9: Automated determined key figures of the EFDM parameterized via the DD-EFMod method for the EMAG GT (own table).

 $(T_H^{\text{standby}} = 1800 \,\text{s})$ as well as the maximum rate of change $(\nabla P_{\text{Act}} = 0.320 \,\frac{\text{kW}}{\text{s}}).$

The results indicate that accurately determining power states is crucial for evaluating other energy flexibility key figure (EFKF). Consequently, assessing the prototypical implementation of online streaming data algorithms, specifically Algorithm 6 and 7, is deliberately focused on this key figure. This focused approach thoroughly examines the effectiveness of data streaming algorithms in real-time processing environments. By concentrating on the key figure power states, we can effectively demonstrate the algorithm's utility in capturing incremental changes and their potential applicability to other key figures in the context of energy flexibility.

The results of the implementation of the Algorithms 6 for mini-batch K-means and Algorithm 7 for CluStream, as part of the sub-process *en*- ergy flexibility key figure determination are given in Table 7.10. This automated determined clusters, which representing the key figure power states are based on a measured data set. In Appendix A.2 the graphical representation for each test run is given.

Comparing the performance of the clustering algorithms with the measured data set provides essential insights into their applicability. The batch K-means and mini-batch K-means algorithms are close to each other in their cluster centers for the power states Disabled, Standby,



Figure 7.6: Resulting flexibility space of EMAG GT represented quantity as a function of the flexibilizable power, duration and rate of change (own figure).

and Working, with mini-batch K-means showing a slightly higher value for the Working state. CluStream shows a significantly different cluster center for the disabled state and higher values for the standby and working states, indicating a different clustering sensitivity than the K-means variants. In this respect, the CluStream algorithm must be rated as critical. Regarding silhouette values, mini-batch K-means slightly outperforms the others, indicating better cluster cohesion and separation. The Davies-Bouldin index, which is lower for CluStream, suggests that CluStream may produce clusters with better separation.

From these results, it's evident that both K-means algorithms offers a good balance between computational efficiency and clustering quality for offline (batch) or online (data streams) clustering, as shown by its silhouette score and Davies-Bouldin Index. This is confirmed by the evaluation of the simulated significantly longer data set in Table 7.11 and Appendix A.2. Therefore, I select the batch K-means algorithm for the automated determination of the key figure power state in the

Value	Batch K-means	Mini-batch K-means	CluStream	
Disabled				
Cluster Center	$222.93\mathrm{W}$	291.11 W	$0.07\mathrm{W}$	
Standby				
Cluster Center	$1172.93\mathrm{W}$	$1151.05\mathrm{W}$	$1344.66\mathrm{W}$	
Working				
Cluster Center	$2081.26\mathrm{W}$	$2224.02\mathrm{W}$	$2236.13\mathrm{W}$	
Silhouttte-Score	0.68	0.69	0.67	
Davies-Bouldin-	0.37	0.36	0.34	
Index				
Samplenumber of	7595	7595	7595	
data set				
Computation time	$0.057\mathrm{s}$	$0.245\mathrm{s}$	$8.42\mathrm{s}$	

Table 7.10: Comparison of clustering algorithms performance relatingto measured data set (own table).

Value	Batch	Mini-batch	CluStream		
	K-means	K-means			
Disabled					
Cluster Center	$258.13\mathrm{W}$	$270.67\mathrm{W}$	$410.15\mathrm{W}$		
Standby					
Cluster Center	$1050.70\mathrm{W}$	$1107.04\mathrm{W}$	$0.00\mathrm{W}$		
Working					
Cluster Center	$2080.77\mathrm{W}$	$2199.71\mathrm{W}$	$2088.21\mathrm{W}$		
Silhouttte-Score	0.78	0.75	0.73		
Davies-Bouldin-	0.24	0.32	0.36		
Index					
Samplenumber of	167400	167400	167400		
data set					
Computation time	$0.474\mathrm{s}$	$4.686\mathrm{s}$	$19.65\mathrm{s}$		

next scenarios in the following sections.

 Table 7.11: Comparison of clustering algorithms performance relating to simulated data set (own table).

In conclusion, the validation and implementation conducted in Scenario 1 confirms the feasibility of automating the determination of energy flexibility key figures for the Energy Flexibility Data Model for machine tools as flexible loads using the DD-EFMod method. The findings demonstrate that determination of key figures possible with historical data, but also with data streams. Furthermore the importance of selecting and evaluating the appropriate algorithms to ensure accuracy and reliability of algorithms for the energy flexibility modeling is shown. This initial success underscores the potential of the DD-EFMod method in enhancing the operational flexibility of industrial environments.



Figure 7.7: System of scenario 2 (own figure).

7.2 Scenario 2 - EMAG GT and Product Storage

This scenario extends the system boundary of the EMAG VLC-100GT machine tool (see Section 7.1) to the product storage in which the processed, manufactured parts are stored. This is intended to demonstrate the applicability of DD-EFMod to the class storage of the EFDM (see Section 2.1.2) and product storage in particular and to show what additional energy-flexibility potential can be realized by taking material storage into account. The results and key figures for the machine tool are taken from the first scenario for these system considerations. This means that the focus of this scenario can be placed on validate the automated determination of the energy flexibility key figures of the EFDM for the storage system by apply the DD-EFMod method.

Class	Key Figure	Description	Feasibility Level
	$C_{\rm S}$	Usable capacity	1
	E_{t_0}	Initial energy content	1
ge	$S_{ m S}$	Suppliers	1
ora	E_{Tar}	Target energy content	2
$\mathbf{S}_{\mathbf{f}}$	$E_{\rm Loss}$	Energy loss	2
	E_{Drain}	Energy drain from storage	2

Table 7.12: Technical key figures in the EFDM for describing the flexibility space and their feasibility levels of scenario 2 following Lindner *et al.* [29] (own table).

7.2.1 Key Figure Definition

Following the steps of DD-EFMod method the *data verification* step starts with the definition of the relevant key figures of the EFDM to determine. As described in the fundamentals chapter, according to the definition of EFDM [29], a storage system can only be considered in conjunction with a flexible load. Therefore, the known key figures from Table 7.9, which are extended by the energy flexibility key figures (EFKF) shown in Table 7.12 to describe the class storage, result for the system consisting of EMAG GT and product storage. The minimum requirements to be determined are the technical key figures of the usable capacity, the initial energy content, and the associated supply system. Together with the EFKF of feasibility level 2, it follows

$$\mathbf{K} = [C_{\mathrm{S}}, E_{\mathrm{t}_0}, S_{\mathrm{S}}, E_{\mathrm{Tar}}, E_{\mathrm{Loss}}, E_{\mathrm{Drain}}]$$
(7.16)

as the vector of relevant EFKF for this scenario. Knowledge of the amount of energy in the product storage is relevant to determining the key figure storage capacity and its current state. The measured variables in Table 7.13 are pertinent to decide on this. In addition, the conditions defined in Table 7.14 are selected for this scenario. The values shown are derived from the fact that 10 raw materials pieces

()			
Measured Value	Symbol	Unit	Priority
Electric Power	P(t)	W	•
Processing time	$t_{ m pro}$	\mathbf{S}	•

Table 7.13: Priority of measured values of the EMAG GT for scenario 2 (own table).

can be processed by the EMAG GT within 15 min (see Section 7.1). As a further boundary condition, I asumed that the storage capacity consumed 40 produced parts to start the next processing step (see Figure 6.1).

Table 7.14: General conditions for the product storage (own table).

Term	Symbol	Value
Working shift time	$t_{ m shift}$	$8\mathrm{h}$
Cycle time	$t_{ m cycle}$	$15\mathrm{min}$
Storage capacity min	C_S^{\min}	20 pcs
Storage capacity soll	C_S^{soll}	40 pcs
Storage capacity max	C_S^{\max}	320 pcs

Based on this prioritization, the definition of the input variables follows

$$\mathbf{X}(t) = \left[P(t), t_{\text{pro}}, t_{\text{shift}}, t_{\text{cycle}}, C_S^{\min}, C_S^{soll}, C_S^{\max}\right]$$
(7.17)

to determine the flexibility space of the storage \mathcal{S} .

7.2.2 Implementation

The *data preparation* step for the following step of *energy flexibility* modeling for the product storage begins with the definition of framework parameters. Therefore the total energy requirement of the sup-

Algorithm 8 Deployment: Storage Capacity

Input: Raw Data $\mathbf{X}(\mathbf{t})$

- Output: Storage EFKF, 1: Load data from **X(t)**;
- 2: Define production cycle;
- 3: Calculate energy consumption per part (Eq.: 7.19);
- 4: Calculate storage capacity (Eq.: 7.20);
- 5: Calculate energy content (Eq.: 7.21);

plier of the storage during a production cycle is relevant with

$$E_{\text{total}} = \frac{\sum \left(\frac{P(t)}{1000} \cdot t_{\text{cycle}}\right)}{3600}$$
(7.18)

resulting in the energy per part with n number of parts per cycle

$$E_{\text{part}} = \frac{E_{\text{total}}}{n} \tag{7.19}$$

can be determined. The maximum and minimum storage capacity then follows with

$$C_{\rm S} = \left(C_S^{\rm min} \cdot E_{\rm part}, C_S^{\rm max} \cdot E_{\rm part} \right) \tag{7.20}$$

and equivalently the initial energy content by

$$E_{t_0} = \left(C_S^{\min} \cdot E_{part}, C_S^{soll} \cdot E_{part} \right).$$
(7.21)

For the scenario, the initial energy content is defined as the target energy content. The energy requirement E_{Drain} of the product storage depends on the production plan

$$E_{\text{Drain}} = f(E_{out}, t) \tag{7.22}$$

and should, therefore, not be relevant for the product storage. This assumption is permissible, as the subsequent processing step is not part of the system boundary of this scenario. The implementation of the steps defined by the DD-EFMod method of *EFKF determination*

an *energy flexibility model synthesis* for the automated determination of the key figures of the EFDM for this scenario is summarized in Algorithm 8.

7.2.3 Evaluation and Results

The dataset used for this second scenario extends the dataset of the first scenario described in Section 7.1.3 with the scenario-specific information given in Table 7.14. This results in a data profile of 7595 samples with a sampling rate of 1 second recorded from the machine tool PLC. The data quality was assessed and confirmed to be consistent with the requirements of the analysis, exhibiting similar transient behaviors as noted in the first scenario. The scenario-specific storage information includes product storage characteristics, allowing a more comprehensive analysis of energy flexibility under varying storage loads. The application of Algorithm 8 for the energy flexibility modeling of the product storage resulted in the key figures listed in Table 7.15. This results in a maximum possible energy content of 18.97 kW h. Within step energy *flexibility model validation*, the positive validation of these key figures with the combination of the key figures from the first scenario is relevant. In conjunction with the machine tool, this results in a significant increase in flexibility potential. This flexibility potential is shown in Figure 7.8. Compared to the isolated consideration of the EMAG GT

Table 7.15: Automated determined key figures of the EFDM parameterized via the DD-EFMod method for the product storage (own table).

Key Figure	Min	Max	
$C_{\rm S}$	$1.16\mathrm{kW}\mathrm{h}$	$18.97\mathrm{kW}\mathrm{h}$	
$E_{ m to}$	$1.19\mathrm{kW}\mathrm{h}$	$2.37\mathrm{kW}\mathrm{h}$	
$S_{ m S}$	EMAG GT		
E_{Tar}	$1.19\mathrm{kW}\mathrm{h}$	$2.37\mathrm{kW}\mathrm{h}$	
$E_{\rm Loss}$	$0 \frac{\%}{h}$		



Figure 7.8: Resulting flexibility space of EMAG GT and product storage represented quantity as a function of the flexibilizable power, duration and rate of change (own figure).

from scenario 1, the combination with the product storage results in an increase in the possible holding duration of flexibilization from 0.5 h to 8.0 h, which corresponds to an increase of 1500 %.

The results of the implementation and validation of this second scenario and the resulting EFDM, which was created using the software concept, are available as a repository in the TUdata Lib directory [110]. The EFDM was systematically parameterized through the steps in the DD-EFMod method, demonstrating automated key figure determination of the EFDM. This process not only validates the applicability of the DD-EFMod method for the EDFM class storage but also highlights how the inclusion of material or product storage can increase the flexibility potential.



Figure 7.9: System of scenario 3 (own figure).

7.3 Scenario 3 - Cooling Machine and Cold Storage

The third scenario focuses on the supply systems of the use case. For this, I define the system boundary, as shown in Figure 7.9, around the chiller eCHiller45 [99] and the cold storage [100]. As in the previous scenarios, I have recorded the corresponding measured values. These measured data are used to validate the steps of the DD-EFMod method to realize the EFDM parameterization. Applying the method to this scenario checks whether the developed DD-EFMod method can be applied to different machine types to consider the functional requirement on the method number three (see Chapter 4). In comparison to the previous scenarios, a flexible load and a storage system are considered in combination.
Lindner $et al.$ [29] (own table).			
Class	Key Figure	Description	Feasibility Level
	Р	Power states	1
	$T_{\rm Rea}$	Reaction duration	2
е	$T_{\rm V}$	Validity	2
cib] ad	$T_{ m H}$	Holding duration	2
Lo	$ abla P_{ m Act}$	Activation gradient	2
μ. L	$ abla P_{ m Mod}$	Modulation gradient	2
	∇P_{Dea}	Deactivation gradient	2
	$T_{\rm Reg}$	Regeneration duration	2
v	$C_{\rm S}$	Usable capacity	1
ge*	$E_{ m t_0}$	Initial energy content	1
ora	$S_{ m S}$	Suppliers	1
Stc	E_{Tar}	Target energy content	2

Table 7.16: Technical key figures in the EFDM for describing the flexibility space and their feasibility levels of scenario 3 following Lindner *et al.* [29] (own table).

7.3.1 Key Figure Definition

 $E_{\rm Loss}$

First, in the application and implementation layer of the DD-EFMod method, the step of data verification is done by selecting the relevant energy flexibility key figures of the EFDM of this scenario. These are listed in Table 7.16. As with scenarios 1 and 2, the necessary and sufficient feasibility levels are based on the EFDM [29], so

Energy loss

$$\mathbf{K}_{\text{Chiller}} = [\mathbf{P}, T_{\text{Rea}}, T_{\text{V}}, T_{\text{H}}, \nabla P_{\text{Act}}, \nabla P_{\text{Mod}}, \nabla P_{\text{Dea}}, T_{\text{Reg}}] \qquad (7.23)$$

$$\mathbf{K}_{\text{Storage}} = [C_{\text{S}}, E_{\text{t}_0}, S_{\text{S}}, E_{\text{Tar}}, E_{\text{Loss}}, E_{\text{Drain}}]$$
(7.24)

as vectors of all relevant EFKF follows.

To determine these EFKFs, I analyzed the input variables listed in Table 7.17 as part of the *data verification* step. In addition, the boundary conditions and technical limits defined in Table 7.18, as well as the manufacturer specifications, must be taken into account [100]. From

2

(************)*			
Measured Value	Symbol	Unit	Priority
Electric power	P_{el}	W	•
Thermal power	P_{th}	W	\bullet
Operating point	$p_{ m op}$	$^{\circ}\mathrm{C}$	•
Time	t^{-}	s	•
EER	EER	-	\bullet
Operating state	s_{op}	binary	\bullet
Storage temperatur	$\vartheta_{ m stor}$	$^{\circ}\mathrm{C}$	•

Table 7.17: Priority of measured values of the eChiller45 for scenario 3 (own table).

Table 7.18: General boundary conditions for the cold storage (own table).

Term	Symbol	Value
Cooling network set temperature	$\vartheta_{ m set}$	$15^{\circ}\mathrm{C}$
Technical maximum temperatur	$\vartheta_{ m stor}^{cmax}$	$25^{\circ}\mathrm{C}$
Technical minimum temperatur	$\vartheta_{ m stor}^{cmin}$	$12^{\circ}\mathrm{C}$
Maximum flexibility temperature	$\vartheta^{fmax}_{ m stor}$	$20^{\circ}\mathrm{C}$
Minimum flexibility temperature	$\vartheta^{fmin}_{ m stor}$	$15^{\circ}\mathrm{C}$
Volumen of storage	$V_{ m stor}$	$1000\mathrm{L}$
Mass of storage	$m_{ m stor}$	$1000\mathrm{kg}$
Specific heat capacity	$c_{ m p}$	$4.184 \frac{\text{kJ}}{\text{kg K}}$
Heat loss of the storage	$q_{ m BS}$	$\frac{1.37\mathrm{kWh}}{24\mathrm{h}}$

this data, the vector of input variables

$$\mathbf{X}(t) = [P_{el}, P_{th}, p_{op}, t, EER, s_{op}, \vartheta_{stor}]$$
(7.25)

for further processing derives. Using these measured variables and the boundary conditions in Table 7.18, the *data preparation* step and the *energy flexibility modeling* step is used to determine the flexibility space $\mathcal{F} = f(\mathcal{L}, \mathcal{S})$.

7.3.2 Implementation

Based on the available input data $\mathbf{X}(t)$, it is not possible to identify typical production power states for the cooling machine by using clustering algorithms like batch K-means clustering directly (see Figure 7.10). The reason is, that this machine type is a continuous flexible load. So a procedure different from the application of clustering algorithms used for the EMAG GT in scenario one (see Section 7.1) is necessary. As part of the *data preparation* step the electric power consumption of the eChiller45 cooling machine is first quantified into different power ranges, starting from the power range specified by the manufacturer from $P_{el,min} = 0.3 \,\mathrm{kW}$ to $P_{el,max} = 14.5 \,\mathrm{kW}$. Assuming that the building automation and control allows the chiller to be controlled (I refer to the work of my research colleagues Borst *et al.* [113] and our previous work Fuhrländer-Völker et al. [93]), it is possible to determine **P** using Equal-Width Binning (see Section 2.3 in Equation 2.8 to 2.10). Based on this, the key figures of the flexible load eChiller45 can be determined through the implementation of data analytics algorithms, as realized in Algorithm 9. This is part of the *energy flexibility* modeling step with the sub-processes EFKF determination and energy *flexibility model synthesis* and is described in the following.

Taking the cold storage for implementation into account of the EFDM of this scenario this leads to a holding duration determined by

$$T_{\rm H} = \frac{C_{\rm S}}{\mathbf{P}}.\tag{7.26}$$

Algorithm 9 Deployment: EFKF of eChiller45

Input: Raw Data X(t) Output: EFKF eChiller45

- 1: Data Preparation;
- 2: Calculate P (Eq.: 2.8);
- 3: Calculate $T_{\rm H}$ (Eq.: 7.26);
- 4: Identification of gradients (Eq.: 7.6);
- 5: Calculate T_{Rea} (Eq. 7.27);



Figure 7.10: Measured data of power consumption from cooling machine with defined power states (own figure).

The key figure of the power gradients could be determined based on Equation 7.6. Subsequently, the key figure reaction duration of the cooling machine could be determined according to

$$T_{\text{Rea}} = \frac{P_5^{\text{max}} - P_1^{\text{min}}}{\text{mean}(\nabla P)}.$$
(7.27)

It is possible with the mean value of the power gradients as the maximum reaction time. The implementation of the calculations in the context of the software concept are implemented according to Algorithm 9.

The determination of the EFKF for the cold storage [100] is shown in my previous work Lindner *et al.* [30] and briefly summarized below, taking into account the boundary conditions from Table 7.18. The following equations are implemented in Algorithm 10 in the context of

Algorithm 10 Deployment: EFKF of Cold Storage

Input: Raw Data X(t) Output: EFKF cold storage

- 1: Data Preparation;
- 2: Calculate $C_{\rm S}$ (Eq.: 7.28)
- 3: Identification of E_{t_0} and E_{Tar} (Eq.:7.29-7.30)
- 4: Calculate E_{Loss} (Eq.: 7.32)

the software concept:

$$C_{\rm S} = m_{\rm stor} \cdot c_{\rm p} \cdot \frac{\vartheta_{\rm stor}^{cmax} - \vartheta_{\rm stor}^{cmin}}{r_{\rm u}}$$
(7.28)

$$E_{\rm t_0} = m_{\rm stor} \cdot c_{\rm p} \cdot \frac{\vartheta_{\rm stor}^{fmax} - \vartheta_{\rm stor}^{fmin}}{r_{\rm u}}$$
(7.29)

$$E_{\rm Tar} = E_{\rm t_0} \tag{7.30}$$

$$\Delta T = \vartheta_{\text{stor}}^{fmax} - \vartheta_{\text{stor}}^{fmin} \tag{7.31}$$

$$E_{\text{Loss}} = \frac{q_{\text{BS}}}{V \cdot c_p \cdot \rho \cdot \Delta T} \cdot 100 = 0.98 \frac{\%}{\text{h}}$$
(7.32)

Here, $r_{\rm u} = \frac{1}{3600}$ is the conversion coefficient between J and W h. These equations and their implementation in Algorithm 10 represents the DD-EFMod sub-process of EFKF determination. The results and the steps of *energy flexibility synthesis* is discussed below.

7.3.3 Evaluation and Results

The application of the DD-EFMod method in this scenario was successfully implemented. The data profile for this scenario consists of 18,974 samples with a sampling rate of 1 second, resulting in a dataset size of 3.167 MB. It includes data points recorded from the energy monitoring system of the ETA Research Factory given in Table 7.17, capturing information on both the cooling machine and the cold storage. No outliers were detected, but there were missing data points due to network errors that caused incomplete data transmission. These gaps were



Figure 7.11: Measured data of temperature from cold storage (own figure).

addressed through data imputation [104] during the *data preparation* step. The main challenge of this scenario was that no distinguishable power states of the cooling machine eChiller45 could be identified in the period under investigation. Therefore I used the Equal-Width Binning method in the *data preparation* step. This allowed the operating range to be divided into power states. This is applicable due to the continuous controllability of the machine and leads to the operating ranges shown in Figure 7.10. In addition to the power load profile, the figure also shows the Energy Efficiency Ratio (EER) of the machine, as this can be used to derive the dependency of the cooling capacity of the machine.

I have also considered the cold storage in this scenario and use data analytics algorithms for the *energy flexibility modeling* step. By this implementation of the Algorithms 9 and 10 as part of the *energy flexibility key figure determination* step it was possible to realize the automated determination of the key figures. Thereby technical boundary conditions and manufacturer information are required for the cold storage

Key Figure	Min	Max
Р		
P_1	$0.3\mathrm{kW}$	$2.9\mathrm{kW}$
P_2	$3\mathrm{kW}$	$5.9\mathrm{kW}$
P_3	$6 \mathrm{kW}$	$8.9\mathrm{kW}$
P_4	$9\mathrm{kW}$	$11.9\mathrm{kW}$
P_5	$12\mathrm{kW}$	$14.5\mathrm{kW}$
$T_{\rm Rea}$	-	$2892\mathrm{s}$
$T_{\rm V}$	2023-08-31 06:00:00	2023-08-31 22:00:00
$T_{ m H}$		
T_H^1	$18620\mathrm{s}$	$180000\mathrm{s}$
T_H^2	$9152\mathrm{s}$	$18000\mathrm{s}$
T_H^3	$6067\mathrm{s}$	$9000\mathrm{s}$
T_H^4	$4537\mathrm{s}$	$6000\mathrm{s}$
T_H^5	$3600\mathrm{s}$	$4500\mathrm{s}$
$\nabla P_{ m Act}$	-	$0.005 \frac{\text{kW}}{\text{s}}$
∇P_{Mod}	-	$0.005 \frac{kW}{s}$
∇P_{Dea}	-	$0.005 \frac{kW}{s}$
$T_{\rm Reg}$	-	-

Table 7.19: Automated determined key figures of the EFDM parameterized via the DD-EFMod method for the cooling machine eChiller45 (own table).

Table 7.20: Automated determined key figures of the EFDM parameterized via the DD-EFMod method for the cold storage (own table).

Key Figure	Min	Max
$C_{\rm S}$	$0\mathrm{kW}\mathrm{h}$	$15.10\mathrm{kW}\mathrm{h}$
$E_{ m to}$	$0\mathrm{kW}\mathrm{h}$	$5.81 \mathrm{kW}\mathrm{h}$
$S_{ m S}$	eCh	iller45
E_{Tar}	$0\mathrm{kW}\mathrm{h}$	$5.81\mathrm{kW}\mathrm{h}$
$E_{\rm Loss}$	0.9	$08 \frac{\%}{h}$

system to determine the EFKF, in particular, to maximize the flexibility potential. If only the measurement data shown in the observation



Figure 7.12: Resulting flexibility space of eChiller45 and the cold storage (own figure).

period (see Figure 7.11) is used to determine the EFKF, a lower storage capacity would be identified as the entire storage potential was not utilized during the measurement recording.

The flexibility spaces resulting from the energy flexibility model synthesis the EFDM which was built are listed in the Tables 7.19 and 7.20. This leads to the energy flexibility model validation step to check this EFDM. The resulting flexibility space with the flexibilizable power between 0.3 kW to 14.5 kW can be flexibilized by taking into account the cold storage over a period of 5.2 h to 1.25 h with a change in power of $0.005 \frac{\text{kW}}{\text{s}}$. These results are valid since the operating boundary conditions are met. The resulting flexibility space is shown in Figure 7.12 as an example for an average power of 5 kW. Compared to the flexibility space from scenario 2 (Figure 7.8), the low power gradients indicate that this system is very inertial.



Figure 7.13: System of scenario 4 (own figure).

7.4 Scenario 4 - Identification of Dependencies

The fourth scenario focuses on the Energy Flexibility Data Model (EFDM) class dependency, which models the dependencies among flexible loads in the use case. For this purpose, all previous subsystems, including the machine tool EMAG GT, the product storage, the cooling machine eChiller45, and the cold storage, are considered (see Figure 7.13). This scenario explores the potential for automated determination of the energy flexibility key figures (EFKF) for the class dependency, following the steps outlined in the DD-EFMod method. In the *data verification* step, the necessary EFKF, as defined in Table 2.4, are established, and various analyses of the input measurement data are executed. This includes a correlation analysis to assess direct relationships and a lagged correlation analysis to identify possible dependencies with temporal shifts among the subsystems. Following this, the *energy flex-* *ibility modeling* step involves implementing a data analytics algorithm for Boolean analysis (see Chapter 2), which facilitates the recognition of logical links between the data from the flexible loads. The implementation according to the software concept (see Section 5.2) is explained in the following.

7.4.1 Key Figure Definition

The relevant energy flexibility key figures to be determined are based on the conditions of the EFDM according to [29]. Based on this, the key figures for dependencies between the flexible loads are defined in Table 7.21. For this purpose, the corresponding power values and machine states of the flexible loads (see Table 7.2 and Table 7.17) are analyzed as a combined set of data. Consequently, the vector of EFKF is defined



Figure 7.14: Power consumption of the flexible loads EMAG GT and eChiller45 and the cooling temperature for EMAG GT (own figure).

Table 7.21: Technical key figures in the EFDM for describing the flexibility space and their feasibility levels of scenario 4 following Lindner *et al.* [29] (own table).

Class	Key Figure	Description	Feasibility Level
×*	ID_{Tar}	Target flexible load	1
enc	ID_{Tri}	Trigger flexible load	1
nde	Typ_{Temp}	Temporal type	1
ibei	Typ_{Log}	Logical type	1
Ď			

as

$$\mathbf{K} = [ID_{\mathrm{Tar}}, Typ_{\mathrm{Temp}}, ID_{\mathrm{Tri}}, Typ_{\mathrm{Log}}]$$
(7.33)

(7.34)

as the vector of input data for the following *energy flexibility modeling* steps as

$$\mathbf{X}(t) = \left[P_{el}^{EMAG}, P_{el}^{Chiller}, s_{\rm st}, s_{\rm wk}, s_{\rm op}, s_{\rm da}, T_{\rm pro}P_{th}, p_{\rm op}, t, \vartheta_{\rm stor}\right].$$
(7.35)

For illustrative purposes Figure 7.14 displays the measured power values for the flexible load eChiller45 and EMAG GT, together with the cooling temperature of the machine tool. This figure shows the variable power states of the machine tool with only low electrical power consumption of the chiller.

7.4.2 Implementation

To identify dependencies between the machines and their corresponding measured variables in the *data verifaction* step. Therfore I first carried out a correlation analysis (see Chapter 2). Thereby I focused on the electrical power consumption of the two machines. I then extended the correlation analysis to include a lagged correlation to identify possible time-delayed operational process dependencies.

In addition, I performed a Boolean analysis using the binary machine states to identify possible logical dependencies between the power states of the individual machines as part of the *EFKF determination* subprocess. In order to better evaluate the dependency between several flexible loads, I further introduce the parameter $\mathfrak{D} \in [0, 1]$ as the degree of dependency. This describes the degree of dependency of a flexible load in relation to all flexible loads in the system under consideration and is calculated according to

$$\mathfrak{D}_{i} = \frac{\sum_{j=1}^{n} a_{ij}}{n-1}.$$
(7.36)

Let **A** be an $n \times n$ matrix with a_{ij} elements and with $n \in \mathbb{N}$ as the total number of machines in the system, which represents the dependencies between the machines. Here, $a_{ij} = 1$ means that machine *i* is dependent on machine *j* and $a_{ij} = 0$ means that there is no logical dependency. The degree of dependency \mathfrak{D}_i is, therefore, a measure of how strongly machine *i* is dependent on other machines in the system relative to the maximum possible number of dependencies. This parameter can therefore support the selection of potential energy-flexibility measures and corresponding flexible load measures, as the lower the degree of dependency, the easier it is to implement. The determination of \mathfrak{D} is according to DD-EFMod in the step energy flexibility key figure determination consider the developed software concept in Section 5.6.

7.4.3 Evaluation and Results

The application of the DD-EFMod method to determine the EFKF of the EFDM class dependency was successfully applied for the flexible loads EMAG GT and eChiller45 considered in the system of scenario four. The data profile for this scenario consists of 6.989 samples with a sampling rate of 1 second, resulting in a dataset size of 4.0 MB. The data was sourced from the machine tool PLC system for the EMAG GT

Flexible Load	EMAG GT	eChiller45	
EMAG GT	1	0.053	
eChiller45	0.053	1	

 Table 7.22: Correlation matrix of flexible loads of scenario 4 (own table).

and the energy management system of the ETA Research Factory and represent the input data $\mathbf{X}(t)$. No outliers or missing data points were detected during the observed period. The correlation analysis of the input data applied in the step *data verification* shows no dependency due to the low correlation coefficients (see Table 7.22). The extended analysis using the lagged correlation over the time horizon of 1800 s also indicates only very low to no dependency (see Figure 7.15). However, the evaluation of this result in the *EFKF synthesis* sub-process must be assessed critically, as it is known from expert knowledge that the production operation must be cooled during operation and therefore requires the operation of the cooling machine eChiller45. However, it is possible that other machines that were not explicitly considered in the use case, but which are present in the overall infrastructure of the ETA Research Factory, complicate the correlation analysis accordingly.

To determine the dependency key figures in the EFKF determination sub-process, I performed the Boolean analysis (see Chapter 2) based

Algorithm 11 Deployment: Dependency Identification

Input: $\mathbf{X}(t)$

Output: Dependencies between machines

- 1: Load data from $\mathbf{X}(t)$;
- 2: Compute correlation:
- 3: Compute correlation (Eq.: 2.3)
- 4: Compute lagged correlation (Eq.: 2.5);
- 5: Analyze operating states;
- 6: Compute Boolean analysis;
- 7: Derivate logical dependendcies;



Figure 7.15: Lagged correlation of eChiller45 and EMAG GT (own figure).

on the machines and operating states $(s_{st}, s_{wk}, s_{op}, s_{da})$. This analysis, implemented by Algorithm 11 leads to the following results:

- Mutual Exclusion: False
- Concurrent Operation: True
- EMAG GT operation implies eChiller operation: True
- eChiller operation implies EMAG GT operation: False

As a result, when the machine tool is in operation, the chiller is also in operation. However, the chiller can work without the machine tool being in operation.

From this, the dependency matrix

$$\mathbf{A} = \begin{bmatrix} 0 & 1\\ 0 & 0 \end{bmatrix} \tag{7.37}$$

follows. As part of the energy flexibility model synthesis stpes, this results in $\mathfrak{D}_{\mathrm{EMAG}} = 1.0$ and $\mathfrak{D}_{\mathrm{Chiller}} = 0.0$ for the flexible loads. This means that the EFKF for the dependencies of the machines can be

Table 7.23: Automated determined key figures of the EFDM parameterized via the DD-EFMod method for the dependency between two flexible loads (own table).

Key Figure	Value
$ID_{\rm Tri}$	EMAG GT
Typ_{Temp}	total
ID_{Tar}	eChiller
Typ_{Temp}	total
Typ_{Log}	implies

build as listed in Table 7.23 and transferred to the EFDM.

In summary, the combined flexibility space $\mathcal{F} = f(\mathcal{L}, \mathcal{S})$ of scenario 4 can be created taking into account the previous automated parameterized EFDMs of the previous scenarios. The resulting flexibility space is shown in Figure 7.16. In the DD-EFMod sup-process *EFKF synthesis*, I added the degree of dependency \mathfrak{D} as an additional evaluation variable to quantify how strong one flexible loads influences another flexible load. This can support the realization of a flexible load measure, especially in automated information processing. For the final building of the Energy Flexibility Data Model in a standardized form, I used the external software tool EFDM GUI [94] (see Appendix A.1.1) as described in the Software Concept. The automated parameterized Energy Flexibility Data Model is available in the standardized JSON format in the supplementary material collected in TUDataLib [110].



Figure 7.16: Combined flexibility space of eChiller45 and EMAG GT and all storage units of the use case. The combination of the two flexible loads in one flexibility space increases the possible applications as flexible load measures. The degree of dependency between the two flexible loads, which indicates mutual dependency during operation, must be considered when implementing flexible load measures (own figure).

7.5 Evaluation of Flexible Load Measures

This section shows how the automated parameterized Energy Flexibility Data Model as the result of the DD-EFMod method can be used. For this purpose, I use simulated machine operation of the EMAG GT machine tool to show the revenue that an energy-flexible operation can achieve as defined by a flexible load measure. The reference case operation of the EMAG GT is used as a basis for comparison with a production of 30 parts over a period of 2 h. The data profile for the reference load curve is identical to that described in scenario one in Section 7.1.3, consisting of 7,595 samples with a resolution of 1 second, a dataset size of 0.37 MB, and no missing values or outliers. For the simulated load profile, the dataset contains 8,475 samples with the same 1-second resolution and a size of 0.25 MB. The data was synthetically generated to represent the expected power consumption of the EMAG GT under energy-flexible operation. No outliers or missing values were present in the simulated dataset, ensuring high-quality input for the evaluation process. To determine the energy costs, I use the energy prices with different resolutions of auction Day-Ahead 60 min, auction Intraday 15min, Intraday Continuous 60 min, and Intraday Continuous 15 min from 31.08.2023 of the European Power Exchange (EPEX) energy market [114] (see Figure 7.17). The EPEX electricity market features two primary trading mechanisms: auction trading and continuous trading. Auction trading, including Day-Ahead and Intraday Auctions, operates at specific times with prices set by the market clearing principle, facilitating better planning and demand coverage for the next day [112]. In contrast, continuous trading runs from the previous day up to shortly before delivery, with prices determined on a pay-as-bid basis, offering real-time flexibility for adjusting positions and managing short-term imbalances [115]. Both methods are essential for market stability, catering to different aspects of efficiency and planning.

The reference case operation with a total electrical energy consumption of 14.35 kW h results in total costs of $1.73 \in$ in relation to the Day-Ahead



Figure 7.17: Energy prices of different energy markets on 31.08.2023 (own figure).



Figure 7.18: Electrical energy consumption and costs on Day-Ahead market for reference case operation (own figure).

market. The total consumption is made up of the aggregated operation of the EMAG GT machine tool and the eChiller45. The electric power consumption load profile for the period on 31.08.2023 from 08:00-10:00 o'clock is shown in Figure 7.18.

The energy-flexibility measure *change processing sequence* should be applied (see Table 2.1) and is chosen because it's applicability for the machine tool EMAG GT was shown in [85], [116] and in my previous work with a similar use case [30]. The energy flexibility measure

Table 7.24: Flexible load measure for change processing sequence (own
table).

Key Figure	Value
$ID_{ m FLM}$ $ID_{ m Load}$	Change processing sequence EMAG GT
$\mathbf{P}_{ ext{FLM}}$	[(8 kW, 08:00), (8 kW, 09:00),(5 kW, 09:00),(5 kW, 10:00)]



Figure 7.19: Electrical energy consumption and costs on Day-Ahead market for energy-flexibility measure *change processing sequence* (own figure).

change processing sequence was applied following the definition in VDI 5207 (see Table 2.1). This measure is usually defined as rearranging individual processing steps with different energy requirements to smooth out energy consumption peaks and optimize costs. In my specific case, however, there is no classic reorganization of the processing steps. Instead, the adjustment is made through production in a compact production batch without breaks, which reduces the production time for the entire production batch. The batch size, defined as the number of units processed in a production run, remains unchanged at 30 parts in 2 hours, as in the reference case. The selected batch size in combination with continuous processing without interruptions creates a compact load profile. This enables the entire production batch to be completed in one continuous period. This compact production method makes the prerequisite for a subsequent energy-flexibility measure *shift* start of job to periods with lower electricity prices. This following second energy-flexibility measure *shift start of job* decreases energy costs while production targets are still met. By choosing a fixed batch size and carrying out production without breaks, the *change processing se*quence measure is implemented following VDI 5207, but adapted to the specific requirements by carrying out continuous processing in a compact production batch. The flexible load measure can be defined according to Table 7.24 as the load change profile \mathbf{P}_{FLM} as data tuples of electrical power consumption and timestamp. This would lead to a total consumption of 12.98 kW h, resulting in total costs of $1.57 \in$ in relation to the Day-Ahead market. This would correspond to a cost saving of $0.16 \notin$ or 9.2% compared to the reference case operation. The energy demand over time and the resulting costs are shown in Figure 7.19.

In a further step, the energy-flexibility measure change processing sequence is now combined with the energy-flexibility measure shift start of job (see Table 7.25) [30]. The selected energy flexibility measures change processing sequence and shift start of job provide a balanced approach to achieving energy flexibility without disrupting the grind-

Key Figure	Value
$ID_{ m FLM}$ $ID_{ m Load}$ ${f P}_{ m FLM}$	Shift start of job EMAG GT [(8 kW 12:00) (8 kW 13:00) (5 kW 13:00) (5
- FLM	kW, 14:00)]

Table 7.25: Flexible load measure for *shift start of job* (own table).



Figure 7.20: Electrical energy consumption and costs for energyflexibility measure *shift start of job* on Day-Ahead market (own figure).

ing process or compromising product quality. These measures can be applied to the systems with minimal modifications, making them costeffective and quick to deploy [26]. For this purpose, the operation of the machine tool is simulatively shifted to the period from 12:00 to 14:00 o'clock. This results in a total energy consumption of 12.98 kW h. The total costs are reduced to $1.16 \in$ in relation to the Day-Ahead market due to the adjusted lower electricity prices for this period. This corresponds to a cost saving of $0.57 \in$ or 33.0 % compared to the reference case operation. The time horizon of the energy demand and the re-



Figure 7.21: Electrical energy consumption and costs on Intraday market for energy-flexibility measure *shift start of job* (own figure).

sulting costs are shown in Figure 7.20. Since fine-grained power states and corresponding holding periods can be identified with the help of the EFDM and the developed DD-EFMod method, I finally look at the possibility of revenue when procuring electricity on the Intraday market. For this purpose, I transfer the existing combined flexible load measure from Table 7.25 and calculate the resulting costs with regard to the Intraday Continuous market with a resolution of 15 min. As a result, the total energy consumption remains the same at 12.98 kW h compared to the previous flexible load measure. The total costs are further reduced to $0.53 \notin$ due to the higher volatility of electricity prices for this period. This corresponds to a cost saving of $1.20 \notin$ or 69.4%compared to the reference case operation. The energy demand over time and the resulting costs are shown in Figure 7.21. This clearly shows that detailed knowledge of energy flexibilities and their application in the form of a standardized data model leads to considerable cost savings during implementation.

7.6 Interim conclusion

This chapter is dedicated to validate the DD-EFMod method and the associated software concept developed in Chapter 5) applied on the use case presented in Chapter 6.

The first scenario successfully validated the basic functionality of automatically determining the EFKF for the EFDM based on historical machine data. In the first step *data verification*, the input data was verified for relevance, and the determining EFKFs were defined. In the second step of data preparation, the necessary data for further processing was prepared accordingly (e.g. by eliminating outliers). This was followed by the *energy flexibility modeling* step. This involved the deployment of offline clustering algorithm (K-means) and online data stream clustering algorithms (mini-batch K-means, CluStream) in the energy flexibility key figure determination sub-process to identify power states. Based on this successful application, further automated determination of the EFKF for the EFDM class flexible load for the machine tool EMAG GT was possible by the implementation of data analytics algorithms. As part of the DD-EFMod sub-process of *energy* flexibility model synthesis the automatically determined EFKF was parameterized in the EFDM. Further, the deployment of the data streaming algorithms for power state determination in scenario one confirms research hypothesis number two, but the online clustering algorithms only achieved partial success in this scenario.

The second scenario showed the successful application of the DD-EFMod method to the EFDM class storage by considering the product storage of the EMAG GT. Therefore, the *energy flexibility modeling* step was applied with the deployment of storage-specific data analytics algorithms. This successful application for the EFDM class storage shows the modular applicability of the DD-EFMod method. This scenario also demonstrated the high flexibility potential of product and material storages in production systems.

In a third scenario, the transferability of the DD-EFMod method to other machine types or flexible loads was successfully validated, considering a cooling machine and the linked cold storage. The main challenge of this scenario is the continuous operation behavior of the cooling machine, which results in the machine not having defined power states. To address this, within the application of the DD-EFMod steps of *data verification* and *data preparation*, Equal-Width Binning was implemented. This leads further to the successful data-driven automated parameterization of the EFDM for this scenario within the *energy flexibility modeling* sub-processes. The successful implementation of this scenario confirms, therefore, the transferability of the DD-EFMod method.

In the fourth scenario, I validated the method for data-driven determination of the EFDM key figures for the class dependency between flexible loads. For this purpose, the machines machine tool EMAG GT from the first scenario and the cooling machine from the third scenrio was analyzed in combination following the DD-EFMod steps. In order to identify possible dependencies in the *data verification* step, a correlation analysis was implemented. This analysis did not identify any correlations in the available data at this point. Because of the operating behavior of the machine tool, however, simultaneous operation of the cooling machine is required. This dependency could be determined in the next step of *energy flexibility modeling* the *EFKF determination* sub-process by implementing a Boolean analysis algorithm. This made it possible to automatically determine the required key figures for the EFDM class dependency based on the available data.

Finally, I validate the economic potential of the detailed determination and application of the automatically parameterized EFDM. For this purpose, the automatically determined key figures of the scenarios were used to form concrete load profiles for the machines in the form of flexible load measures. Based on this, the first research hypothesis can be confirmed. Furthermore, I was able to show that the application of energy-flexibility measures based on detailed modeling of energy flexibility offers monetary advantages, especially when oriented toward the volatile Intraday energy market. However, this requires factories to be able to operate in an energy-flexible manner. As shown, this can be realized by the application of the DD-EFMod method I developed in this thesis. In the following summarizing section, I evaluate the prototypical implementation of the DD-EFMod method and the results achieved meet the requirements specified in Chapter 4. This is summarized in Table 7.26.

Criteria	Fullfilment
FR-M 1 Applicability	•
FR-M 2 Modularity	•
${\bf FR-M}$ 3 Transferability	•
${\bf FR-M}$ 4 Reproducibility	•
FR-EF 1 Feasibility	•
FR-EF 2 Accuracy	•
FR-EF 3 Maximization	\bullet
FR-EF 4 Explainability	•
FR-EF 5 Safty	\bullet
QR 1 Quality	0
$\mathbf{QR} \ 2$ Robustness	\bullet
${f QR}$ 3 Explainability	•
$\mathbf{QR} \ 4 \ \mathrm{Scalability}$	•
${\bf QR}~{\bf 5}$ Real-time capability	•

Table 7.26: Fulfillment of the requirements for the method and implementation as part of the validation (own table).

FR-M 1 Applicability:

The method was implemented using an open-source software concept and real measured machine data. This ensures that the method is widely accessible and based on real data. The validation through the prototypical implementation in various scenarios with real machine data confirms the successful application of the DD-EFMod method for the data-driven determination of energy flexibility key figures and parameterization of the EFDM.

FR-M 2 Modularity

The modularity of the software concept makes it possible to determine only individual EFKFs if required. This increases the flexibility and adaptability of the method to specific requirements. The successful application and results in the scenarios clearly demonstrate the modularity of the DD-EFMod method.

FR-M 3 Transferability The method was successful tested consider different classes (flexible loads, storage, dependencies) of the EFDM of production infrastructure machine types. This confirms the versatile applicability and adaptability of the DD-EFMod method to different system components. The results fundamentally demonstrate the possibility of transfer to different machine types.

FR-M 4 Reproducibility

Several test runs with the same database using different scenarios underline the reproducibility of the DD-EFMod method. The consistency of the EFKF and other parameters when repeatedly applied to the same data confirms the reliability and repeatability of the method, which is essential for scientific and technical application.

FR-EF 1 Feasibility

The transfer of the automatically determined EFKFs of the EFDM in flexible load measures demonstrates the practical feasibility of the method. Validation by simulated flexible load measures not only shows the potential for energy and cost savings, and also the feasibility of transferring the method to real systems, as has already been demonstrated in my previous work Lindner *et al.* [30]

FR-EF 2 Accuracy

The accuracy of the method's results is based on measured values, which makes the parameterized EFKFs reliable. The validation process of the DD-EFMod method in the sub-process *energy flexibility* model synthesis including the EFKF validation and step energy flexibility model validation, is based on expert knowledge and defined metrics and therefore fulfills the implementation requirements.

FR-EF 3 Maximization

The implementation takes into account all theoretical power states and storage capacities in order to exploit the maximum energy-flexibility potential. The validation shows that the theoretically calculated maximum potential must be integrated into a flexible load measure within the technical operating limits to be technically feasible. Not all maximum operating limits could be utilized during the validation.

FR-EF 4 Explainability:

The methodology ensures the traceability and validity of the parameterized EFDMs, whereby this is achieved through the *algorithm selection* step by selecting explainable machine learning and data analysis algorithms. The machine learning and data analysis algorithms selected for validation in this thesis meet the explainability requirements.

FR-EF 5 Safty:

Technical operating limits are an essential aspect of both the validation of individual EFKFs and the validation of the entire EFDM in order to enable safe, energy-flexible operation. The automated determined EFKFs meet these requirements. In addition, it is necessary to check that the limits are fullfilied by the flexible load measures.

QR 1 Quality

The development of the DD-EFMod method includes the step *model* quality sufficient, which is crucial for the evaluation of the selected machine learning or data analytics algorithm for the automated determination of EFKFs, and ensures the quality of the method. The validation of the selected algorithms in this thesis shows a good model quality but with potential for improvement.

QR 2 Robustness

The selection and testing of efficient methods on different data sets shows that not every algorithm is suitable for every key figure, which requires an individual *algorithm selection* step for each key figure.

QR 3 Explainability

The method enables the selection of explainable algorithms through the *algorithm selection* step, which contributes to the transparency and comprehensibility of the results.

QR 4 Scalability

The applied algorithms in this work show asymptotic behavior and provide similar results even for larger data sets, which confirms the scalability of the selected algorithms.

QR 5 Real-time capability

Real-time requirements were considered in the *algorithms selection* step, with all tested algorithms being able to complete the calculations within the specified time frame of a maximum of 15 minutes.

In summary, the method for Data-Driven Energy Flexibility Modeling (DD-EFMod) and the prototypical implementation based on the software concept fulfills the defined requirements from chapter 4 well. However, there is still room for improvement in some technical details, especially in the evaluation and selection of machine learning or data analytics algorithms. The method represents a simplification in determining the energy flexibility key figures of the Energy Flexibility Data Model. It can support energy or production managers in implementing energy-flexibility measures. Due to the possibility of automated parameterization and processing of an EFDM and its key figures, the method I developed thus offers a good starting point for realizing the automated application of industrial energy flexibility.

8 Conclusion and Outlook

"I think the first and most important thing is that you've got to have a goal and a vision."

Arnold Schwarzenegger

This chapter gives a summary of the main findings and achievements of this thesis. Finally, an outlook on possible further improvements and future research topics are given.

8.1 Summary and Conclusion

The efficient and sustainable operation of industrial production systems is associated with challenges. In addition to maintaining product quality and managing production schedules, adapting production to volatile energy prices and renewable energy sources is becoming more important. To achieve this, it is essential to consider all dependencies between production machines, supply systems, and storages. Therefore, industrial energy flexibility provides a strategic solution by enabling production systems to adjust to volatile energy prices or to renewable energy sources without sacrificing productivity. This can be realized by energy-flexibility measures. This adjustment has the potential to reduce energy costs and furthermore to reduce the CO_2 footprint of a company by leveraging the increased use of renewable energies. Furthermore, trading energy flexibility in future energy markets could generate additional revenues for a company while saving energy costs.

A core aspect of realizing energy flexibility is the modeling of energyflexible systems within a generic data model, such th Energy Flexibility Data Model. Given the complexity of industrial systems and the extensive expert knowledge required, the correct modeling of each relevant system is a difficult task.

This thesis addresses these issues by proposing a data-driven method, to automate the parameterization of energy flexibility models, which is essential to realize energy-flexibility measures, particularly in dynamically changing market conditions. This method, called the Data-Driven Energy Flexibility Modeling (DD-EFMod) significantly streamlines the process of enabling energy-flexible factory operations. The research work and the the development of the DD-EFMod was therefore lead by the following research question: "Is it possible to develop a method for automated parameterization of an energy flexibility model, based on data from a production system, to describe the energy flexibility of the production system?"

To answer this research question, two research hypotheses was derived from these central research question:

- 1. "Data-driven automatically parameterized energy flexibility models of machines in production infrastructure can be used to enable energy-flexible operations."
- 2. "Machine learning and data stream mining algorithms can be used to automate the parameterization of energy flexibility models of machines from the production infrastructure."

To adress the research question and the research hypotheses, this research work was carried out according to the Design Research Methodology. Based on the research scope in Chapter 1, a fundamental understanding of the state of the art in science and technology is established in the Descriptive Study I (Chapter 2 and 3). Therefore this work presents the fundamentals of industrial energy systems and energy flexibility in interaction with digital production. Furthermore, existing approaches to data analysis and modeling of energy flexibility are presented, and the research gap is identified through a systematic literature review and analysis.

This work focuses on the prescriptive study. Based on the identified research gap, the research questions, hypotheses, and requirements for the method for data-driven automated modeling of energy flexibility are formulated in Chapter 4. This is followed by the development of the Data-Driven Energy Flexibility Modeling (DD-EFMod) method in Chapter 5, which provides a structured procedure for the automated modeling of energy flexibility.

The DD-EFMod method, as shown in Figure 5.1, has been developed to facilitate the necessary modeling of energy flexibility through datadriven approaches. This method extends the process to enable factories to energy-flexible operation during the *application and implementation* step. Therefore, the application and implementation layer of the DD-EFMod method specifically uses the different steps data verification, data preparation, and the energy flexibility model setup to describe energy flexibility, reducing the need for manual intervention. In the energy flexibility modeling layer of the DD-EFMod method, the automated determination of the individual energy flexibility key figures of the Energy Flexibility Data Model is realized by the implementation of suitable machine learning and data analytics algorithms in the subprocess energy flexibility key figures determination. These determined key figures are then parameterized, validated, and built to the Energy Flexibility Data Model within the energy flexibility model synthesis sub-process. These structured steps provide the DD-EFMod method to automate and simplify the task of modeling energy flexibility.

In the Descriptive Study II, the DD-EFMod is initially applied and validated on the use case presented in Chapter 6 with a focus on machines which represents electrical flexible loads. The validation was carried out as part of the work, and the resulting conclusions confirm the second research hypothesis. This was proven in particular by the successful implementation of DD-EFMod in the context of the use case. However, it was found that the various implemented machine learning algorithms delivered different results during the deployment. For example, the offline batch K-means clustering algorithm and the online batch incremental mini-batch K-means clustering algorithm produced good and reliable results, particularly when determining the key figure for power states. On the other hand, the data stream mining online learning instance incremental CluStream clustering algorithm for identifying the power states on the basis of the available data does not have the same performance as the K-means batch algorithms, but in principle, it is possible to use these algorithms.

The first research hypothesis can also be confirmed, as the simulative applied energy-flexibility measure *change processing sequence* and *shift start of job*, carried out as defined flexible load measures based on the automatically parameterized Energy Flexibility Data Model shows

good results.

In particular, it could be shown that the detailed modeling of the energy flexibility as EFDM of the machine tool and the cooling machine enables potentially higher cost savings, as a more specific machine operation, oriented to the energy market, is possible. The application of the energy flexibility measure *change processing sequence* as a EFDM based flexible load measure resulted in cost savings of 9.2%. Additional cost savings of up to 69.4% were achieved by combining the energy flexibility measures *change processing sequence* and *shift start of job* in regard to the Intraday energy market as part of the validation of the use case. The research question can be answered positively and confirmed by confirming the research hypotheses. This also shows that the DD-EFMod method developed as part of this work simplifies the enablement process for energy flexibilization of factories.

8.2 Outlook and Further Research

The implementation and the developed Data-Driven Energy Flexibility Modeling method proposed in this thesis form a starting point for filling an essential gap regarding the modeling and application of energy flexibility in manufacturing. While the proposed method demonstrates significant potential, it also presents several limitations that should be considered in its application and future developments:

- The parameterization of Energy Flexibility Data Models and machine learning models must be repeated for each new machine or system, as the models are not directly transferable.
- The approach assumes the availability of high-quality, consistent data as a prerequisite, which may not always be achievable in every production environment.
- Reliable access to machine data and the network infrastructure is critical. Network stability and data transmission issues can affect the quality of the datasets used for modeling and analysis.

- The dataset characteristics, including transient behaviors and specific operating conditions (e.g., in the ETA Research Factory), may limit the direct applicability of the findings to other production environments or machine types without additional adaptation efforts.
- The simulated dataset used for validation was tailored specifically for the machine tool EMAG GT under predefined operating conditions. This may restrict the generalization of results to other configurations or scenarios.

Despite these limitations, the DD-EFMod method provides significant opportunities for further development and enhancement. Addressing these challenges is not only crucial for reinforcing the robustness of the method but also serves as a foundation for exploring new research opportunities and practical applications in energy flexibility. Building on these insights, several potential areas of future work have been identified, focusing on expanding and refining the method's capabilities:

- Expanding the DD-EFMod method to include the parameterization of logistical and economic energy flexibility key figures, broadening the scope of energy flexibility assessment. In particular, the key figures of the Energy Flexibility Data Model are to be considered therefore.
- Developing automatic machine learning pipelines to fully automate the machine learning model selection and the method's application, thereby reducing manual intervention and expert knowledge.
- Integrating the energy flexibility models into asset administration shell submodels (given in [117]) and management systems for real-time decision-making and operational control. This can make a decisive contribution to standardized data exchange and thus enhance the automated use of energy-flexible systems [118, 38].
- Validation of the DD-EFMod method for different use cases and industries is crucial to ensure its robustness and applicability. In the context of this work, the focus was on electrical flexible

loads. Future research should therefore investigate the application to other types of flexibilities, e.g. heat supply, aggregation of multiple flexibilities [90], bidirectionally chargeable electric vehicles for factories [119, 120] or chemical processes, to confirm the method's applicability in a much larger range of industrial energy flexibility scenarios.

• Investigating alternative models, such as those proposed by [77] and [82], or further developing the EFDM to incorporate data exchange with external simulation and planning tools like [121] or [122], offers promising opportunities for expanding the dimensions of energy flexibility in manufacturing. These enhancements could facilitate more sophisticated optimization strategies, enabling real-time energy management and more precise alignment with market signals or operational constraints.

These research topics aim to address the current limitations and extend the potential of energy flexibility in manufacturing, contributing to achieving sustainable and adaptable industrial production.
A Appendix

A.1 Further Related Work

A.1.1 The EFDM GUI Service

The EFDM GUI is a web application tailored for creating energy flexibility data models (EFDM) that facilitate the standardized description of energy flexibility. The service was developed as an element of the Synergy project in the period from 2019 to 2023 and has since been provided by Institute of Production Management, Technology and Machine Tools (PTW) at the Technical University of Darmstadt [94].

This application supports energy managers in manually entering data related to storage systems, flexible loads, their dependencies, and specific energy flexibility measures into the system [5, pp. 255–260]. The DD-EFMod is used to automatically determine the energy flexibility key figures of the Energy Flexibility Data Model. These key figures are then transferred to the EFDM GUI, which facilitates their integration into the required JSON format for further use. The interface of the EFDM GUI, allows the input and management of this data through an HTML-based graphical user interface. Users can manually fill out various forms on the EFDM GUI to compile data, which can then be downloaded as a JSON file for (automated) marketing of energy flexibility [29].

The EFDM GUI is designed to ensure platform-independent, easily readable JSON formatted data that conforms to a uniform schema, enhancing (semi-)automated IT processing between industrial companies and energy markets. It includes multiple tabs for detailed editing of classes such as Flexible Loads, Dependencies, and Energy Storage, each contributing to a comprehensive representation of energy flexibility within an enterprise (see Chapter 2).

This tool also allows the integration and interaction with external energy management and market services, aligning with the Energy Synchronization Platform (ESP) to optimize energy-flexible operations across systems. The EFDM serves as a crucial basis for internal and external communications, enabling users to effectively manage and update energy flexibility data in line with industry standards.

A.1.2 The Energy Synchronization Platform (ESP)

The Energy Synchronization Platform [37] concept describes the architecture of components, interfaces and data models for the automation and standardization of energy flexibility trading as well as the integration of the relevant stakeholders. The Energy Synchronization Platform concept is implemented by the company platform and market platform.

The Energy Synchronization Platform is divided into two sub-platforms, the company platform and the market platform, which are able to exchange data and interact via an interface. The Energy Synchronization Platform thus describes the interaction of several company platforms on a central market platform to carry out transparent energy flexibility trading supported by information technology. A company platform offers the necessary functionalities for the IT connection and control of energy-flexible production processes and infrastructure in a serviceoriented infrastructure. The market platform can be described as a multilateral platform that makes it possible to connect and control various energy flexibility markets and services .

The separation of the Energy Synchronization Platform into two logical platform types is necessary to securely encapsulate their specific domain knowledge, technologies, and methods without impairing the overall system's operation and performance. The Energy Synchronization Platform represents the overall framework for cooperation between the company platform and the market platform. In this global framework, stakeholders, technical interfaces, data flows, and regulations are defined for successful interaction and integration of the actual software platforms, namely the company platform and market platform [38].

The Energy Synchronization Platform provides for several company platforms existing in parallel, which, for example, provide the flexibility of connected energy-flexible systems and processes. Various services are required on the company platform for provision and orchestration. In contrast, a market platform that communicates with all company platforms via a uniform interface acts as an intermediary. By setting up the market platform as a multilateral platform, companies are given access to a large number of existing and future markets, platforms, and supporting services. To enable companies to market flexibility, various services are also provided and executed via the market platform. These support companies in various aspects of flexibility provision, flexibility assessment, and energy flexibility trading [123].





Figure A.1: Mini-batch K-means clustering results EMAG GT (own figure).



Figure A.2: CluStream clustering results EMAG GT (own figure).



Figure A.3: Batch K-means clustering results EMAG GT simulated data set (own figure).



Figure A.4: Mini-batch K-means clustering results EMAG GT simulated data set (own figure).



Figure A.5: CluStream clustering results EMAG GT simulated data set (own figure).

A.3 Detailed Description of the EFDM

The Energy Flexibility Data Model (EFDM) developed within the SynErgie-Project serves for the generic and standardized description and modeling of energy flexibility [29]. It describes energy flexibility in form of a flexibility space and concrete *flexible load measure*. The goal is not to create a completely realistic representation of a flexibility, but the mapping of technically and energetically relevant information in a granularity that enables the communication of flexibility between industrial companies and energy markets on the basis of the Energy Synchronization Platform [37]. It is represented in the form of a JSON schema [29].

Table A.1: Detailed description of the class flexible load of the EFDM (mandatory).

Key figure	At-	Value range	Data type	Unit	Re-	Implicit	
JSON object	tributes				quired	default	
Symbol						value	
Flexible Load ID			String For-		yes		
flexibleLoadId			mat: UUID				
ID _{Load}	The ID of a flexible load for unique identification. The Universally Unique Iden-						
	tifier (UUID) is generated aut	omatically and is	used for	identific	ation and	
	assignment	within an IT syste	em in a company a	nd for fu	rther pro	cessing by	
	external, m	arket-side services					
Reaction Duration		\mathbb{R}_{0+}	Number	s	no	0	
reaction Duration	Time requi	red by a technical	system between t	he call-u	p and th	e start of	
T_{Rea}	a flexible loa	d measure. The ca	all must therefore	be made	at least	with this	
	lead time b	efore the start. T	his key figure is in	nportant	for the co	orrect and	
	timely exec	ution of flexibility	calls and is depen	dent on	the comm	nunication	
	partner. T	hus, the reaction d	uration of a machin	ne itself i	is shorter	than the	
	reaction dura	tion that includes a	communication ch	nain via s	everal up	stream IT	
	systems.						
Validity Period	temporal-	start; end; total	String For-		no	total	
validity	Type		mat: enum				
Tra-	from	\mathcal{H}	String For-		no	constant	
10			mat: Date-			availabil-	
			Time (ISO			ity	
			8601)				
	until	\mathcal{H}	String For-		no	constant	
			mat: Date-			availabil-	
			Time (ISO			ity	
			8601)				
	Subset of the	he company-intern	al planning horizo	n in whic	h the flea	cible load is	
	available. 7	The beginning and	end of this validity	period are	e specified	1 with the	
	attributes f	rom and until. If thi	s period is not spec	ified, con	tinuous a	vailability	
	of the flexib	le load is assumed.	The temporalType	start spec	ifies that	the start	
	of each flex	ible load measure be	longing to this flea	ible load 1	nust lie v	within the	
	validity perio	d. end specifies that	t the end of each	flexible loa	d measure	belonging	
	to this <i>flexi</i>	ole load must lie wi	thin the valiality pe	rioa. 10ta · · · ·	i specine	s that the	
	total durat	ion of each <i>flexible</i>	loaa measure belong	ing to tr	iis <i>fiexible</i>	loaa must	
Barran States	ne within t	me valially perioa.	Number	1-337		0. in f	
Power States	min;max	ĸ	Number	KW	yes	0; inf	
power							
P Duration	minumanı	p+.	Number			holding	
duration / tune	duration Ture	aliveryDuration	Enum	5	100	Duration	
T / T.	auranon Type	holdingDuration	, Lium			Duration	
'H / 'type	L	nonungouration	1				

Order order	order	chronological; arbitrary	Enum		no	arbitrary	
Po		arbitiary					
	A set of power states with which the flexible load can run during each of the						
	(number of changes $+ 1$) holding periods. A positive sign means that the						
	flexible load	causes an increase	in power consump	tion. Neg	ative Por	wer States	
	represent a	decrease in power	consumption. Non-	linear st	ates can l	be approx-	
	imated by a	corresponding pairs	s of values.				
	The time period the flexible load runs in its power states. Duration corresponds						
	with initial power modulation (<i>deliveryDuration</i>). Specifies whether <i>power states</i> can be retrieved in any order OR whether a						
	chronologic	al activation of th	e next power states	must oc	cur when	n the first	
	power state i	s activated. This e	nables the modeling	ng of a de	efined loa	d profile.	
UsaaeNumber	min;max	No	Integer		no	0;inf	
usageNumber					no		
NUSe	The allowed	d number of uses of	of the flexible load v	vithin th	e validity	period. By	
000	specifying	a minimum value,	the necessity of a	calling a	flexible lo	frequency	
	can be desc	ribed.	ximum varue, a mi	intation 0	i the can	nequency	
		No	Integer		no	inf	
Modulation Number	The maxim	um number of pou	ver state changes (n	nodulatio	ns) allow	ved within	
No.	one use of	a flexible load. The	two modulations	correspor	nding to	the initial	
Mod	activation a	and the final deact	ivation are not cou	inted.			
Power Gradient	activa-	\mathbb{R}_{0} +	Number	s kW	no	0;inf	
powerGradients	tionGra-						
	dient						
$\nabla P_{\rm D}$	(min;max)						
V Dea	modula-	R. 1	Number	kW	no	0:inf	
	tionGra-	0+		s		.,	
	dient						
	(min;max)						
	deacti-	\mathbb{R}_{0} +	Number	$\frac{kW}{s}$	no	0;inf	
	vation-						
	Gradi-						
	ent (min-mar)						
	The absolu	te value of the pov	ver gradient of a fl	exible load	indicate:	s how fast	
	the power of	an be increased or	decreased.				
	The activation	on gradient describes	s the possible powe	r gradier	t during	the initial	
	activation of	of a flexible load.					
	The modulat	ion gradient describe	es the power gradie	nt during	g a chang	e of power	
	states.					- 41 - 61	
	deactivation	n period	bes the possible pov	ver gradi	ent durin	g the linal	
Regeneration	dedetrivation	R-1	Number	s	no	0	
Duration	The (minin	num) time for which	ch no (other) meas	sure of t	he same :	flexible load	
regeneration Duration	may be act	ivated after the de	eactivation of a fle	xible load	measure h	nas ended.	
T_{Reg}	This time is	s only to be taken	into account if a de	eactivatio	on is carri	ied out. If	
	another me	asure follows seam	lessly, no deactivat	ion takes	s place be	etween the	
	measures.	This is therefore 1	not a violation of	the prese	cribed reg	generation	
	time.			€h			
Costs florrible Logd Costs	Cost	R	Number	kW	no	0	
CT J	0051						
Load	costPe-	R	Number	€/usag	e no	0	
	rUsage						
	fixedCost	R	Number	€	no	0	
	The costs a	associated with th	e use of the flexible	le load, es	ccluding	electricity	
	costs. The	costs of a flexible loa	d consist of variab	le cost, u	sage cost	and fixed	
	cost: Varia	Die costs refer to	the total amount of the former	or energy	converte	ea. Usage	
	validity perio	d regardless of the	a jeriole loud. Fixe	a costs 1	ncurred (uning the	
	of a power generation plant).						

Order Confirmation	deadline-	absolute; relative	String For-	$\frac{\in h}{kW}$	no	relative
Deadline	Type		mat: enum			
orderConfirmationDeadl	ine					
^t OCD						
	dead-	\mathbb{R}^+_0 (relativ)	Number(rela-	s	no	reaction-
	lineValue	\mathcal{H} (absolut)	tive)			Duration
			String For-			
			mat: Date-			
			Time (ISO			
			8601) (abso-			
			lute)			
	Latest possi	ible time by which	confirmation of an	accepted	l offer of	the flexible
	load must ha	ave been received.	By this time, the	flexibility	y provide	r needs to
	know wheth	er the flexible load n	nust be held ready.	After the	e booking c	on firmation
	deadline is e	xceeded, the flexib	le load offer is rem	oved fron	n the ma	rket. The
	order confirm	ation deadline can b	e specified either a	absolutel	y with a	fixed time
	or relative	to the start time	of the flexible load	l measure	(e.g. 10	0 seconds
	before the start of a <i>flexible load measure</i>). If no value is specified, constant					
	availability for offering is assumed until the latest possible start time within					
	the validity p	period minus the rea	iction duration.			
Prices	vari-	R	Number	$\frac{\in h}{kW}$	no	0
prices	ablePrice					
p						
	pricePe-	R	Number	€/usag	e no	0
	rUsage					
	fixedPrice	R	Number	€	no	0
	Prices to be	realized on the ma	arket as a minimun	n for offe	ring the f	lexible load.
	The prices o	f a <i>flexible load</i> are o	composed of a varia	able price	, a usage	price and
	a fixed pric	e analogous to the	costs. Prices are o	nly neede	d if the	flexible load
	is to be act	ively offered on th	e market.			
Location	meterLoca-		String		no	
location	tion					
L						
	volt-	R+	Number	kV	no	
	ageLevel					
	Meter point	designation for th	ne geographical an	d power ;	grid topo	logical lo-
	cation of a	flexible load. The un	ambiguous allocat	ion is ma	de by me	ans of the
	meter point	designation known	n in the energy ind	ustry / in	n market	communi-
	cation. In C	Germany, it corresp	oonds to a 33-digit	alphanur	neric cod	e number.
	The specific	cation of the grid	voltage level to w	hich the	flexible lo	ad is con-
	nected. Th	e voltage level at	the grid connection	n point i	s to be u	sed, inde-
	pendent of deviating voltage levels in any company grid.					

Table A.2: Detailed description of the class storage of the EFDM (op-						
tional).						
Key figure	At-	Value range	Data type	Unit	Re-	Implicit
JSON object	tributes				quired	default
Symbol						value
Storage ID			String For-		yes	
storageId			mat: UUID			
ID _{Stor}	The ID of	a storage for uniq	ue identification.	The Uni	versally U	nique Iden-
	tifier (UUID)) is generated aut	omatically and is	used for	identific	ation and

external, market-side services.

assignment within an IT system in a company and for further processing by

Usable Capacity	(min;max)	R	Number	kW h	yes		
usableCapacity	Lower and u	apper limit of the s	torage energy cont	ent. Thes	se limits r	nust never	
C_{S}	be exceeded	d or fallen short of	f. Depending on th	ne choice	of refere	nce point,	
5	it is also possible to specify negative storage capacities, e.g. if it possible to						
	exceed or fa	all below storage li	imits.				
Initial Energy	(min;max)	R	Number	kW h	yes		
Content	Energy con	tent of the storage	e at the start time	e of the	validity per	riod of the	
initial Energy Content	associated j	<i>Aexible load</i> (supplier.	s). If the storage is	connect	ed to seve	eral <i>flexible</i>	
E_{to}	loads, the ir	itial energy content	refers to the earlie	est start	time of t	the validity	
-0	periods of al	1 flexible loads. If t	he initial energy c	ontent ca	nnot be	quantified	
	exactly, the	predicted range c	an be specified via	<i>min</i> and	max.	<u> </u>	
Target Energy	(min;max)	R	Number	kW h	no	minimu-	
Content						m Usable	
targetEnergyContent						Capacity	
ETan						maximum	
141						Usable Ca-	
						pacitu	
	Energy con	tent that the stor	age must have at	the end	time of t	the validity	
	period of the	associated flexible	loads (suppliers). If	the stor	age is con	nected to	
	several fleri	ble loads, the target	energy content refe	rs to the	latest er	nd time of	
	the validity periods of all flexible loads. If the target energy content is flexible						
	the desired target range is specified can be specified via min and man						
Energy Loss		[0 100]	Number	%/h	no	0	
energyLoss	Proportion	of the energy con	tent that is conti	nuously	lost e.g	through	
E-	exchange w	ith the environment	nt. The percentage	refers to	the act	ual energy	
Loss	contange with the environment. The percentage refers to the actual energy						
Sumpliana	flomi	not to the storag	String For		1100		
Suppliers	JICLI-		String Por-		yes		
suppliers	oleLoaala	n+	mat: UUID	07		100	
SS	conversion	K '	Number	70	no	100	
	efficiency						
	defined in the same floribility mass. The conversion officionary is a single for						
	defined in the same <i>flexibility space</i> . The conversion efficiency is specified for						
_	each supply	system.	1			I	
Drain	power	R	Number	kW	no	0	
drain	timestamp	\mathcal{H}	String For-		no	total Va-	
E_{Drain}			mat: Date-			lidity Pe-	
			Time			riod	
	A non-influ	enceable energy d	emand in the form	of a load	d profile	that must	
	be met. Th	e interpolation be	tween the power sta	tes is line	ar.		
Costs	variable-	R	Number	€ kWb	no	0	
storageCosts	Cost			K VV II			
cStor	costPe-	R	Number	<u> </u>	no	0	
5101	rUsage			usage			
	fixedCost	R	Number	€	no	0	
	The costs a	ssociated with the	use of the energy	storage	which de	pend only	
	on the (time	e-dependent) epers	v content of the et	orage T	he costs of	f a storage	
	facility con	eist of variable co	et usage cost and	fixed or	net Vari	able coste	
	refer to the	total amount -f	at, usage cost and	ange ac-	a prico -	able costs	
		E E E E E E E E E E E E E E E E E E E	neigy converted. U	sage cost	s arise p	er use or a	
	storage tan	K. FIXED COSTS ARIS	se from keeping a s	torage fa	cuity on	standby.	

Table A.3: Detailed description of the class dependency of the EFDM (optional).

Key figure	At-	Value range	Data type	Unit	Re-	Implicit	
JSON object	tributes	_			quired	default	
Symbol					-	value	
Dependency ID			String For-		ves		
dependency 1D			mat: UUUD		900		
ID-	The ID of	a dependency for	unique identificat	ion Th	o Univers	alla Unique	
Dep	The D of a dependency for unique identification. The <i>Universally Unique</i>						
	<i>Identifier (UUD)</i> is generated automatically and is used for identification and						
	assignment	assignment within an 11 system in a company and for further processing by					
	external, m	arket-side services					
Triggering Flexible	trigger-		String For-		yes		
Load	ingFlexi-		mat: UUID				
triggeringFlexibleLoad	bleLoadId						
ID _{Tri}	temporal-	start; total; end	String For-		yes		
	Type		mat: Enum				
	The ID of t	he flexible load that	triggers the depe	ndency.	The ID m	ust corre-	
	spond to a	flexible load ID in t	he same <i>flexibility</i> :	space. In	addition,	one time	
	parameter ((start, total or end)	of the triggering fle	rible load	affected	by the de-	
	pendency is	to be indicated.	The temporalType				
	start spe	ecifies that with th	ne activation of a	lexible loa	d measure	belonging	
	to this flexil	ble load, the depend	ency applicability du	ration of t	he <i>target</i>	flexible load	
	begins.	, 1	0 11 0				
	end spec	ifies that at the de	activation end of a	flexible loo	d measure	belonging	
	to this flexil	ble load, the depend	ency appicability due	, <i>ration</i> of t	he taraet	flexible load	
	begins	·····			· <i>j</i> ,		
	total and	aifes that during	p florible load mean	wee below	ging to t	hia florible	
	ouu spe	pondonav annliashili	a fierior of the to	waat flowib	le lead in a	unlid	
	toaa, the de	pendency <i>application</i>		пуст псхю	ie ibau is	vanu.	
Target Flexible Load	iargeiriex-		String For-		yes		
	ibieLoaala		mat: UUID				
ID_{Tar}	temporal-	start; total; end	String For-		yes		
	Type		mat: Enum				
	The ID of t	he flexible load that	is affected by the	triggering	flexible loa	d. The ID	
	must corres	pond to a flexible l	oad ID in the same	flexibility	space. In	addition,	
	one time pa	rameter (start, total	or end) of the targe	t flexible lo	ad that a	re affected	
	by the depe	endency is to be in	dicated. The tempe	oralType			
	start spe	cifies that the star	t of a flexible load m	easure of t	he target	flexible load	
	is affected l	by the dependency	applicability duration	ı.			
	end spec	ifies that the deac	tivation end of a fle	$exible \ load$	measure of	f the <i>target</i>	
	flexible load i	is affected by the o	dependency applicat	ility durat	ion.		
	total spe	cifies that the <i>flexib</i>	ble load measure of th	e target fle	xible load	is affected	
	by the depe	endency applicability	duration starting a	t activati	on until	the end of	
	deactivation	n.					
Logical Type		implies;	String For-		yes		
logicalType		excludes	mat: Enum		-		
$Typ_{I or}$	Specifies wl	hether a use of the	triggering flexible lo	ad require	es (implie	es) or pre-	
10g	vents (exclu	ides) the activation	n of the target flexib	le load wi	thin the	applicability	
	duration.	,,					
Applicability Duration	min	R	Number	s	no	Ex-	
applicabilityDuration						cludes:0	
T						Implied	
¹ App						inf	
		D	Number		no	IIII Fa	
	max	ĸ	Number	s	no	Ex-	
						cludes:0	
						Implies:+in	
	The time p	eriod for which th	ie target flexible load	must be	activate	d at least	
	once (impli	es) or may not be	activated at all (excludes)	after the	e triggering	
	flexible load	has been used. Th	he respective time	linkage 1	nust be t	taken into	
	account.	i	1				
Applicability	formu-		String		no	None	
Conditions	laLeft						
applicability Conditions	formula-		String		no	None	
$A_{\rm C}$	Right						

compara- tor	equals; less; lessEqual; greater; greaterEqual	String For- mat: Enum	no	None
Additional of fulfilled. In excludes a c (measure), flexible load target flexible	conditions that mu other words, an a orresponding activ so that the applic metrics are incorp <i>load</i> metrics are ir	ast be met for the c ctivation of the tri vation of the target j cability conditions porated on the left acceptorated on the	lependency to be a ggering flexible load flexible load in a com are fulfilled. That t side of the form right side.	considered implies or figuration e <i>triggering</i> nula. The

Table A.4: Detailed description of the class flexible load measure (FLM) of the EFDM (mandatory).

Key figure	At-	Value range	Data type	Unit	Re-	Implicit
JSON object	tributes				quired	default
Symbol						value
Load Measure ID			String For-		yes	
flexibleLoadMeasureId			mat: UUID			
ID _{FLM}	The ID of th	ne load measure for	unique identificat	ion. The	Universa	lly Unique
	Identifier (UUID) is generate	d automatically an	nd is use	d for ide	ntification
	and assignn	nent within an IT s	system in a compar	iy and for	further	processing
	by external	, market-side servi	ces.			
Flexible Load ID			String For-		yes	
flexibleLoadId			mat: UUID			
ID _{Load}	The ID of t	he Flexible Load t	o which the flexible	load meas	<i>sure</i> is dir	ected.
Load Change Profile	power	R	Number	kW	yes	
loadChangeProfile	timestamp	\mathcal{H}	String For-		yes	
\mathbf{P}_{FLM}			mat: Date-			
			Time (ISO			
			8601)			
	Load profile	e that represents th	e power reduction	or increas	se of the f	lexible load.
	Positive po	wer states mean th	nat the $flexible \ load$	measure r	equests a	n increase
	in power co	nsumption from th	he flexible load. Neg	ative pov	ver states	require a
	decrease in	power consumptio	on. Interpolation b	etween p	ower states	s is linear;
	a step chan	ge in power can b	e mapped by speci	fying two	o equal ti	mestamps
	with differe	nt power values.				
Reward		R	Number		no	0
reward	Total rev	received	by a compa	ny for	execut	ing the
r	flexible load 1	neasure.				

References

- United Nations. Adoption of the Paris Agreement, 2015. https: //unfccc.int/resource/docs/2015/cop21/eng/109r01.pdf (visited on 05/23/2023).
- [2] European Commission. "COMMUNICATION FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT, THE EUROPEAN COUNCIL, THE COUNCIL, THE EUROPEAN ECONOMIC AND SOCIAL COMMITTEE AND THE COM-MITTEE OF THE REGIONS The European Green Deal," European Commission, Brüssel, 2019. https://eur-lex.europa. eu/legal-content/EN/TXT/?uri=CELEX%3A52019DC0640 (visited on 05/23/2023).
- [3] Bundesregierung Deutschland. Federal Climate Change Act of 12 December 2019 (Federal Law Gazette I, p. 2513), as last amended by Article 1 of the Act of 18 August 2021 (Federal Law Gazette I, p. 3905), 2021. https://www.gesetze-iminternet.de/englisch_ksg/englisch_ksg.html (visited on 05/23/2023).
- [4] Nagel, L.; Helgenberger, S.; Okunlola, A. & Sperfeld, F., Soziale und wirtschaftliche Chancen der Energiewende, IASS Fact Sheet, 2019. DOI: 10.2312/iass.2019.015. https://publications. rifs-potsdam.de/pubman/item/item_4185897 (visited on 05/26/2023).
- [5] Sauer, A.; Buhl, H. U.; Mitsos, A. & Weigold, M. Energieflexibilität in der deutschen Industrie - Band 2: Markt- und Stromsystem, Managementsysteme und Technologien energieflexibler

Fabriken. Stuttgart: Fraunhofer Verlag, 2022, vol. 2. https: //synergie-projekt.de/wp-content/uploads/2020/08/ SynErgie_Band_2.pdf (visited on 10/14/2022).

- [6] Lutz, C. & Breitschopf, B. "Systematisierung der gesamtwirtschaftlichen Effekte und Verteilungswirkungen der Energiewende," Fraunhofer Institut für System- und Innovationsforschung ISI, 2016. https://www.isi.fraunhofer.de/ content/dam/isi/dokumente/ccx/2021/Systematisierung_ der _ gesamtwirtschaftlichen _ Effekte . pdf (visited on 05/25/2023).
- [7] AG Energiebilanzen e.V. "Auswertungstabellen zur Energiebilanz Deutschland," AG Energiebilanzen e.V., 2021. https:// ag-energiebilanzen.de/wp-content/uploads/2021/09/ awt_2021_d.pdf (visited on 05/22/2023).
- Bundesnetzagentur für Elektrizität, Gas, Telekommunikation, Post und Eisenbahnen. SMARD / Marktdaten, 2023. https://www.smard.de/home/downloadcenter/downloadmarktdaten/ (visited on 05/25/2023).
- Bundesministerium für Wirtschaft und Klimaschutz (BMWK).
 "Energieeffizienz in Zahlen 2022," 2023. https://www.bmwk.de/ Redaktion/DE/Publikationen/Energie/energieeffizienzin-zahlen-2022.pdf?__blob=publicationFile&v=3 (visited on 05/22/2023).
- [10] Council of the European Union. "Digitalisation for the benefit of the environment: Council approves conclusions." (2020), https: //www.consilium.europa.eu/en/press/press-releases/ 2020/12/17/digitalisation-for-the-benefit-of-theenvironment-council-approves-conclusions/ (visited on 05/26/2023).
- International Energy Agency. "World Energy Outlook 2023," IEA, Paris, 2023. https://www.iea.org/reports/worldenergy-outlook-2023.

- [12] Sauer, A.; Abele, E.; Buhl, H. U.; Fraunhofer IRB-Verlag & Fraunhofer-Institut für Produktionstechnik und Automatisierung. Energieflexibilität in der deutschen Industrie Ergebnisse aus dem Kopernikus-Projekt - Synchronisierte und energieadaptive Produktionstechnik zur flexiblen Ausrichtung von Industrieprozessen auf eine fluktuierende Energieversorgung (SynErgie). 2019.
- Bachmann A.; Bank L.; Bark C.; Bauer D.; Blöchl B.; Brugger M.; Buhl H. U.; Dietz B.; Donnelly J.; Friedl T.; Halbrügge S.; Hauck H.; Heil J.; Hieronymus A.; Hinck T.; Ilieva-König S.; Johnzén C.; Koch C.; Köberlein J.; Köse E.; Lochner S.; Lindner M.; Mayer T.; Mitsos A.; Roth S.; Sauer A.; Scheil C.; Schilp J.; Schimmelpfennig J.; Schulz J.; Schulze J.; Sossenheimer J.; Strobel N.; Tristan A.; Vernim S.; Wagner J.; Wagon F.; Weibelzahl M.; Weigold M.; Weissflog J.; Wenninger S.; Wöhl M.; Zacharias J. & Zäh M. F., *Energieflexibel in die Zukunft Wie Fabriken zum Gelingen der Energiewende beitragen können*, 2021. DOI: 10.24406/FIT-N-638765. http://publica.fraunhofer.de/dokumente/N-638765.html (visited on 11/10/2021).
- Schott, P.; Sedlmeir, J.; Strobel, N.; Weber, T.; Fridgen, G. & Abele, E., A Generic Data Model for Describing Flexibility in Power Markets, Energies, vol. 12, no. 10, p. 1893, May 18, 2019.
 DOI: 10.3390/en12101893. https://www.mdpi.com/1996-1073/12/10/1893 (visited on 02/10/2020).
- Blessing, L. T. & Chakrabarti, A. DRM, a Design Research Methodology. London: Springer, 2009. DOI: 10.1007/978-1-84882-587-1. http://link.springer.com/10.1007/978-1-84882-587-1 (visited on 06/05/2023).
- [16] Groover, M. P. Automation, Production Systems, and Computer-Integrated Manufacturing 4th Edition. Pearson, 2015.
- [17] Westkämper, E.; Spath, D.; Constantinescu, C. & Lentes, J., Eds. *Digitale Produktion*. Berlin, Heidelberg: Springer Berlin

Heidelberg, 2013. DOI: 10.1007/978-3-642-20259-9. https: //link.springer.com/10.1007/978-3-642-20259-9 (visited on 02/20/2024).

- [18] Panten, N. Deep Reinforcement Learning zur Betriebsoptimierung hybrider industrieller Energienetze, Ph.D. dissertation, Technische Universität Darmstadt, Darmstadt, 2019. https: //hds.hebis.de/ulbda/Record/HEB45648695X (visited on 07/17/2023).
- [19] Gholami, H.; Abu, F.; Lee, J. K. Y.; Karganroudi, S. S. & Sharif, S., Sustainable Manufacturing 4.0—Pathways and Practices, Sustainability, vol. 13, no. 24, p. 13956, 24 Jan. 2021. DOI: 10.3390/su132413956. https://www.mdpi.com/2071-1050/13/24/13956 (visited on 02/20/2024).
- Bauer, D.; Kaymakci, C.; Bauernhansl, T. & Sauer, A., Intelligent Energy Systems as Enabler for Increased Resilience of Manufacturing Systems, Procedia CIRP, 54th CIRP CMS 2021
 Towards Digitalized Manufacturing 4.0, vol. 104, pp. 217–222, Jan. 1, 2021. DOI: 10.1016/j.procir.2021.11.037. https://www.sciencedirect.com/science/article/pii/S2212827121009355 (visited on 02/20/2024).
- [21] IRENA. "World Energy Transitions Outlook 2023: 1.5°C Pathway," International Renewable Energy Agency, Abu Dhabi, 2023. https://mc-cd8320d4-36a1-40ac-83cc-3389-cdn-endpoint.azureedge.net/-/media/ Files/IRENA/Agency/Publication/2023/Jun/IRENA_ World_energy_transitions_outlook_2023.pdf?rev= db3ca01ecb4a4ef8accb31d017934e97 (visited on 02/20/2024).
- [22] Isaksen, L. & Ignelzi, P. C. "Demand-side management glossary," Electric Power Research Inst., Palo Alto, CA (United States); Analysis and Control of Energy Systems, Inc., Burlingame, CA (United States); Pacific Consulting Services, Albany, CA (United States), EPRI-TR-101158, Oct. 1,

1992. https://www.osti.gov/biblio/6960128 (visited on 06/19/2023).

- [23] U.S. Department of Energy. "Benefits of demand response in electricity markets and recommendations for achieving them: A Report to the United States Congress Pursuant to Section 1252 of the Energy Policy Act of 2005," Washington DC, 2006. https://www.energy.gov/sites/prod/files/ oeprod/DocumentsandMedia/DOE_Benefits_of_Demand_ Response_in_Electricity_Markets_and_Recommendations_ for_Achieving_Them_Report_to_Congress.pdf (visited on 06/10/2023).
- [24] Verein Deutscher Ingenieure e.V. VDI 5207 Blatt 1 Energieflexible Fabrik Grundlagen, Berlin, 2020.
- [25] DIN EN 62264-1:2014-07, Integration von Unternehmensführungs und Leitsystemen - Teil 1: Modelle und Terminologie (IEC_62264-1:2013); Deutsche Fassung EN_62264-1:2013, 2014. DOI: 10.31030/2156368. https://www.beuth.de/de/-/-/207270059 (visited on 06/13/2023).
- [26] Verein Deutscher Ingenieure e.V. VDI 5207 Blatt 2 Energieflexible Fabrik Identifikation und technische Bewertung, Berlin, 2021.
- [27] Sossenheimer, J. Hybrides Energiemessstellenkonzept zum ganzheitlichen Energiemonitoring von Fertigungsmaschinen und Komponenten, Ph.D. dissertation, Technische Universität Darmstadt, Darmstadt, 2023. https://hds.hebis.de/ulbda/ Record/HEB506134725 (visited on 07/17/2023).
- Buhl, H. U.; Duda, S.; Schott, P.; Weibelzahl, M.; Wenninger, S.; Fridgen, G.; Menci, S. P.; Schöpf, M.; Stiphoudt, C. van; Weigold, M. & Lindner, M. Energieflexibilitätsdatenmodell der Energiesynchronisationsplattform, 2021. DOI: 10.24406/IGCV-N-642370. http://publica.fraunhofer.de/dokumente/N-642370.html (visited on 11/30/2021).

- [29] Lindner, M. & Koch, T. Energy Flexibility Data Model (EFDM), version 1.0.0, Zenodo, 2023. DOI: 10.5281/zenodo.8409627. https://zenodo.org/records/8409626 (visited on 10/30/2023).
- [30] Lindner, M.; Grosch, B.; Elserafi, G.; Dietrich, B. & Weigold, M., Holistic Approach for an Energy-Flexible Operation of a Machine Tool with Cooling Supply, Energies, vol. 16, no. 9, p. 3943, 9 Jan. 2023. DOI: 10.3390/en16093943. https://www.mdpi.com/ 1996-1073/16/9/3943 (visited on 05/08/2023).
- [31] Law Insider. "Day-ahead market gate closure time definition," Law Insider. (2023), https://www.lawinsider.com/ dictionary/day-ahead-market-gate-closure-time (visited on 06/19/2023).
- [32] Bauernhansl, T.; Krüger, J.; Reinhart, G. & Schuh, G., WGP-Standpunkt Industrie 4.0, 2016. https://publica. fraunhofer.de/handle/publica/297876 (visited on 11/20/2022).
- [33] Palttform Industrie 4.0. "Was ist Industrie 4.0?" Was ist Industrie 4.0? (2023), https://www.plattform-i40.de/IP/ Navigation/DE/Industrie40/WasIndustrie40/was-istindustrie-40.html (visited on 05/14/2024).
- [34] DKE Deutsche Kommission Elektrotechnik Elektronik Informationstechnik in DIN und VDE. DIN SPEC 91345:2016-04
 Referenzarchitekturmodell Industrie 4.0 (RAMI4.0), 2016. DOI: https://dx.doi.org/10.31030/2436156.https: //www.plattform-i40.de/IP/Redaktion/DE/Downloads/ Publikation/din-spec-rami40.html (visited on 02/16/2022).
- [35] Körner, M.-F.; Bauer, D.; Keller, R.; Rösch, M.; Schlereth, A.; Simon, P.; Bauernhansl, T.; Fridgen, G. & Reinhart, G., *Extending the Automation Pyramid for Industrial Demand Response*, Procedia CIRP, 52nd CIRP Conference on Manufacturing Systems (CMS), Ljubljana, Slovenia, June 12-14, 2019,

vol. 81, pp. 998-1003, Jan. 1, 2019. DOI: 10.1016/j.procir. 2019.03.241. https://www.sciencedirect.com/science/article/pii/S2212827119305463 (visited on 05/14/2024).

- Bank, L.; Wenninger, S.; Köberlein, J.; Lindner, M.; Kaymakci, C.; Weigold, M.; Sauer, A. & Schilp, J. Integrating Energy Flexibility in Production Planning and Control - An Energy Flexibility Data Model-Based Approach, Hannover : Institutionelles Repositorium der Leibniz Universität Hannover, 2021. DOI: 10. 15488/11249. https://www.repo.uni-hannover.de/handle/ 123456789/11336 (visited on 10/12/2021).
- [37] Fridgen, G.; Menci, S. P.; Stiphoudt, C. van; Schilp, J.; Köberlein, J.; Bauernhansl, T.; Sauer, A.; Grigorjan, A.; Schel, D.; Schlereth, A.; Schulz, F.; Weigold, M.; Lindner, M.; Schimmelpfennig, J. & Winter, C. "Referenzarchitektur der Energiesynchronisationsplattform," Fraunhofer-Gesellschaft, 2021. DOI: 10.24406/IGCV-N-642369. http: //publica.fraunhofer.de/dokumente/N-642369.html (visited on 11/30/2021).
- [38] Ahrens, R.; Bank, L.; Bauer, D.; Bauernhansl, T.; Fridgen, G.; Grigorjan, A.; Kalchschmid, V.; Kaymakci, C.; Köberlein, J.; Lindner, M.; Lodwig, R.; Oeder, A.; Potenciano Menci, S.; Sauer, A.; Schel, D.; Schimmelpfennig, J.; Schlereth, A.; Schöpf, M.; Schulz, F.; Schulz, J.; Stiphoudt, C. van; Weigold, M. & Winter, C. Integration der Flexibilitätsvermarktung (B3), in Energieflexibilität in der deutschen Industrie - Band 2: Marktund Stromsystem, Managementsysteme und Technologien energieflexibler Fabriken, Sauer, A.; Buhl, H. U.; Mitsos, A. & Weigold, M., Eds., vol. 2, Stuttgart: Fraunhofer Verlag, 2022, pp. 237–272. https://synergie_projekt.de/wpcontent/uploads/2020/08/SynErgie_Band_2.pdf (visited on 10/14/2022).
- [39] Köberlein, J.; Bank, L.; Roth, S.; Köse, E.; Kuhlmann, T.; Prell,B.; Stange, M.; Münnich, M.; Flum, D.; Moog, D.; Ihlenfeldt,

S.; Sauer, A.; Weigold, M. & Schilp, J., Simulation Modeling for Energy-Flexible Manufacturing: Pitfalls and How to Avoid Them, Energies, vol. 15, no. 10, p. 3593, 10 Jan. 2022. DOI: 10.3390/en15103593. https://www.mdpi.com/1996-1073/ 15/10/3593 (visited on 02/20/2024).

- [40] Jerry Banks. Handbook of Simulation, 1st ed. John Wiley & Sons, Ltd, 1998. DOI: 10.1002/9780470172445. https:// onlinelibrary.wiley.com/doi/10.1002/9780470172445 (visited on 11/22/2020).
- [41] Banks, J.; Carson II, J. S.; Nelson, B. L. & Nicol, D. M. Discrete-Event System Simulation (Always Learning), Fifth edition, Pearson new international edition. Harlow: Pearson, 2014, 559 pp.
- [42] Hoyle, R. Structural Equation Modeling: Concepts, Issues, and Applications. Thousand Oaks, Calif.: SAGE Publications, Inc, May 3, 1995, 312 pp.
- [43] "Parametrisierung Schreibung, Definition, Bedeutung, Synonyme, Beispiele," DWDS. (Feb. 9, 2023), https://www.dwds.de/wb/Parametrisierung (visited on 11/12/2024).
- [44] Anderson, C. Creating a Data-Driven Organization: Practical Advice from the Trenches, First edition. Beijing Boston Farnham Sebastopol Tokyo: O'Reilly, 2015, 285 pp.
- [45] Hou, Z. & Wang, Z., From model-based control to data-driven control: Survey, classification and perspective, Information Sciences, vol. 235, pp. 3–35, Jun. 1, 2013. DOI: 10.1016/j.ins. 2012.07.014.
- [46] Zeidler, E.; Grosche, G.; Ziegler, V.; Ziegler, D.; Bronštejn, I. N. & Semendjaev, K. A., Eds. Springer-Taschenbuch der Mathematik, 3., neu bearb. und erw. Aufl. Wiesbaden: Springer Spektrum, 2013, 1310 pp.

- [47] Aggarwal, C. C. & Reddy, C. K., Eds. Data Clustering: Algorithms and Applications (Chapman & Hall/CRC Data Mining and Knowledge Discovery Series). Boca Raton: Chapman and Hall/CRC, 2014, 1 p.
- [48] Raschka, S. Python Machine Learning: Unlock Deeper Insights into Machine Learning with This Vital Guide to Cutting-Edge Predictive Analytics (Community Experience Distilled). Birmingham Mumbai: Packt Publishing open source, 2015, 425 pp.
- [49] Han, J.; Kamber, M. & Pei, J. Data Mining: Concepts and Techniques. Elsevier, Jun. 9, 2011, 740 pp. Google Books: pQws07tdpjoC.
- [50] Data Mining: Practical Machine Learning Tools and Techniques. Elsevier, 2011. DOI: 10.1016/C2009-0-19715-5. http:// linkinghub.elsevier.com/retrieve/pii/C20090197155 (visited on 06/22/2018).
- [51] Suthaharan, S. Machine Learning Models and Algorithms for Big Data Classification (Integrated Series in Information Systems). Boston, MA: Springer US, 2016, vol. 36. DOI: 10.1007/978-1-4899-7641-3. http://link.springer.com/10.1007/978-1-4899-7641-3 (visited on 07/17/2018).
- [52] Mertins, A. Signaltheorie: Grundlagen der Signalbeschreibung, Filterbänke, Wavelets, Zeit-Frequenz-Analyse, Parameter- und Signalschätzung ; mit 5 Tabellen (OnlinePLUS), 3., überarb. und erw. Aufl. Wiesbaden: Springer Vieweg, 2013, 447 pp.
- [53] Tietze, U.; Schenk, C. & Gamm, E. Halbleiter-Schaltungstechnik: neuer Teil: nachrichtentechnische Schaltungen, 12. Aufl. Berlin: Springer, 2002, 1606 pp.
- [54] Murphy, K. P. Machine Learning: A Probabilistic Perspective (Adaptive Computation and Machine Learning Series). Cambridge, MA: MIT Press, 2012, 1067 pp.

- Bishop, C. M. Pattern Recognition and Machine Learning. 2006. https://link.springer.com/book/9780387310732 (visited on 12/06/2023).
- Bifet, A.; Gavalà, R.; Holmes, G. & Pfahringer, B. Machine Learning for Data Streams with Practical Examples in MOA. MIT Press, 2018. https://moa.cms.waikato.ac.nz/book/.
- [57] Read, J.; Bifet, A.; Pfahringer, B. & Holmes, G. Batch-Incremental vs. Instance-Incremental Learning in Dynamic and Evolving Data, 2012.
- [58] Beringer, J. Online-Data-Mining auf Datenströmen: Methoden zur Clusteranalyse und Klassifikation, Ph.D. dissertation, Ottovon-Guericke-Universität Magdeburg, Magdeburg, 2007. https: //d-nb.info/987149202/34.
- [59] Sculley, D. Web-scale k-means clustering, in Proceedings of the 19th International Conference on World Wide Web, Raleigh North Carolina USA: ACM, Apr. 26, 2010, pp. 1177-1178. DOI: 10.1145/1772690.1772862. https://dl.acm.org/doi/10. 1145/1772690.1772862 (visited on 01/30/2024).
- [60] Bifet, A.; Holmes, G.; Kirkby, R. & Pfahringer, B. "Data stream mining: A practical approach," University of Waikato, Waikato, 2009. https://www.cs.waikato.ac.nz/~abifet/MOA/ StreamMining.pdf (visited on 06/22/2018).
- [61] Sayed-Mouchaweh, M. Learning from Data Streams in Dynamic Environments. Springer Cham, 2016. https: //link.springer.com/book/10.1007/978-3-319-25667-2 (visited on 04/23/2018).
- [62] Chan, T. F.; Golub, G. H. & LeVeque, R. J. Updating Formulae and a Pairwise Algorithm for Computing Sample Variances, in COMPSTAT 1982 5th Symposium Held at Toulouse 1982, Caussinus, H.; Ettinger, P. & Tomassone, R., Eds., Heidelberg: Physica-Verlag HD, 1982, pp. 30–41. DOI: 10.1007/978-3-642-

51461-6_3. http://link.springer.com/10.1007/978-3-642-51461-6_3 (visited on 01/31/2024).

- [63] Rousseeuw, P. J., Silhouettes: A graphical aid to the interpretation and validation of cluster analysis, Journal of Computational and Applied Mathematics, vol. 20, pp. 53-65, Nov. 1, 1987. DOI: 10.1016/0377-0427(87)90125-7. https://www.sciencedirect.com/science/article/pii/0377042787901257 (visited on 12/01/2023).
- [64] Halkidi, M.; Batistakis, Y. & Vazirgiannis, M., On Clustering Validation Techniques, Journal of Intelligent Information Systems, vol. 17, no. 2, pp. 107–145, Dec. 1, 2001. DOI: 10.1023/A:1012801612483. https://doi.org/10.1023/A: 1012801612483 (visited on 12/01/2023).
- [65] scikit-learn developers. "2.3. Clustering," scikit-learn. (2023), https://scikit-learn/stable/modules/clustering.html (visited on 12/01/2023).
- [66] Wirth, R. & Hipp, J., CRISP-DM: Towards a Standard Process Model for Data Mining, 2000. http://www.cs.unibo.it/ ~montesi/CBD/Beatriz/10.1.1.198.5133.pdf (visited on 09/05/2023).
- [67] Studer, S.; Bui, T. B.; Drescher, C.; Hanuschkin, A.; Winkler, L.; Peters, S. & Müller, K.-R., Towards CRISP-ML(Q): A Machine Learning Process Model with Quality Assurance Methodology, Machine Learning and Knowledge Extraction, vol. 3, no. 2, pp. 392–413, 2 Jun. 2021. DOI: 10.3390/make3020020. https://www.mdpi.com/2504-4990/3/2/20 (visited on 06/20/2023).
- [68] Page, M. J.; McKenzie, J. E.; Bossuyt, P. M.; Boutron, I.; Hoffmann, T. C.; Mulrow, C. D.; Shamseer, L.; Tetzlaff, J. M.; Akl, E. A.; Brennan, S. E.; Chou, R.; Glanville, J.; Grimshaw, J. M.; Hróbjartsson, A.; Lalu, M. M.; Li, T.; Loder, E. W.; Mayo-Wilson, E.; McDonald, S.; McGuinness, L. A.; Stewart, L. A.; Thomas, J.; Tricco, A. C.; Welch, V. A.; Whiting, P. & Moher,

D., The PRISMA 2020 statement: An updated guideline for reporting systematic reviews, BMJ, vol. 372, n71, Mar. 29, 2021. DOI: 10.1136/bmj.n71. pmid: 33782057. https://www.bmj.com/content/372/bmj.n71 (visited on 07/14/2023).

- [69] Tawfik, G. M.; Dila, K. A. S.; Mohamed, M. Y. F.; Tam, D. N. H.; Kien, N. D.; Ahmed, A. M. & Huy, N. T., A step by step guide for conducting a systematic review and meta-analysis with simulation data, Tropical Medicine and Health, vol. 47, no. 1, p. 46, Aug. 1, 2019. DOI: 10.1186/s41182-019-0165-6. https://doi.org/10.1186/s41182-019-0165-6 (visited on 01/02/2023).
- [70] Glock, C. H. & Hochrein, S., Purchasing Organization and Design: A Literature Review, Business Research, vol. 4, no. 2, pp. 149–191, Dec. 1, 2011. DOI: 10.1007/BF03342754. https://doi.org/10.1007/BF03342754 (visited on 01/02/2023).
- [71] Walther, J. Hierarchical Electrical Load Forecasting of Industrial Production Systems in the Manufacturing Industry based on Deep Learning, Ph.D. dissertation, Technische Universität Darmstadt, Darmstadt, 2022. https://tuprints.ulb.tudarmstadt.de/21767/ (visited on 07/17/2023).
- Strobel, N. Einsatz inhärenter Energiespeicher in Produktionssystemen zum elektrischen Lastmanagement, Ph.D. dissertation, Technische Universität, Darmstadt, 2021. DOI: 10.26083/ tuprints-00017581. https://tuprints.ulb.tu-darmstadt. de/17581/ (visited on 07/17/2023).
- [73] Hersi, M.; Traversy, G.; Thombs, B. D.; Beck, A.; Skidmore, B.; Groulx, S.; Lang, E.; Reynolds, D. L.; Wilson, B.; Bernstein, S. L.; Selby, P.; Johnson-Obaseki, S.; Manuel, D.; Pakhale, S.; Presseau, J.; Courage, S.; Hutton, B.; Shea, B. J.; Welch, V.; Morrow, M.; Little, J. & Stevens, A., Effectiveness of stop smoking interventions among adults: Protocol for an overview of systematic reviews and an updated systematic review, Systematic

Reviews, vol. 8, no. 1, p. 28, Jan. 19, 2019. DOI: 10.1186/ s13643-018-0928-x.

- [74] Akhavan-Hejazi, H. & Mohsenian-Rad, H., Power systems big data analytics: An assessment of paradigm shift barriers and prospects, Energy Reports, vol. 4, pp. 91-100, 2018. DOI: 10. 1016/j.egyr.2017.11.002. https://www.sciencedirect. com/science/article/pii/S2352484717300616.
- [75] Li, J.; Chen, Z.; Cheng, L. & Liu, X., Energy data generation with wasserstein deep convolutional generative adversarial networks, Energy, vol. 257, p. 124694, 2022. DOI: 10.1016/j. energy.2022.124694. https://www.sciencedirect.com/ science/article/pii/S0360544222015973.
- [76] Calikus, E.; Nowaczyk, S.; Sant'Anna, A.; Gadd, H. & Werner, S., A data-driven approach for discovering heat load patterns in district heating, Applied Energy, vol. 252, p. 113 409, 2019. DOI: 10.1016/j.apenergy.2019.113409. https://www.sciencedirect.com/science/article/pii/ S0306261919310839.
- [77] Corsetti, E.; Riaz, S.; Riello, M. & Mancarella, P., Modelling and deploying multi-energy flexibility: The energy lattice framework, Advances in Applied Energy, vol. 2, p. 100030, May 26, 2021. DOI: 10.1016 / j.adapen.2021.100030. https://www.sciencedirect.com/science/article/pii/ S2666792421000238 (visited on 03/04/2022).
- [78] Tsay, C.; Cao, Y.; Wang, Y.; Flores-Cerrillo, J. & Baldea, M., Identification and online updating of dynamic models for demand response of an industrial air separation unit, IFAC PA-PERSONLINE, vol. 54, no. 3, pp. 140–145, 2021. DOI: 10.1016/ j.ifacol.2021.08.232.
- [79] Ulbig, A. & Andersson, G., Analyzing operational flexibility of electric power systems, International Journal of Electrical Power & Energy Systems, The Special Issue for 18th Power

Systems Computation Conference. Vol. 72, pp. 155-164, Nov. 1, 2015. DOI: 10.1016/j.ijepes.2015.02.028. https://www.sciencedirect.com/science/article/pii/ S0142061515001118 (visited on 07/28/2023).

- [80] Zhao, Q.; Li, H.; Wang, X.; Pu, T. & Wang, J., Analysis of users' electricity consumption behavior based on ensemble clustering, Global Energy Interconnection, vol. 2, no. 6, pp. 479-488, 2019. DOI: 10.1016/j.gloei.2020.01.001. https://www.sciencedirect.com/science/article/pii/ S2096511720300013.
- [81] DIN Deutsches Institut für Normung e.V. DIN SPEC 91366:2018-04, Referenzmodell zur Charakterisierung der Energieflexibilität von Industrieunternehmen, Berlin, 2018. DOI: 10.31030/2831910. https://www.beuth.de/de/-/-/286549869 (visited on 04/12/2022).
- [82] DIN Deutsches Institut für Normung e.V. DIN SPEC 91410-2 Energieflexibilität - Teil 2: Identifizierung und Bewertung von Flexibilität in Gebäuden und Quartieren, Berlin, 2021. https: //aproxy.ulb.tu-darmstadt.de:2053/8C4/search/itemdetail/DE30089884 (visited on 07/24/2023).
- [83] Lee, E.; Baek, K. & Kim, J., Evaluation of demand response potential flexibility in the industry based on a data-driven approach, ENERGIES, vol. 13, no. 6355, 23 Dec. 2020. DOI: 10. 3390/en13236355.
- [84] Sadeghianpourhamami, N.; Demeester, T.; Benoit, D. F.; Strobbe, M. & Develder, C., Modeling and analysis of residential flexibility: Timing of white good usage, AP-PLIED ENERGY, vol. 179, pp. 790–805, Oct. 1, 2016. DOI: 10.1016/j.apenergy.2016.07.012.
- [85] Schraml, P. Methode zur reduktion maximaler elektrischer lasten spanender werkzeugmaschinen, Ph.D. dissertation, Technische Universität Darmstadt, Aachen, 2018. http:

//scans.hebis.de/HEBCGI/show.pl?42782066_toc.pdf
(visited on 10/06/2020).

- [86] Schulze, C.; Plank, M.; Linzbach, J.; Herrmann, C. & Thiede, S., Energy flexible management of industrial technical building services: A synergetic data-driven and simulation approach for cooling towers, Procedia CIRP, vol. 81, pp. 775–780, 2019. DOI: 10. 1016/j.procir.2019.03.193. https://www.sciencedirect. com/science/article/pii/S2212827119304986.
- [87] Bahmani, R.; Stiphoudt, C. van; Menci, S. P.; Schöpf, M. & Fridgen, G., Optimal industrial flexibility scheduling based on generic data format, Energy Informatics, vol. 5, no. 1, p. 26, Sep. 7, 2022. DOI: 10.1186/s42162-022-00198-4.
- [88] Unterberger, E.; Hofmann, U.; Min, S.; Glasschroeder, J. & Reinhart, G. Modeling of an energy-flexible production control with SysML, in 51ST CIRP CONFERENCE ON MANU-FACTURING SYSTEMS, Wang, L., Ed., ser. Procedia CIRP, vol. 72, KTH Royal Inst Technol; Int Acad Prod Engn, 2018, pp. 432–437. DOI: 10.1016/j.procir.2018.03.111.
- [89] Degefa, M. Z.; Sperstad, I. B. & Sæle, H., Comprehensive classifications and characterizations of power system flexibility resources, Electric Power Systems Research, vol. 194, p. 107022, May 1, 2021. DOI: 10.1016/j.epsr.2021.107022. https://www.sciencedirect.com/science/article/pii/S037877962100002X (visited on 02/28/2023).
- [90] Lindner, M.; Wenninger, S.; Fridgen, G. & Weigold, M. Aggregating Energy Flexibility for Demand-Side Management in Manufacturing Companies – A Two-Step Method, in Production at the Leading Edge of Technology, Behrens, B.-A.; Brosius, A.; Drossel, W.-G.; Hintze, W.; Ihlenfeldt, S. & Nyhuis, P., Eds., ser. Lecture Notes in Production Engineering, Cham: Springer International Publishing, 2021, pp. 631–638. DOI: 10.1007/978-3-030-78424-9_69.

- [91] ISO/IEC 25019. "Systems and software engineering Systems and software Quality Requirements and Evaluation (SQuaRE)
 Quality-in-use model," ISO. (2023), https://www.iso.org/standard/78177.html (visited on 01/08/2024).
- [92] Lodwig, R. & Schulz, F., Energieflexible produktion durch Anlagenkenntnis, Werkstattstechnik online, vol. 111, no. 07-08, pp. 565–569, 2021. DOI: 10.37544/1436-4980-2021-07-08-97.
- [93] Fuhrländer-Völker, D.; Lindner, M. & Weigold, M., Design Method for Building Automation Control Programs to Enable the Energetic Optimization of Industrial Supply Systems, Procedia CIRP, 54th CIRP CMS 2021 Towards Digitalized Manufacturing 4.0, vol. 104, pp. 229-234, 2021. DOI: 10.1016/j.procir.2021.11.039.https://www.sciencedirect.com/science/article/pii/S2212827121009379 (visited on 11/29/2021).
- [94] Koch, T. & Lindner, M. "Service: EFDM-GUI," SynErgie. (2023), https://synergie-projekt.de/ergebnis/serviceefdm-gui (visited on 01/09/2024).
- [95] Grosch, B.; Ranzau, H.; Dietrich, B.; Kohne, T.; Fuhrländer-Völker, D.; Sossenheimer, J.; Lindner, M. & Weigold, M., A framework for researching energy optimization of factory operations, Energy Informatics, vol. 5, no. 1, p. 29, Sep. 7, 2022. DOI: 10.1186/s42162-022-00207-6. https://energyinformatics.springeropen.com/articles/10.1186/s42162-022-00207-6 (visited on 09/14/2022).
- [96] Abele, E.; Beck, M.; Flum, D.; Schraml, P.; Panten, N.; Junge, F.; Bauerdick, C.; Helfert, M.; Sielaff, T.; Daume, C.; Weber, M.; Strobel, N.; Redelberger, A.; Pototzky, L.; Rothenbücher, S.; Landgraf, G.; Rummel, W.; Heimbach, K.; Haase, P.; Stock, S.; Schwarz, J.; Schaal, S.; Kunde, R.; Krönauer, A.; Technische Universität Darmstadt, I. f. P.; AG, B. R.; EMAG-Maschinenfabrik, S.; GmbH, I. S.; KG, M.-E. S. G.

bibinitperiod C. & Energieforschung, B. Z. f. A. "Gemeinsamer Schlussbericht zum Projekt ETA-Fabrik : Energieeffiziente Fabrik für interdisziplinäre Technologie- und Anwendungsforschung," Darmstadt, 2019. DOI: 10.2314/KXP:1667471384.

- [97] Efficient Energy GmbH. Water echiller, 2023. https: //efficient-energy.de/wp-content/uploads/2018/04/EE_ Broschuere_ENG_20180411.pdf (visited on 11/21/2023).
- [98] EMAG GmbH & Co. KG. "CNC-Drehmaschine VLC 100 mit verschiedenen Technologien." (Nov. 21, 2023), https: //www.emag.com/de/produkte-services/maschinen/ drehmaschinen/customized-vlc/vlc-100/ (visited on 11/21/2023).
- [99] Efficient Energy GmbH. Betriebsanleitung Datenblatt eChiller35/45, 2021. DOI: 10.48328/tudatalib-1363.
- [100] Viessmann Climate Solutions SE. "Vitocell 100-E/-W Heizwasser-Pufferspeicher." (Jun. 26, 2023), https : / / www.viessmann.de/de/produkte/warmwasserbereiter/ vitocell-100-e.html (visited on 11/24/2023).
- Bauer, D.; Benedikt, F.; Bianchini, I.; Borst, F.; Silva, C. da; Dietrich, B.; Emde, A.; Fuhrländer-Völker, D.; Hofmann, P.; Ilieva-König, S.; Kaymakci, C.; Lindner, M.; Moog, D.; Retzlaff, J.; Riethmüller, T.; Sauer, A.; Schillinger, C.; Seyed Sadjjadi, B.; Sossenheimer, J.; Stiphoudt, C. van; Weber, T.; Weigold, M. & Wenninger, S. Werkzeuge für die energetische Flexibilisierung (B1), in Energieflexibilität in der deutschen Industrie - Band 2: Markt- und Stromsystem, Managementsysteme und Technologien energieflexibler Fabriken, Sauer, A.; Buhl, H. U.; Mitsos, A. & Weigold, M., Eds., vol. 2, Stuttgart: Fraunhofer Verlag, 2022, pp. 163–208. https://synergie-projekt.de/wpcontent/uploads/2020/08/SynErgie_Band_2.pdf (visited on 10/14/2022).

- [102] Fuhrländer-Völker, D. Automation Architecture for Demand Response on Aqueous Parts Cleaning Machines, Ph.D. dissertation, Technische Universität Darmstadt, Darmstadt, 2023. DOI: 10.26083/tuprints-00024259.
- [103] Python Software Foundation. Python Release Python 3.10.11, version 3.10.11, 2023. https://www.python.org/downloads/ release/python-31011/ (visited on 11/30/2023).
- [104] Pedregosa, F.; Varoquaux, G.; Gramfort, A.; Michel, V.; Thirion, B.; Grisel, O.; Blondel, M.; Prettenhofer, P.; Weiss, R.; Dubourg, V.; Vanderplas, J.; Passos, A.; Cournapeau, D.; Brucher, M.; Perrot, M. & Duchesnay, E., *Scikit-learn: Machine learning in Python*, Journal of Machine Learning Research, vol. 12, pp. 2825–2830, 2011. https: //jmlr.csail.mit.edu/papers/v12/pedregosa11a.html.
- [105] Virtanen, P.; Gommers, R.; Oliphant, T. E.; Haberland, M.; Reddy, T.; Cournapeau, D.; Burovski, E.; Peterson, P.; Weckesser, W.; Bright, J.; van der Walt, S. J.; Brett, M.; Wilson, J.; Millman, K. J.; Mayorov, N.; Nelson, A. R. J.; Jones, E.; Kern, R.; Larson, E.; Carey, C. J.; Polat, İ.; Feng, Y.; Moore, E. W.; VanderPlas, J.; Laxalde, D.; Perktold, J.; Cimrman, R.; Henriksen, I.; Quintero, E. A.; Harris, C. R.; Archibald, A. M.; Ribeiro, A. H.; Pedregosa, F.; van Mulbregt, P. & SciPy 1.0 Contributors, *SciPy 1.0: Fundamental algorithms for scientific computing in python*, Nature Methods, vol. 17, pp. 261–272, 2020. DOI: 10.1038/s41592-019-0686-2. https://scipy.org/.
- [106] Harris, C. R.; Millman, K. J.; Van Der Walt, S. J.; Gommers, R.; Virtanen, P.; Cournapeau, D.; Wieser, E.; Taylor, J.; Berg, S.; Smith, N. J.; Kern, R.; Picus, M.; Hoyer, S.; Van Kerkwijk, M. H.; Brett, M.; Haldane, A.; Del Río, J. F.; Wiebe, M.; Peterson, P.; Gérard-Marchant, P.; Sheppard, K.; Reddy, T.; Weckesser, W.; Abbasi, H.; Gohlke, C. & Oliphant, T. E., Array programming with NumPy, Nature, vol. 585, no. 7825,

pp. 357-362, Sep. 17, 2020. DOI: 10.1038/s41586-020-2649-2. https://www.nature.com/articles/s41586-020-2649-2 (visited on 11/30/2023).

- [107] Wes McKinney. Data Structures for Statistical Computing in Python, in Proceedings of the 9th Python in Science Conference, van der Walt, S. & Jarrod Millman, Eds., 2010, pp. 56–61. DOI: 10.25080/Majora-92bf1922-00a.
- The pandas development team. Pandas-dev/pandas: Pandas, version 2.0.3, Zenodo, Feb. 2020. DOI: 10.5281/zenodo.
 3509134. https://doi.org/10.5281/zenodo.3509134.
- [109] Montiel, J.; Halford, M.; Mastelini, S. M.; Bolmier, G.; Sourty, R.; Vaysse, R.; Zouitine, A.; Gomes, H. M.; Read, J.; Abdessalem, T., et al., River: Machine learning for streaming data in Python, 2021.
- [110] Lindner, M. Supplementary Material: Method for Data-Driven Automated Parameterization of Energy Flexibility Models of Production Systems, Software, Apr. 29, 2024. DOI: 10.48328 / tudatalib - 1426. https://tudatalib.ulb. tu-darmstadt.de/handle/tudatalib/4222 (visited on 05/11/2024).
- [111] scikit-learn developers. "Sklearn.cluster.KMeans," scikit-learn. (2023), https://scikit-learn/stable/modules/generated/ sklearn.cluster.KMeans.html (visited on 12/01/2023).
- [112] Next Kraftwerke GmbH. "Was ist der Intraday-Handel." (2024), https://www.next-kraftwerke.de/wissen/intradayhandel (visited on 01/30/2024).
- [113] Borst, F.; Frank, M. G.; Theisinger, L. & Weigold, M. ThermalSystemsControlLibrary: A Modelica Library for Developing Control Strategies of Industrial Energy Systems, presented at the 15th International Modelica Conference 2023, Aachen, October 9-11, Aachen, Oct. 11, 2023, pp. 209–215. DOI: 10.3384/ecp204.

https://ecp.ep.liu.se/index.php/modelica/issue/view/ 83/86 (visited on 02/07/2023).

- [114] EPEX SPOT. "Market Data Services | EPEX SPOT." (2023), https://www.epexspot.com/en/marketdataservices (visited on 12/11/2023).
- [115] PVNED. "Intraday trading EPEX," PVNED. (2024), https: //www.pvned.eu/en/intraday-trading-epex/ (visited on 05/28/2024).
- [116] Grosch, B. E. Simplified Implementation of Energy-Aware Production Scheduling in Job Shops, Ph.D. dissertation, Technische Universität Darmstadt, Darmstadt, Dec. 20, 2023. DOI: 10. 26083 / tuprints - 00026459. https://tuprints.ulb.tudarmstadt.de/26459/ (visited on 01/14/2024).
- [117] Plattform Industrie 4.0. Details of the Asset Administration Shell, Federal Ministry for Economic Affairs and Climate Action (BMWK), 2024. https://www.plattform-i40.de/IP/ Redaktion/DE/Downloads/Publikation/Details_of_the_ Asset_Administration_Shell_Part1_V3.pdf?__blob= publicationFile&v=1 (visited on 06/29/2024).
- [118] Lindner, M.; Bank, L.; Schilp, J. & Weigold, M., Digital Twins in Manufacturing: A RAMI 4.0 Compliant Concept, Sci, vol. 5, no. 4, p. 40, 4 Dec. 2023. DOI: 10.3390/sci5040040. https: //www.mdpi.com/2413-4155/5/4/40 (visited on 10/10/2023).
- [119] Roth, S.; Spitzer, S.; Braunreuther, S. & Reinhart, G. Modeling and simulation of electric vehicles as battery storage in an energy flexible factory, presented at the 11. Internationale Energiewirtschaftstagung IEWT, Wien, 2019. https: //iewt2019.eeg.tuwien.ac.at/download/contribution/ fullpaper / 128 / 128 _ fullpaper _ 20190303 _ 150154 . pdf (visited on 06/29/2024).

- Kern, T. & Kigle, S., Modeling and evaluating bidirectionally chargeable electric vehicles in the future European energy system, Energy Reports, vol. 8, pp. 694–708, 2022. DOI: 10.1016/j. egyr.2022.10.277.
- [121] DIgSILENT GmbH. "Power System Solutions DIgSILENT." (2024), https://www.digsilent.de/de/home.html (visited on 06/02/2024).
- [122] KISTERS AG. "KISTERS FlexManager steuert Flexibilitäten auf Basis von Smart-Meter-Daten," KISTERS. (2024), https://www.kisters.eu/de/success-story/kistersflexmanager-steuert-flexibilitaeten-auf-basis-vonsmart-meter-daten/ (visited on 06/02/2024).
- [123] SynErgie. "Konzept der Energiesynchronisationsplattform," Konzept der Energiesynchronisationsplattform. (2024), https:
 / / synergie - projekt . de / ergebnis / konzept - der energiesynchronisationsplattform (visited on 06/01/2024).

Own Publications

The following publications have been published while working on this thesis:

2023

Lindner, M., Bank, L., Schilp, J., and Weigold, M., Digital Twins in Manufacturing: A RAMI 4.0 Compliant Concept, Sci, vol. 5, no. 4, p. 40, 4 Dec. 2023. DOI: 10.3390/sci5040040.

Lindner, M. and Koch, T. Energy Flexibility Data Model (EFDM), version 1.0.0, Zenodo, 2023. DOI: 10.5281/zenodo.8409627.

Lindner, M., Grosch, B., Elserafi, G., Dietrich, B., and Weigold, M., Holistic Approach for an Energy-Flexible Operation of a Machine Tool with Cooling Supply, Energies, vol. 16, no. 9, p. 3943, 9 Jan. 2023. DOI: 10.3390/en16093943.

Lindner, M., Öztürk, T., Zink, R., Berchtenbreiter, V., Wünschel, W., Frieß, T., and Weigold, M. Industrie 4.0 Interoperabilität Durch OPC UA Mit Companion Specifications - Mehrwerte Für Stakeholder Des Maschinen- Und Anlagenbaus, VDMA e.V., Ed. Frankfurt am Main: VDMA e.V., 2023. https://vdma.org/documents/34570/778 03117/VDMA_Leitfaden_Mehrwerte_DE.pdf/4bd334d9-3bb3-be2e-5 c83-6fa31bebd2be?t=1681481945330 (visited on 06/08/2023)

2022

Walther, J., Dietrich, B., Grosch, B., Lindner, M., Fuhrländer-Völker, D., Strobel, N., and Weigold, M., A Methodology for the Classification
and Characterisation of Industrial Demand-Side Integration Measures, Energies, vol. 15, no. 3, p. 923, 3 Jan. 2022. DOI: 10.3390/en15030923.

Ahrens, R., Bank, L., Bauer, D., Bauernhansl, T., Fridgen, G., Grigorjan, A., Kalchschmid, V., Kaymakci, C., Köberlein, J., Lindner, M., Lodwig, R., Oeder, A., Potenciano Menci, S., Sauer, A., Schel, D., Schimmelpfennig, J., Schlereth, A., Schöpf, M., Schulz, F., Schulz, J., Stiphoudt, C. van, Weigold, M., and Winter, C. Integration der Flexibilitätsvermarktung (B3), in Energieflexibilität in der deutschen Industrie - Band 2: Markt- und Stromsystem, Managementsysteme und Technologien energieflexibler Fabriken, Sauer, A., Buhl, H. U., Mitsos, A., and Weigold, M., Eds., vol. 2, Stuttgart: Fraunhofer Verlag, 2022, pp. 237–272. https://synergie-projekt.de/wp-content/uploads /2020/08/SynErgie_Band_2.pdf (visited on 10/14/2022)

Bauer, D., Benedikt, F., Bianchini, I., Borst, F., Silva, C. da, Dietrich, B., Emde, A., Fuhrländer-Völker, D., Hofmann, P., Ilieva-König, S., Kaymakci, C., Lindner, M., Moog, D., Retzlaff, J., Riethmüller, T., Sauer, A., Schillinger, C., Seyed Sadjjadi, B., Sossenheimer, J., Stiphoudt, C. van, Weber, T., Weigold, M., and Wenninger, S. Werkzeuge für die energetische Flexibilisierung (B1), in Energieflexibilität in der deutschen Industrie - Band 2: Markt- und Stromsystem, Managementsysteme und Technologien energieflexibler Fabriken, Sauer, A., Buhl, H. U., Mitsos, A., and Weigold, M., Eds., vol. 2, Stuttgart: Fraunhofer Verlag, 2022, pp. 163–208. https://synergie-projekt.de/wp-content/uploads/2020/08/SynErgie_Band_2.pdf (visited on 10/14/2022)

Grosch, B., Ranzau, H., Dietrich, B., Kohne, T., Fuhrländer-Völker, D., Sossenheimer, J., **Lindner, M.**, and Weigold, M., A framework for researching energy optimization of factory operations, Energy Informatics, vol. 5, no. 1, p. 29, Sep. 7, 2022. DOI: 10.1186/s42162-022-00207-6.

2021

Schilp, J., Bank, L., Köberlein, J., Schilp, J., Bank, L., Köberlein, J., Bauernhansl, T., Sauer, A., Kaymakci, C., Eigenbrod, H., Fridgen, G., Bahmani, R., Menci, S. P., Schöpf, M., Stiphoudt, C. van, Weigold, M., and Lindner, M. "Optimierung auf der Energiesynchronisationsplattform," Fraunhofer-Gesellschaft, 2021. DOI: 10.24406/IGCV-N-642371.

Lindner, M., Wenninger, S., Fridgen, G., and Weigold, M. Aggregating Energy Flexibility for Demand-Side Management in Manufacturing Companies – A Two-Step Method, in Production at the Leading Edge of Technology, Behrens, B.-A., Brosius, A., Drossel, W.-G., Hintze, W., Ihlenfeldt, S., and Nyhuis, P., Eds., ser. Lecture Notes in Production Engineering, Cham: Springer International Publishing, 2021, pp. 631–638. DOI: https://doi.org/10.1007/978-3-030-78424-9_69

Fuhrländer-Völker, D., Lindner, M., and Weigold, M., Design Method for Building Automation Control Programs to Enable the Energetic Optimization of Industrial Supply Systems, Procedia CIRP, 54th CIRP CMS 2021 - Towards Digitalized Manufacturing 4.0, vol. 104, pp. 229–234, 2021. DOI: 10.1016/j.procir.2021.11.039

Fridgen, G., Menci, S. P., Stiphoudt, C. van, Schilp, J., Köberlein, J., Bauernhansl, T., Sauer, A., Grigorjan, A., Schel, D., Schlereth, A., Schulz, F., Weigold, M., **Lindner, M.**, Schimmelpfennig, J., and Winter, C. "Referenzarchitektur der Energiesynchronisationsplattform," Fraunhofer-Gesellschaft, 2021. DOI: 10.24406/IGCV-N-642369

Buhl, H. U., Duda, S., Schott, P., Weibelzahl, M., Wenninger, S., Fridgen, G., Menci, S. P., Schöpf, M., Stiphoudt, C. van, Weigold, M., and Lindner, M. "Energieflexibilitätsdatenmodell der Energiesynchronisationsplattform," Fraunhofer-Gesellschaft,2021. DOI: 10.24406/IGCV-

N-642370

Bauernhansl, T., Sauer, A., Kaymakci, C., Schlereth, A., Schilp, J., Kalchschmid, V., Roth, S., Weigold, M., Lindner, M., Zäh, M. F., Schulz, J., Vernim, S., Schimmelpfennig, J., and Winter, C. "Demonstratoren der Energiesynchronisationsplattform," Fraunhofer-Gesellschaft, 2021. DOI: 10.24406/IGCV-N-642373.

Bank, L., Wenninger, S., Köberlein, J., **Lindner, M.**, Kaymakci, C., Weigold, M., Sauer, A., and Schilp, J. Integrating Energy Flexibility in Production Planning and Control - An Energy Flexibility Data Model-Based Approach, Hannover : Institutionelles Repositorium der Leibniz Universität Hannover, 2021. DOI: 10.15488/11249.

Bachmann A., Bank L., Bark C., Bauer D., Blöchl B., Brugger M., Buhl H. U., Dietz B., Donnelly J., Friedl T., Halbrügge S., Hauck H., Heil J., Hieronymus A., Hinck T., Ilieva-König S., Johnzén C., Koch C., Köberlein J., Köse E., Lochner S., **Lindner M.**, Mayer T., Mitsos A., Roth S., Sauer A., Scheil C., Schilp J., Schimmelpfennig J., Schulz J., Schulze J., Sossenheimer J., Strobel N., Tristan A., Vernim S., Wagner J., Wagon F., Weibelzahl M., Weigold M., Weissflog J., Wenninger S., Wöhl M., Zacharias J., and Zäh M. F., Energieflexibel in die Zukunft – Wie Fabriken zum Gelingen der Energiewende beitragen können, 2021. DOI: 10.24406/FIT-N-638765.

$\boldsymbol{2020}$

Reinhart, G., Bank, L., Brugger, M., Bauernhansl, T., Sauer, A.,
Bauer, D., Kaymakci, C., Schel, D., Schlereth, A., Fridgen, G., Buhl,
H. U., Bojung, C., Schott, P., Weibelzahl, M., Wenninger, S., Weigold,
M., Lindner, M., Ronge, K., Oeder, A., Schimmelpfennig, J., Winter,
C., Jarke, M., and Ahrens, R. "Konzept der Energiesynchronisationsplattform. Diskussionspapier V3," Fraunhofer IGCV, Augsburg, 2020.
DOI: 10.24406/igcv-n-602416

Curriculum Vitae

Experience

05/2019 - 04/2024	 Institute of Production Management, Technology and Machine Tools Technical University of Darmstadt Darmstadt Germany Research associate Head of Research Unit Cyberphysical production systems (since 04/2023)
11/2018 - 04/2019	Institute of Process Measurement and Sen- sor Technology TU Ilmenau Ilmenau Germany Research assistant
06/2016 - 08/2018	Fraunhofer IOSB-AST Ilmenau Germany Research student assistant
09/2015 - 02/2016	SAXON Junkalor GmbH Dessau-Roßlau Germany Bachelor student
10/2014 - 02/2015	SAXON Junkalor GmbH Dessau-Roβlau Germany Internship
02/2012 - 09/2012	von der Heydt Service GmbH & Co. KG Dessau-Roßlau Germany Mechatronics technician
09/2008 - 01/2012	von der Heydt Service GmbH & Co. KG Dessau-Roßlau Germany Dual education as mechatronics technician

Education

05/2019 - today	PhD student at the Department of Mechanical Engineering Technical University of Darmstadt Darm- stadt Germany
04/2016 - 12/2018	Master of Science - Electrical and Computer Engineering TU Ilmenau Ilmenau Germany Master thesis: "Entwicklung von Online- Algorithmen zur Analyse von PMU Daten- strömen bei sich verändernden Prozessbe- dingungen"
10/2012 - 03/2016	 Bachelor of Engineering - Electrical and Computer Engineering Hochschule Anhalt University of Applied Sciences Köthen Germany Bachelor thesis: "Analyse und Opti- mierung einer analogen Signalverarbeitung für Staubmessgeräte"
07/2008	Abitur – Advanced-school-leaving- certificate Ludwigsgymnasium Köthen Köthen Germany